PAD-ENM-0064/R1

Groundwater Assessment Report for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky



This document is approved for public release per review by: 5-1-2013 LATA Kentucky Classification Support Date

PAD-ENM-0064/R1

Groundwater Assessment Report for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky

Date Issued—May 2013

Prepared for the U.S. DEPARTMENT OF ENERGY Office of Environmental Management

Prepared by LATA ENVIRONMENTAL SERVICES OF KENTUCKY, LLC managing the Environmental Remediation Activities at the Paducah Gaseous Diffusion Plant under contract DE-AC30-10CC40020

PROFESSIONAL GEOLOGIST AUTHORIZATION

DOCUMENT IDENTIFICATION:

Groundwater Assessment Report for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, PAD-ENM-0064/R1

Stamped and signed pursuant to my authority as a duly registered geologist under the provisions of KRS Chapter 322A.

Kenneth R. Davis

And Arofessional and Ar 2013 PG1194

PR	OFES	SIONAL	L GEOLOGIST AUTHORIZATION	iii		
TA	BLES			vii		
FIC	GURE	S		ix		
AC	RON	YMS		xi		
EX	ECUI	TIVE SU	MMARY	ES-1		
1.	C-74	6-U LAN	NDFILL GROUNDWATER BACKGROUND	1		
	1.1	C-746-1	U LANDFILL CONSTRUCTION HISTORY	1		
	1.2	GROUI	NDWATER CHARACTERIZATION BEFORE OPERATION			
	1.3	GROUI	NDWATER MONITORING PLAN	3		
2.	BAS	IS FOR (CONDUCTING ASSESSMENT/PRELIMINARY SCREENING	7		
	2.1	HISTO	RY	7		
	2.2	PURPC	DSE OF THE ASSESSMENT			
	2.3	SUMM	ARY OF THE ASSESSMENT SCREENING			
		2.3.1	Inorganic Parameters Screening against Kentucky or EPA MCL	9		
		2.3.2	Radionuclide Parameters Screening against Kentucky or EPA MCL	9		
		2.3.3	Organic Parameters Screening against Kentucky or EPA MCL	9		
		2.3.4	PCBs Screening against EPA MCL	15		
		2.3.5	Parameters Screening against EPA RSLs and Secondary MCLs	15		
		2.3.6	Parameters without MCLs or RSLs			
		2.3.7	Assessment Parameters Summary			
		2.3.8	Additional Assessment Parameters Summary	17		
	2.4	CONCI	LUSIONS OF THE ASSESSMENT SCREENING	17		
3.	DISC	CUSSION	N OF CONSTITUENTS NOT ELIMINATED BY SCREENING			
	31	BETA	ACTIVITY	21		
	011	3.1.1	Physical Properties			
		3.1.2	Gross Beta Particle Activity C-746-U Landfill			
	3.2	TCE				
		3.2.1	Physical Properties			
		3.2.2	Mobility of TCE in the Terrestrial Environment			
		3.2.3	TCE Concentrations at C-746-U Landfill			
		3.2.4	TCE at the C-746-U Landfill			
		3.2.5	Results of Assessment Actions			
4			I DISCUSSION OF CONSTITUENTS IN DCA WELL SAMPLES	22		
4.		אאטרני הספות	L DISCUSSION OF CONSTITUENTS IN KOA WELL SAMPLES			
	4.1 1 0	MVCM DI2201	EVIIM			
	7.2 MANCANESE					
	4.5 1 1					
	4.4	SULFA				

CONTENTS

	4.5	PCBs		
		4.5.1	Physical Properties	
		4.5.2	PCBs at C-746-U Landfill	
		4.5.3	Results of Assessment Actions	
	4.6	PCB, T	OTAL	
	4.7	TOTAL	L ORGANIC CARBON	
	4.8	DISCU	SSION OF LANDFILL SOURCES	
5	нур	POGEC	N OGIC SETTING	41
5.	51	PHYSI	CAL CHARACTERISTICS OF THE AREA	
	5.2	GENE	RAL GEOLOGY	
	5.3	SUBSU	JRFACE GEOLOGY AT THE C-746-U LANDFILL	
	5.4	HYDR	OGEOLOGY AT PGDP AND THE C-746-U LANDFILL	
	5.5	CONC	EPTUAL SITE MODEL	
6.	ENH	ANCED	SAMPLING OF EXISTING WELLS	
	6.1	EXIST	ING WELLS	
7	CON	CLUSI	ONG	52
1.	CON 7 1	DECOI	JNS MMENDATIONS	
	/.1	KECO	WIMENDATIONS	
8. I	REFE	RENCES	S	
AP	PEND	IX A:	CORRESPONDENCE	A-1
AP	PEND	IX B:	TREND CHARTS	B-1
٨D	DENID		HOME SITE WELL WATED SAMPLE ANALYTICAL DESULTS	C 1
Л	LIND	IA C.	HOME SITE WELL WATER SAMI LE ANALTHCAL RESOLTS	
AP	PEND	IX D:	PCB IN SOILS FIELD INVESTIGATION SUMMARY	D-1
AP	PEND	IX E:	ASSESSMENT OF WELL REHABILITATION SOURCE OF PCB	
			CONTAMINATION IN C-746-U LANDFILL WELLS	E-1
AP	PEND	DIX F:	ASSESSMENT OF IMPACTS OF OVERPUMPING MONITORING	
			WELLS IN THE VICINITY OF THE C-746-U LANDFILL ON PCB	F 1
			CUNCENTRATIONS	F-1
ΔP	PEND	IX G	LITHOLOGIC LOGS/WELL LOGS	G-1
1 11				

TABLES

ES.1.	Summary of RGA Well Screening of Inorganic Parameters with Kentucky or EPA MCL	ES-9
ES.2.	Summary of RGA Well Screening of Radionuclide Parameters with Kentucky or EPA	
	MCL	. ES-10
ES.3.	Summary of RGA Well Screening of Organic Compound Parameters with Kentucky	
	MCL	. ES-10
ES.4.	Summary of RGA Well Screening of PCBs against EPA MCL	. ES-11
ES.5.	Summary of RGA Well Screening of Parameters without Kentucky or EPA MCL, but	
	with Tapwater RSL or EPA Secondary MCL	. ES-11
ES.6.	Summary of RGA Well Screening of List of Parameters without Kentucky or EPA MCL	
	and without Tapwater RSL or EPA Secondary MCL	. ES-13
1.	Summary of RGA Well Screening of List of Inorganic Parameters with Kentucky or EPA	
	MCL	10
2.	Summary of RGA Well Screening of Radionuclide Parameters with Kentucky or EPA	
	MCL	11
3.	Summary of RGA Well Screening of Organic Compound Parameters with Kentucky	
	MCL	12
4.	Summary of RGA Well Screening of PCBs against EPA MCL	12
5.	Summary of RGA Well Screening of Parameters without Kentucky or EPA MCL, but	
	with Tapwater RSL or EPA Secondary MCL	13
6.	Summary of RGA Well Screening of List of Parameters without Kentucky or EPA MCL	
	and without Tapwater RSL or EPA Secondary MCL	14
7.	Summary of Conclusions for Assessment Parameters in All Wells Based on Screening	16
8.	Summary of Conclusions for Additional Assessment Parameters in All Wells Based on	
	Screening	17
9.	PGDP RGA Wells with PCB Analyses and the C-746-S&T and C-746-U Landfill Wells	51
10.	Summary of Possible Contaminant Sources	53

FIGURES

1.	Location of the U-Landfill Depicted Relative to 2009 RGA TCE Plume Map	2
2.	C-746-U Landfill and MWs Located in the Vicinity	4
3.	Location of C-746-U Landfill Relative to 2009 Tc-99 RGA Plume Map and Water Policy	
	Boundary	22
4.	Technetium-99 Contours for the RGA	24
5.	Technetium-99 versus Beta Activity in MW372 at the C-746-U Landfill	25
6.	Technetium-99 versus Beta Activity in MW373 at the C-746-U Landfill	25
7.	Technetium-99 versus Beta Activity in MW358 at the C-746-U Landfill	
8.	Technetium-99 versus Beta Activity in MW366 at the C-746-U Landfill	
9.	Technetium-99 versus Beta Activity in MW367 at the C-746-U Landfill	27
10.	Technetium-99 versus Beta Activity in MW369 at the C-746-U Landfill	27
11.	Technetium-99 versus Beta Activity in MW361 at the C-746-U Landfill	
12.	Technetium-99 versus Beta Activity in MW364 at the C-746-U Landfill	
13.	Technetium-99 versus Beta Activity in MW365 at the C-746-U Landfill	29
14.	Technetium-99 versus Beta Activity in MW370 at the C-746-U Landfill	29
15.	Technetium-99 versus Beta Activity in MW371 at the C-746-U Landfill	30
16.	Composite Trichloroethene Contours for the RGA for Calendar Year 2009	31
17.	Groundwater Monitoring Well Network for the C-746-S&T Landfills	34
18.	Regional Geologic Map	43
19.	Schematic Stratigraphic Column of the PGDP Region	44
20.	Geologic Section Depicting Generalized Stratigraphy Underlying C-746-U Landfill	45
21.	Topographic Map of the Paducah Gaseous Diffusion Plant C-746-U Landfill	46
22.	Location of RGA Wells with PCB Analyses in Addition to the C-746-S&T and C-746-U	
	Landfill Wells	50

ACRONYMS

COD	chemical oxygen demand
CSM	Conceptual Site Model
DAF	dilution attenuation factor
DOE	U.S. Department of Energy
EPA	U.S. Environmental Protection Agency
KAR	Kentucky Administrative Regulations
KDWM	Kentucky Division of Waste Management
LATA Kentucky	LATA Environmental Services of Kentucky, LLC
LCD	Lower Continental Deposits
LRGA	Lower Regional Gravel Aquifer
MCL	maximum contaminant level
MDA	minimum detectable activity
MW	monitoring well
N/A	not applicable
ND	nondetect
PGDP	Paducah Gaseous Diffusion Plant
RGA	Regional Gravel Aquifer
RSL	regional screening level
SMCL	secondary maximum contaminant level
TOC	total organic carbon
UCD	Upper Continental Deposits
UCRS	Upper Continental Recharge System
URGA	Upper Regional Gravel Aquifer

EXECUTIVE SUMMARY

This Groundwater Assessment Report for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, summarizes results of the activities conducted under the approved Groundwater Assessment Plan for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky (PRS 2008). The C-746-U Landfill was placed in assessment in response to the finding that some constituents were found in groundwater samples from monitoring wells (MWs) located in the vicinity of the C-746-U Landfill at above background levels. The C-746-U Landfill originally was placed into assessment on March 2, 1999.

Tables in the Executive Summary provide a summary of the conclusions from the evaluation of the groundwater data for parameters evaluated as part of the C-746-U Landfill monitoring program. These

tables include those parameters (and associated wells) identified for assessment by the Kentucky Division of Waste Management (KDWM) in August 2006 as well as additional parameters (and wells) subsequently identified for assessment by the U.S. Department of Energy (DOE). The data used to support this assessment are groundwater analyses of quarterly and semiannual monitoring for the period 2002 through 2012 and the focused sampling of October 2006. Additional soils analyses from a polychlorinated biphenyls (PCBs) source investigation in 2008 and groundwater analyses from an overpumping test conducted during 2011 were used to support the assessment of Total PCBs.

Concentrations of some constituents in some Regional Gravel Aquifer (RGA) wells continue to benchmarks [Kentucky exceed maximum contaminant levels (MCLs), U.S. Environmental Protection Agency (EPA) Drinking Water Standard MCLs, and EPA regional screening levels (RSLs) for tapwater], including C-746-U Landfill background concentrations developed in accordance with the permit. The concentrations of these constituents were reviewed with the principal focus on the results collected since January 2010 (that includes the most recent eight quarters of data). This report summarizes an evaluation of the C-746-U Landfill parameters, discusses each of the constituents, and describes the following:

Summary of the Screening against Benchmarks

To support the evaluation of whether there is evidence that the C-746-U Landfill is contributing to groundwater contamination, the concentrations (from samples from Regional Gravel Aquifer [RGA] wells in the vicinity of the C-746-U Landfill) of a broad range of parameters were screened against benchmarks.

Benchmarks: The following benchmarks were used in the screening:

- KY MCLs: Maximum contaminant levels (MCLs), as identified in 401 *KAR* 47:030;
- EPA MCLs: MCLs (primary) promulgated by U.S. EPA;
- Regional screening levels (RSLs) for tapwater, developed by EPA to use for screening;
- Secondary MCLs (SMCLs): EPA guidelines for constituents that may have an aesthetic impact on drinking water; and
- C-746-U Landfill background developed according to the permit as discussed further in the Results and Discussion section of this Executive Summary and in Section 2.1.
- Whether concentrations of individual constituents are above benchmarks;
- Whether the concentrations are indicative of a C-746-U Landfill source. [NOTE: As discussed below, the constituent concentrations in Upper Continental Recharge System (UCRS) wells are considered to be representative only of the conditions local to the well or sourced from overlying soils. Thus, no discussion of potential "upgradient" sources is presented for the UCRS because the concept is not consistent with the site conceptual model.] The constituent concentrations found in the RGA wells are

evaluated while considering the potential for contributions of sources located upgradient (with respect to the C-746-U Landfill) of the monitored well];

- Suspected sources of the elevated concentrations of the specific constituent in the specific well; and
- Mechanisms for the release of these constituents, where these suspected sources are attributed to a particular mechanism.

FINDINGS

This groundwater assessment has determined that there is no evidence that would indicate a release from the C-746-U Landfill. The evaluation process used in this groundwater assessment and a summary of the findings that support this conclusion are detailed below.

EVALUATION PROCESS, SCREENING

The current and historical analytical results from Regional Gravel Aquifer (RGA) wells located in the vicinity of the C-746-U Landfill were screened against benchmarks, including Kentucky (KY) maximum contaminant levels (MCLs), U.S. Environmental Protection Agency (EPA) MCLs, and EPA regional screening levels (RSLs). Subsequently, those constituents without MCLs or RSLs were evaluated against EPA SMCLs and historical background levels (calculated in accordance with the C-746-U Landfill permit). Each result was compared with each benchmark concentration, and any exceedances of the benchmarks were tabulated with a special focus on those constituents that exceeded the benchmark over the past eight calendar quarters. Sixty-four parameters were evaluated with concentrations compared with benchmarks. The results of the screening are summarized as follows:

- Table 1 (and Table ES.1) summarizes the screening of 14 inorganic parameters against KY or EPA MCLs;
- Table 2 (and Table ES.2) summarizes the screening of 10 radionuclides against KY or EPA MCLs;
- Table 3 (and Table ES.3) summarizes the screening of 8 volatile organic compounds against KY or EPA MCLs;
- Table 4 (and Table ES.4) summarizes the screening of 9 PCBs against EPA MCLs (PCBs do not have a KY MCL);
- Table 5 (and Table ES.5) summarizes the screening of 15 parameters against RSLs or SMCLs (these parameters do not have primary MCLs); and
- Table 6 (and Table ES.6) summarizes the screening of 8 parameters (that don't have RSLs or MCLs) against historical C-746-U Landfill background calculated in accordance with the permit.

The screening results presented in these tables document those constituents that have not had a confirmed exceedance of any benchmark in any RGA well. The lack of any current or historical exceedance of a benchmark in an RGA well resulted in this constituent's being screened from further evaluation.

The evaluation continued for those constituents that have had an RGA exceedance of a benchmark, with particular focus on those constituents that had a confirmed exceedance of a benchmark over the past eight

calendar quarters. This subsequent evaluation into the source(s) of these constituents looked at the following:

- Spatial patterns of each constituent present above a benchmark;
- Identified plumes;
- Concentrations of these constituents in other Paducah Gaseous Diffusion Plant (PGDP) wells; and
- Concentrations of these constituents in C-746-U Landfill leachate.

The additional evaluation discusses sources of the constituents in RGA wells that were not screened out using the process described above.

EVALUATION PROCESS, ADDITIONAL DISCUSSION

For those constituents that exceeded benchmarks, the concentrations of these constituents were compared with the following:

- Concentrations in RGA groundwater present in wells located upgradient of the C-746-U Landfill. The concentration (and distribution) of each constituent was evaluated against the upgradient concentrations to identify whether the C-746-U Landfill contributed to the concentrations or whether upgradient sources could account for the constituent concentration.
- Concentrations of naturally occurring constituents in RGA groundwater typical of the PGDP, but outside the vicinity of the C-746-U Landfill. If concentrations of naturally occurring constituents found in the C-746-U Landfill RGA wells were within the range of concentrations found elsewhere at PGDP, the evaluation concluded that there was no indication of a statistically quantifiable contribution of RGA contamination from the C-746-U Landfill.
- Concentrations of constituents present in leachate from both the C-746-U Landfill and the C-746-S Landfill to identify if the landfill leachate is a potential source. This evaluation considered the fact that the dilution attenuation factor (DAF) through the Upper Continental Recharge System (UCRS) has been estimated at 58, through groundwater modeling conducted for other projects (see Section 4.8). Thus, for the leachate to be the source of the constituent concentrations, its concentration in the leachate would have to be ~58 times greater in the leachate than in the RGA groundwater. The relative concentrations of RGA groundwater constituents were compared to leachate concentrations in Appendix B.
- Concentrations in UCRS groundwater. As described in more detail in the discussion of the conceptual site model, migration of constituents occurs vertically through the UCRS. Thus, concentrations of constituents in the UCRS were evaluated to identify any potential UCRS sources of RGA contamination. Because of the vertical-only migration of constituents through the UCRS and the fact that there are no UCRS wells completed through the C-746-U Landfill, the presence of any constituents in a UCRS well could not be properly attributed to migration from the C-746-U Landfill operations (though constituents in these wells, if found, could be evidence of a secondary source of C-746-U Landfill-related contamination). The results of the UCRS well sampling were considered when evaluating the potential sources of constituents in RGA wells.
- **Concentrations during the past eight calendar quarters.** Those constituents that were infrequently detected historically *and* did not have an exceedance during the past eight calendar quarters typically

were screened from further evaluation; however, the summary tables provide information on the frequency and timing of the exceedances.

The comparisons listed above were used to identify potential source(s) of those constituents confirmed to be present at concentrations that were above benchmarks during the past eight calendar quarters.

In addition to the discussion of parameters found above benchmarks (as summarized above), the evaluation of the presence of polychlorinated biphenyls (PCBs) was much more thorough. Several additional investigations have been performed over the years to address the historical presence of PCBs in RGA wells in the vicinity of the C-746-U Landfill. Although PCB concentrations in RGA wells have been below EPA MCLs for the past eight calendar quarters, the information that supports the determination of the source of historical PCBs in RGA groundwater is summarized.

SUMMARY OF RESULTS

Screening Observations

Following are most of the constituents that were screened from further consideration, because they did not have any parameter concentrations above benchmarks for the past eight calendar quarters.

- Of the 41 constituents with EPA or Kentucky MCLs, only 2, beta activity and trichloroethene (TCE), had an exceedance of a benchmark in RGA wells over the past 8 calendar quarters (see Tables 1, 2, and 3). Beta activity and TCE are discussed further.
- Although the nine PCBs have not had a confirmed exceedance of the EPA MCL in RGA wells over the past eight calendar quarters, the source of the historical exceedances is further discussed (see Table 4).
- Only 3 of 15 constituents exceed their respective RSLs (e.g., cobalt, iron, and manganese) in RGA wells, but these constituents do not exceed the C-746-U Landfill background (see Table 5). Some additional discussion is provided.
- Three constituents exceed the C-746-U Landfill background or an SMCL (e.g., calcium, magnesium, and dissolved solids) in RGA wells, but these constituents exceed these benchmarks only in wells MW372 and MW373, wells located upgradient of the C-746-U Landfill. Additional discussion of the potential source(s) of these constituents is provided, along with additional discussion of other parameters that have similar patterns.

Additional Discussion/Summary of Conclusions

As summarized above, most of the 64 parameters evaluated as part of this groundwater assessment have concentrations in RGA groundwater that are below benchmarks. The few constituents that have exceeded benchmarks over the past 8 calendar quarters (or have historical exceedances) are not properly attributed to a C-746-U Landfill source as summarized below.

Beta Activity. Beta activity has exceeded the Kentucky MCL of 50 pCi/L in three wells in the vicinity of the C-746-U Landfill. An evaluation of the assessment data demonstrates that the beta activity in these wells apparently is sourced from upgradient of the C-746-U Landfill and is associated with migration of the historical Tc-99 plume, as follows:

- None of the other RGA wells in the vicinity of the C-746-U Landfill have beta activity that exceeds the KY MCL.
- Figure B.6 shows recent values for beta activity that demonstrate that beta activity is stable to declining in all wells and is greater in the upgradient well MW372 than in the two downgradient wells (MW366 and MW367) located along the same flow path.
- Figure B.7 also indicates that beta activity in these three wells is stable to declining, with the activity in the upgradient well MW372 decreasing at a slightly greater rate.
- The beta activity in leachate (see Tables B.2 and B.3) is not high enough to account for the beta activity found in the RGA wells given the DAF of 58 for UCRS materials.
- Tc-99 is a beta activity emitter. The Tc-99 concentrations are well-correlated to the beta activity as shown by the graphs of Tc-99 and beta activity for each of the C-746-U Landfill RGA wells depicted in Figures 5–15.
- The wells with beta activity exceedances of the KY MCL are immediately downgradient of an arm of the Tc-99 plume as shown in Figures 3 and 4.

In summary, the available data do not statistically support an interpretation of a release from the C-746-U Landfill. Rather, the upgradient Tc-99 plume is the most likely source of the beta activity found in RGA wells in the vicinity of the C-746-U Landfill. There is no statistically quantifiable contribution of Tc-99 to the RGA by the C-746-U Landfill.

TCE. TCE is the only volatile organic compound with concentrations that exceed the Kentucky and EPA MCL in wells in the vicinity of the C-746-U Landfill. An evaluation of the assessment data demonstrates that the TCE in these wells (MW357, MW358, MW361, MW372, and MW373) apparently is sourced from upgradient of the C-746-U Landfill and is associated with migration of the historical TCE plume, as follows.

- None of the other RGA wells in the vicinity of the C-746-U Landfill have TCE concentrations that exceed the KY MCL.
- Figure B.9 shows recent values for TCE that are generally within a factor of 2 of the MCL.
- TCE concentrations in upgradient wells MW372 and MW373 generally are higher than those in other C-746-U Landfill wells, as shown in Figure B.9.
- Recent values for TCE are greater in the upgradient wells MW372 and MW373 (and just above the MCL of 5 μ g/L) than in the two downgradient wells (MW366 and MW367) located along the same flow path (where concentrations are below the MCL).
- There is no detectable TCE in leachate from the C-746-U Landfill, as summarized in Table B.2.
- TCE use at the site was discontinued prior to the opening of the C-746-U Landfill; thus, it would not have been used in ancillary operations at the C-746-U Landfill.
- The wells with TCE exceedances of the MCL are immediately downgradient of a portion of the historical TCE plume, as shown in Figure 1.

In summary, the available data do not support an interpretation of a statistically quantifiable release from the C-746-U Landfill to the RGA. Rather, the upgradient TCE plume is the most likely source of the TCE found in RGA wells in the vicinity of the C-746-U Landfill. There is no statistically quantifiable contribution of TCE to the RGA by the C-746-U Landfill.

PCBs. PCBs have not had a confirmed exceedance of the EPA MCL of 0.0005 mg/L in any of the RGA wells in the vicinity of the C-746-U Landfill over the past eight calendar quarters; however, PCBs have been a historical concern. An evaluation of the assessment data demonstrates that the historical PCB concentrations may be attributed to cross-contamination of wells with PCBs during well rehabilitation in 2003. The C-746-U Landfill is not the source of the historical PCB exceedances, as follows:

- Appendix E presents information on well sampling that demonstrates that PCBs were not an issue in C-746-U Landfill wells until 2003 (when the well rehabilitation program was performed). This appendix also demonstrates that PCB concentrations have decreased over the years.
- Figure 22 and Table 9 shows that PCBs have not been an issue in other RGA wells at PGDP. This is consistent with the fact that PCBs are not soluble in water; thus, though PCBs were historically used at PGDP, they are not found in RGA groundwater at locations across the site.
- Appendix F presents information on the results of the overpumping tests conducted on C-746-U Landfill wells. This overpumping has contributed to the finding that concentrations in RGA wells are below EPA MCLs. Once the RGA wells were rehabilitated to minimize the residual sediment, the PCB concentrations decreased to below the EPA MCL. Note: There is one UCRS well (MW365) that still has concentrations above the EPA MCL, but UCRS wells cannot be overpumped effectively to remove PCB-contaminated sediment to the same degree as RGA wells (well pumps dry)—a finding consistent with a historical (and not-continuing) PCB source.
- Appendix D summarizes the investigation of PCBs in soils conducted in the vicinity of UCRS well MW365 that did not find any PCB source in UCRS soils in the vicinity of the well.

These results support a finding that, regardless of the source of the detected PCBs, the levels have been below EPA MCLs for eight quarters, and currently there are no PCB values that represent exceedances attributable to the C-746-U Landfill. The historical PCB contamination may have been the result of cross-contamination of wells during rehabilitation that occurred in 2003.

Cobalt, Iron, and Manganese. Although cobalt, iron, and manganese were found in RGA wells above EPA RSLs, they were not found at concentrations above the C-746-U Landfill background (as determined in accordance with the permit requirements). In addition, the concentrations of these constituents are comparable to concentrations found in RGA wells at other locations at PGDP, as summarized in Table B.1. Thus, there is no indication of a statistically quantifiable C-746-U Landfill source, and the concentrations found in RGA wells are found in wells at PGDP locations well outside any potential influence of the C-746-U Landfill.

Dissolved Solids, Calcium, and Magnesium. These constituents do not have a primary MCL or RSL, but they are found in some C-746-U Landfill wells at concentrations that are elevated above background. These constituents (and other constituents that may be considered elevated) do not have the C-746-U Landfill as their source because of the following:

• The highest concentrations of these constituents are found in upgradient wells MW372 and MW373, as summarized in Tables 5 and 6; and

• The concentrations of these constituents are not sufficiently elevated in the leachate for the C-746-U Landfill to be a statistically quantifiable source, as summarized in Table B.2.

Potential Alternate Sources. The upgradient wells MW372 and MW373 have concentrations of a few constituents that are above benchmarks; however, other C-746-U Landfill upgradient wells do not have elevated concentrations so the source of the concentrations in these two wells cannot be pinpointed. The potential sources of these few exceedances are non-C-746-U sources, like biofouling, upgradient RGA sources, or excessive turbidity.

This evaluation found that there is no identified upgradient source that accounts for these constituents. For example, Table B.3 compares concentrations in C-746-S Landfill leachate to concentrations in MW372 and MW373 and finds that the C-746-S Landfill is not the source of the elevated concentrations in these wells. The constituents are not elevated in other wells located upgradient of the C-746-U Landfill and downgradient of the C-746-S Landfill.

Finally, there is no threat to human health from any constituents present in wells in the vicinity of the C-746-U Landfill because the C-746-U Landfill is located within the Water Policy Box created to support Comprehensive Environmental Response, Compensation, and Liability Act actions so that exposure to groundwater in this area is limited, irrespective of the source of constituents. The PGDP Action Memorandum for the Water Policy provides municipal water to users within the Water Policy Box, which includes all areas in the vicinity of the C-746-U Landfill.

In summary, the assessment has found that the few constituents that have concentrations above benchmarks do not have the C-746-U Landfill as their source. Some additional evaluation of the source(s) of elevated concentrations of constituents in upgradient wells MW372 and MW373 may be appropriate, but this evaluation can be performed outside of the assessment process. This report presents results of the evaluation as follows:

- Section 1 presents background information on the C-746-U Landfill and the groundwater assessment.
- Section 2 presents the basis for the assessment and preliminary screening of concentrations against benchmarks.
- Section 3 provides discussion of the nature of constituents present above benchmarks.
- Section 4 provides additional discussion on the potential source(s) of these constituents in C-746-U Landfill wells.
- Section 5 describes the hydrogeologic setting.
- Section 6 presents a summary of the enhanced sampling.
- Section 7 provides conclusions of the assessment.
- References are provided in Section 8.

The seven appendices to this document provide additional supporting information and data: Appendix A, Correspondence; Appendix B, Trend Charts; Appendix C, Home Site Well Water Sample Analytical Results; Appendix D, PCBs in Soils Field Investigation Summary; Appendix E, Assessment of Well Rehabilitation Source of PCB Contamination in C-746-U Landfill Wells; Appendix F, Assessment of

Impacts of Overpumping in Monitoring Wells in the Vicinity of the C-746-U Landfill on PCB Concentrations; and Appendix G, Lithologic Logs/Well Logs.

BACKGROUND

DOE owns and operates waste treatment, storage, and disposal units at PGDP near Paducah, Kentucky. Three of these units are landfills that are regulated by DOE, under the provisions of the Atomic Energy Act, and by the Commonwealth of Kentucky under the Resource Conservation and Recovery Act and the Kentucky Solid Waste Landfill Regulations.

The three landfills are located on approximately 80 acres of DOE-owned property immediately north of PGDP. The permitted landfills currently included within this property are the following:

- C-746-S Residential Landfill
- C-746-T Inert Landfill
- C-746-U Contained Landfill

Both C-746-S and C-746-T are permitted currently for post-closure activities only. The C-746-U Landfill is an operating solid waste landfill that covers an area of about 60 acres located directly north of the C-746-S&T Landfills. It is operated by LATA Environmental Services of Kentucky, LLC, (LATA Kentucky) and owned by DOE. KDWM issued Solid Waste Permit SW07300045 in November 1996 to allow construction and operation of the C-746-U Landfill and reissued combined Solid Waste Permits SW07300014, SW07300015, SW07300045, effective November 5, 2006, to allow continuing maintenance of the C-746-S&T Landfills and operation of the C-746-U Landfill.

KDWM issued correspondence August 29, 2006, that placed the C-746-U Landfill in groundwater assessment. In its correspondence, KDWM identified contaminants and wells that require assessment, based on quarterly groundwater monitoring reports that have been submitted by DOE. A Groundwater Assessment Plan is required by 401 *KAR* 48:300, Section 8, for facilities in groundwater contamination assessment. The *Groundwater Assessment Plan for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky,* PRS-PROJ/0006/R2, describes the steps used to perform the groundwater evaluations (PRS 2008). The plan contains additional information required by 401 *KAR* 48:300, Section 8. A groundwater assessment was performed in accordance with the approved plan and the results of this assessment are summarized below.

RESULTS AND DISCUSSION

Tables ES.1 through ES.6 summarize the results of the groundwater assessment. Although concentrations of some constituents in some wells continue to exceed benchmarks (including historical background levels), the concentration levels remain low and further evaluation of these concentrations indicate that the C-746-U Landfill is not the source of the elevated levels. The assessment of the RGA wells does not indicate any landfill-related above background levels of constituents present above MCLs or other risk-based levels of concern.

Table ES.1 summarizes the screening results that demonstrate that none of the inorganic parameters with a Kentucky or EPA primary MCL exceeds its MCL in RGA wells located in the vicinity of the C-746-U Landfill. Most inorganic constituents (antimony, barium, chromium, copper, lead, mercury, selenium, silver, uranium, nitrate, and fluoride) never have exceeded the MCL. Arsenic, beryllium, and cadmium have not had exceedances of a Kentucky MCL or EPA MCL since 2004.

	Kontuolay	EDA			Exceed	
Parameter	MCL	MCL	Units	Exceeds?	2010?	Comments
Antimony	N/A	0.006	mg/L		No	
						No exceedance of Kentucky MCL.
Arsenic	0.05	0.01	mg/L	EPA MCL	No	Last exceedance of EPA MCL in 2004.
Barium	2.0	2.0	mg/L		No	
						Only exceedance of EPA MCL in
Beryllium	N/A	0.004	mg/L	EPA MCL	No	2005; newer results snow no exceedances
Derymum	1011	0.001	ing 11	KY/EPA	110	
Cadmium	0.005	0.005	mg/L	MCL	No	Only exceedance 2003.
Chromium	0.100	0.100	mg/L		No	
Copper	N/A	1.3	mg/L		No	
Lead	0.05	0.015	mg/L		No	
Mercury	0.002	0.002	mg/L		No	
Selenium	0.05	0.05	mg/L		No	
Silver	0.05	N/A	mg/L		No	
Uranium	N/A	0.030	mg/L		No	
Nitrate (as N)	10.0	10.0	mg/L		No	
Fluoride	4.0	4.0	mg/L		No	

 Table ES.1. Summary of RGA Well Screening of Inorganic Parameters with Kentucky or EPA MCL

N/A = not applicable

Table ES.2 summarizes the screening results that demonstrate that beta activity is the only radionuclide that exceeds its MCL in the RGA wells located in the vicinity of the C-746-U Landfill. Additional discussion of the source of the beta activity is presented in the report. The EPA MCL for beta activity is 4 mrem/year. The dose level will vary depending upon the radionuclide that provides the activity. The derived MCL for beta activity from technetium-99 (Tc-99), for example, is 900 pCi/L (according to EPA methodology). Additional evaluation and discussion of the source of the beta activity is provided in this report. Most radionuclide parameters (alpha activity, strontium-90, tritium, iodine-131, radium, radium-224, radium-226, radium-228, and Tc-99) do not exceed the MCL.

Table ES.3. summarizes the screening results that demonstrate that TCE is the only organic compound that exceeds its MCL in the RGA wells located in the vicinity of the C-746-U Landfill. Additional discussion of the source(s) of the TCE is provided. Most organic compound parameters (i.e., benzene, carbon tetrachloride, 1,2-dichloroethane, p-dichlorobenzene, 1,1-dichloroethene, 1,1,1-trichloroethane, vinyl chloride) have never been detected in C-746-U Landfill wells.

Table ES.4. summarizes the screening results that demonstrate that the concentrations of PCBs in the RGA no longer exceed the EPA MCL. This report also summarizes additional discussion and evaluation that demonstrates that the historical PCB contamination is the likely result of cross-contamination from the 2003 well rehabilitation event.

Table ES.5. summarizes the screening results that demonstrate that the concentrations of cobalt, iron, and manganese are present above their RSL; however, these constituents are not present in concentrations above C-746-U Landfill background (as determined in accordance with the C-746-U Landfill permit). In addition, these constituents are present at concentrations within the range of concentrations found in other

Table ES.2. Summary of RGA Well Screening of Radionuclide Parameters with Kentucky or EPA MCL

	Kontuola	EDA			Exceed	
Parameter	MCL	MCL	Units	Exceeds?	2010?	Comments
				KY/EPA		
Alpha Activity	15	15	pCi/L	MCL	No	Only MCL exceedance in 2004.
						12 KY MCL exceedances since
						2010; additional
						evaluation/discussion to ascertain
Beta Activity	50	4 mrem	pCi/L	KY MCL	Yes	source(s).
						Only KY MCL exceedance in
Strontium-90	8	N/A	pCi/L	KY MCL	No	2003.
						No KY MCL exceedance. Only
Tritium	20,000	N/A	pCi/L		No	detection in 2007.
						No exceedance. Single high result
						not confirmed upon
Iodine-131	3	N/A	pCi/L	KY MCL	No	resampling/analysis.
Radium	5	5 ^a	pCi/L		No	
Radium-224	5	N/A	pCi/L		No	
				KY/EPA		
Radium-226	5	5 ^a	pCi/L	MCL	No	Exceeded in 2002 only.
				KY/EPA		
Radium-228	5	5 ^a	pCi/L	MCL	No	Not exceeded since 2008.
		4				
Technetium-99	N/A	mrem ^b	pCi/L		No	

N/A = not applicable ^a EPA MCL for radium-226 and 228 combined ^b EPA methodology equates 4 mrem to 900 pCi/L

Table ES.3. Summary of	of RGA Well Screen	ing of Organic	Compound Paramet	ers with Kentucky MCL

	Kentucky	EPA		Exceeds	Exceed since	
Parameter	MCL	MCL	Units	?	2010?	Comments
Benzene	0.005	0.005	mg/L		No	
Carbon Tetrachloride	0.005	0.005	mg/L		No	
1,2-Dichloroethane	0.005	0.005	mg/L		No	
				KY/EPA		Additional evaluation to
Trichloroethene	0.005	0.005	mg/L	MCL	Yes	ascertain source(s).
P-Dichlorobenzene	0.075	0.075	mg/L		No	
1,1-Dichloroethene	0.007	0.007	mg/L		No	
1,1,1-Trichloroethane	0.200	0.200	mg/L		No	
Vinyl Chloride	0.002	0.002	mg/L		No	

Parameter	Kentucky MCL	EPA MCL	Units	Exceeds?	Exceed since 2010?	Comments
PCB-1016	N/A	0.0005	mg/L	EPA MCL	No	
PCB-1221	N/A	0.0005	mg/L		No	
PCB-1232	N/A	0.0005	mg/L		No	
PCB-1242	N/A	0.0005	mg/L	EPA MCL	No	2010 high result not confirmed by resampling/analysis. Additional evaluation of PCBs summarized.
PCB-1248	N/A	0.0005	mg/L	EPA MCL	No	
PCB-1254	N/A	0.0005	mg/L		No	
PCB-1260	N/A	0.0005	mg/L		No	
PCB-1262	N/A	0.0005	mg/L		No	
PCB-1268	N/A	0.0005	mg/L		No	

Table ES.4. Summary of RGA Well Screening of PCBs against EPA MCL

N/A = not applicable

Table ES.5. Summary of RGA Well Screening of Parameters without Kentucky or EPA MCL,
but with Tapwater RSL or EPA Secondary MCL

					Exceed	
	EPA	Tapwater			since	
Parameter	MCL	RSL	Units	Exceeds?	2010?	Comments
1,2,4-						
Trimethylbenzene	N/A	0.015	mg/L		No	
Acetone	N/A	12	mg/L		No	
	SMCL					One RSL exceedance in upgradient
Aluminum	only	16	mg/L	RSL	No	well MW373 in 2002.
						No RSL exceedance; however, 82
						highest results from upgradient
Boron	N/A	3.1	mg/l		No	(MW372, MW373) wells.
Carbon Disulfide	N/A	0.72	mg/l		No	
	SMCL					No RSL. No exceedance of SMCL
Chloride	only	N/A	mg/L		No	of 250 mg/L.
						Concentrations exceed RSL, but are
						not above C-746-U Landfill
						background; additional discussion to
Cobalt	N/A	0.0047	mg/L	RSL	Yes	ascertain source(s).
						500 mg/L (SMCL) exceeded only in
						upgradient MW373 since 2010
	SMCL					(exceed 10 times). Additional
Dissolved Solids	only	N/A	mg/L	SMCL	Yes	discussion to ascertain source(s).
						Above RSL concentrations, but not
	SMCL					above C-746-U Landfill
Iron	only	11	mg/L	RSL	Yes	background; additional discussion.
						Above RSL concentrations, but not
						above C-746-U Landfill
	SMCL					background; additional discussion to
Manganese	only	0.32	mg/L	RSL	Yes	ascertain source(s).
Molybdenum	N/A	0.078	mg/L		No	

					Exceed	
	EPA	Tapwater			since	
Parameter	MCL	RSL	Units	Exceeds?	2010?	Comments
						No exceedance of RSL for nickel
Nickel	N/A	0.300	mg/L		No	soluble salts
						Last exceedance of SMCL
						(250 mg/L) in 2005; above C-746-U
						Landfill background concentrations
						in upgradient well MW372;
	SMCL					additional discussion to ascertain
Sulfate	only	N/A	mg/L	SMCL	No	source(s).
						No exceedance of RSL for
Vanadium	N/A	0.078	mg/L		No	vanadium and compounds.
Zinc	N/A	4.700	mg/L		No	

Table ES.5. Summary of RGA Well Screening of Parameters without Kentucky or EPA MCL, but with Tapwater RSL or EPA Secondary MCL (Continued)

N/A = not applicable

wells from both affected and nonaffected PGDP areas. Thus, the concentrations of cobalt, iron, and manganese are consistent with both the C-746-U Landfill background and concentrations found in other areas of PGDP, although background concentrations may be elevated (as discussed in later sections of this report).

Other constituents (e.g., dissolved solids, sulfate) are below their SMCL; however, some of these constituents have concentrations that are above background. Both of these constituents have above background concentrations in upgradient wells MW372 and MW373. Additional discussion of the potential source(s) of dissolved solids and sulfate is provided.

Table ES.6. summarizes the comparison of constituents without MCLs or RSLs to background and demonstrates that the higher concentrations of calcium, magnesium, potassium, and sodium are more typically found in upgradient well locations, MW372 and MW373. In accordance with the permit, background concentrations were calculated based upon eight quarters of data collected from wells considered to be upgradient. The fact that several constituents have higher concentrations in upgradient wells establishes a background that is higher than concentrations found in other downgradient or sidegradient wells; thus, the C-746-U Landfill does not appear to be a source of statistically quantifiable incremental concentrations of these constituents. Additional discussion is provided that evaluates potential sources. Total organic carbon (TOC) concentrations are consistent with values seen in other wells at PGDP based upon data collected from the Groundwater Operable Unit, as updated during preparation of the Soils Operable Unit Remedial Investigation Report. A few excursions of TOC historically were seen in upgradient wells. There has been no above background concentration of TOC and no excursions since 2010.

PCBS AND THE WELL OVERPUMPING TEST

Beginning in July and October of 2003, groundwater monitoring at the C-746-S, -T, and -U Landfills detected PCBs. Subsequently, PCB levels have declined over time to levels below USEPA MCLs in RGA wells. Because detected concentrations of PCBs have been below USEPA MCLs for eight quarters, a continuing source of PCBs to the RGA wells in the vicinity of the C-746-U Landfill is unlikely. This groundwater assessment evaluated the potential historical source(s) of PCB contamination and

Table ES.6. Summary of RGA Well Screening of List of Parameters without Kentucky or EPA MCL and without Tapwater RSL or EPA Secondary MCL

			Number	Number	
			of	of	
Parameter	Maximum	Units	Samples	Detects	Comments
					Above background concentrations in upgradient
					wells MW372 and MW373. Highest 39 results
					are all in MW373 (upgradient); thus, no
Calcium	85.2	mg/L	542	542	indication of a C-746-U Landfill source.
Chemical					
Oxygen Demand	830	mg/L	537	36	No detections of COD since 2006.
					Infrequently detected. No indication of a
					C-746-U Landfill source because of 4 detects
Iodide	3.2	mg/L	537	6	since 2010, two in upgradient wells.
					Above background concentrations in upgradient
					well MW372 only. Highest 37 results all in
					MW373 (upgradient). No indication of a
Magnesium	32.9	mg/L	542	542	C-746-U Landfill source.
					No above background concentrations; however,
					38 of 52 highest results found in MW373
Potassium	4.34	mg/L	542	515	(upgradient); thus, no C-746-U Landfill source.
					No above background concentrations; however,
					28 of highest 40 results from upgradient wells
					(MW372, MW373, MW369) others from
Sodium	128	mg/L	542	542	MW360.
Thorium-232	ND	pCi/L	215	0	
					No above background concentrations;
Total Organic					concentrations consistent with non-C-746-U
Carbon	324	mg/L	537	263	Landfill PGDP background.

determined that the PCB contamination in the C-746-U Landfill wells may have been the result of crosscontamination associated with well rehabilitation efforts conducted in April and May 2003.

To evaluate the theory that cross-contamination was the source of the earlier exceedances, an overpumping test of four C-746-U Landfill MWs was conducted beginning in March 2011 to stress the groundwater flow into the wells. Well purging associated with the overpumping and the associated (low flow) sampling demonstrated that none of the RGA wells have PCB concentrations greater than the EPA MCL. Although PCBs continue to be detected in a UCRS well in the vicinity of the C-746-U Landfill at concentrations just above the MCL, this report demonstrates that the PCBs in this well are not the result of migration from the C-746-U Landfill and may also be the result of historical cross-contamination during well rehabilitation. Concentrations in the UCRS well likely have not decreased to below the EPA MCL because the well cannot be purged effectively enough to remove the residual PCBs.

CONCEPTUAL MODEL

The conceptual model for the site describes how groundwater flow in the UCRS is vertically downward; thus, groundwater monitoring results from samples taken from the UCRS wells are indicative only of the conditions in the immediate vicinity of that particular well or sourced from affected soils located immediately above/near the well. In evaluating the concentrations in a UCRS well, the location of the well also is considered. For example, the UCRS well with PCB concentrations above the EPA MCL is located well-distal to the working areas of the C-746-U Landfill. Any contaminants identified in this

UCRS well are not representative of any C-746-U Landfill source. None of the constituents¹ present in the UCRS at above benchmark concentrations have a C-746-U Landfill source.

Nevertheless, this report evaluates those constituent/well pairs in both the UCRS and RGA that exceed benchmarks (including background) to potentially identify non-landfill sources of contaminants to the RGA groundwater. Of those constituents with RGA exceedances, there is no indication of a local-to-the-C-746-U Landfill UCRS source. The following are examples:

- TCE has not been detected in UCRS wells above the MCL.
- Only one UCRS well (MW362) had one detection of beta activity (out of 44 samples) above 50 pCi/L. Since that single exceedance in 2001, beta activity has not exceeded 6.77 pCi/L.
- Only sulfate has above background concentrations in UCRS wells MW362 and MW375; however, concentrations are well below the SMCL and also well below the concentrations seen in RGA wells; thus, the UCRS is not the source of the elevated above background sulfate concentrations seen in the RGA in the vicinity of the C-746-U Landfill.

CONCLUSIONS AND RECOMMENDATIONS

This assessment documents that the only constituents with confirmed above MCL concentrations in RGA well samples collected since 2010 are these:

- Beta activity in MW372 (upgradient), MW366, and MW367 exceed the Kentucky MCL of 50 pCi/L; MW366 and MW367 are on the same flow path as the upgradient well MW372. Additional evaluation demonstrates that the beta activity exceedances are attributable to and consistent with an upgradient source that is higher in the upgradient well MW372.
- TCE in MW372 (upgradient), MW373 (upgradient), MW357, MW358, and MW361 has exceeded the Kentucky/EPA MCL of 5 μg/L since January 2010. TCE in upgradient wells MW372 and MW373 is associated with an upgradient source; in addition, TCE in wells located along the northwestern edge of the landfill is associated with a different source located upgradient/crossgradient of the C-746-U Landfill.

NOTE: The most recent quarterly report [*C-746-U Contained Landfill Third Quarter Calendar Year* 2012 (July-September) Compliance Monitoring Report, Paducah Gaseous Diffusion Plant, Paducah, Kentucky (DOE 2012)] documents that only the upgradient wells MW372 and MW373 have TCE concentrations that exceed the MCL.

This assessment documents that the only constituents with confirmed above RSL concentrations (for those constituents without an MCL) in RGA well samples collected since 2010 are cobalt, iron, and manganese; however, none of these constituents has concentrations that are above the C-746-U Landfill background. In addition, the concentrations of cobalt, iron, and manganese are consistent with those found elsewhere in RGA groundwater on and off the PGDP site [as discussed in the Groundwater Operable Unit Feasibility Study (DOE 2001) as updated by the Soils Operable Unit Remedial Investigation Report (DOE 2013)]. Although the presence of these three constituents is consistent with

¹ This statement does not refer to statistical evaluation of field parameters like conductivity and oxidation-reduction potential.

C-746-U background levels and concentrations found at other PGDP locations, some additional discussion is presented regarding the presence of cobalt, iron, and manganese in upgradient well MW373.

This assessment documents that the only RGA constituents² with above background concentrations since 2010 are these:

- Calcium, dissolved solids, magnesium, and sulfate in upgradient well MW372; and
- Calcium and dissolved solids in upgradient well MW373.

The presence of these above background concentrations only in upgradient wells indicates that upgradient sources or other non-C-746-U-Landfill sources are responsible for these increases; and the fact that concentrations downgradient of the C-746-U Landfill are not elevated indicates that the C-746-U Landfill is not a source of statistically quantifiable incremental contamination to the RGA.

Some of the above background constituents identified in the RGA wells (TCE, beta activity) are the result of migration of contaminants through the RGA flowing into the C-746-U Landfill area from upgradient. The upgradient RGA groundwater contains TCE and Tc-99 (a source of beta activity). Although the RGA groundwater associated with the TCE and Tc-99 sources may contribute varying amounts of other constituents (e.g., calcium, dissolved solids, magnesium, sodium, sulfate, etc.), these contributions cannot be differentiated from the natural variation in concentration seen at other PGDP locations. In fact, elevated levels of constituents typically are seen only at upgradient wells MW372 and MW373—further indication of the fact that the C-746-U Landfill is not a source of RGA constituent concentrations.

Potential sources of these constituents are discussed, including upgradient RGA groundwater, incomplete well rehabilitation and biofouling, and natural variation in groundwater chemistry that was not effectively characterized by background monitoring. The potential alternate sources to benchmark exceedances are discussed.

This groundwater assessment report has been developed to satisfy the requirements for completion of the assessment of the C-746-U Landfill. Based on this assessment, one recommendation has been identified, as follows:

• Abandon the open well at the old home site (see Section 4.5.3). This well presents the potential for a contaminant pathway to the UCRS.

² This statement does not refer to statistical evaluation of field parameters like conductivity and oxidation-reduction potential.

1. C-746-U LANDFILL GROUNDWATER BACKGROUND

1.1 C-746-U LANDFILL CONSTRUCTION HISTORY

The U.S. Department of Energy (DOE) owns and operates waste treatment, storage, and disposal units at the Paducah Gaseous Diffusion Plant (PGDP) near Paducah, Kentucky. Three of these units are landfills that are regulated by DOE under the provisions of the Atomic Energy Act, the Commonwealth of Kentucky under the Resource Conservation and Recovery Act, and the Kentucky solid waste landfill regulations.

Three permitted landfills are located on approximately 80 acres of DOE-owned property immediately north of the PGDP. The permitted landfills currently included within this property are these:

- C-746-S Residential Landfill
- C-746-T Inert Landfill
- C-746-U Contained Landfill

The C-746-U Landfill is an operating solid waste landfill located directly north of the C-746-S&T Landfills. Figure 1 presents the location of the C-746-U Landfill relative to PGDP, the S&T Landfills, and the trichloroethene (TCE) plume.

NOTE: These landfills are located well within the Water Policy Box—an area within which exposure to groundwater is limited. The PGDP Action Memorandum for the Water Policy was developed to support Comprehensive Environmental Response, Compensation, and Liability Act projects and outlines how municipal water is supplied to residents located within the Water Policy Box, which extends to all areas in the vicinity of the C-746-U Landfill.

Construction and operation of the C-746-U Landfill was permitted in November 1996 under Solid Waste Permit SW07300045. The permitted C-746-U Landfill area covers about 60 acres and includes a liner and leachate collection system. The C-746-U Landfill began receiving waste in 1997 and continues to receive waste. The Kentucky Division of Waste Management (KDWM) allows for continued operation of the C-746-U Landfill under the combined Solid Waste Landfill Permit SW07300014, SW07300015, SW07300045, effective November 5, 2006. Waste accepted includes construction debris, industrial waste, asbestos material, incinerator ash, cardboard, paper, plastics, and environmental media (e.g., soils). Materials delivered to the landfill may be in bulk form (e.g., soils in dump trucks) or containerized in boxes or drums of varying sizes.

Water that infiltrates the landfill drains to the landfill leachate collection system piping. These liquids drain, via gravity, to a belowground lift station that pumps the leachate into leachate storage tanks on the landfill property. Leachate from the C-746-U Landfill is treated in the on-site leachate treatment facility and the PGDP's C-615 Sewage Treatment Plant and discharged at an outfall in accordance with a Kentucky Pollutant Discharge Elimination System permit.

Twenty-two of the approximately 60 acres of the C-746-U Landfill are designated to be developed for waste disposal, with an ultimate disposal capacity of 1.5 million yd³ of waste materials. The landfill area is divided into 23 phases to accommodate this amount of material. C-746-U Landfill currently is operating in Phases 4 and 5. Phases 1, 2, and most of Phase 3 have a long-term cover.



Figure 1. Location of the U-Landfill Depicted Relative to 2009 RGA TCE Plume Map

The landfill was constructed over a compacted subgrade and liner and leachate collection system that includes these elements (from bottom to top):

- Thirty-six inches of low permeability clay;
- An 80-mil thick polyethylene synthetic liner;
- Twelve inches of drainage material (which contains the leachate collection piping);
- A geotextile fabric; and
- Twelve inches of cushioning material.

The landfill liner system is constructed on a compacted and contoured surface to allow any infiltrating leachate to drain via gravity to the central leachate piping, then toward each cell's east end where it continues to drain to the lift station.

1.2 GROUNDWATER CHARACTERIZATION BEFORE OPERATION

An initial background groundwater characterization was performed in accordance with 401 *KAR* 47:180; 401 *KAR* 48:300, Section 3; and the original Groundwater Monitoring Plan provided in the Technical Permit Application. (See Section 5, Hydrogeologic Setting, for a description of the groundwater systems at the C-746-U Landfill.) Background groundwater characterization analysis was conducted during the course of one year before placement of waste in the landfill. Background groundwater characterization consisted of sampling performed on the original wells (MW269 through MW277). A second background determination was completed, consistent with permit requirements, after the wells located in the vicinity of the C-746-U Landfill were replaced due to corrosion of the wells.

1.3 GROUNDWATER MONITORING PLAN

The groundwater at the C-746-U Landfill is monitored in accordance with the C-746-U Solid Waste Landfill Groundwater Monitoring Plan, Paducah Gaseous Diffusion Plant, Paducah, Kentucky, BJC/PAD-205/R1 (BJC 2001). The groundwater monitoring program included installation of new MWs that were needed because the previously installed wells were corroded. (The well corrosion threatened the integrity of the monitoring system, in part because well corrosion was contributing to increased chromium concentrations and turbidity levels.) Six clusters of wells were installed with corrosion resistant materials. Installation of the wells was completed in 2002. These new wells were sampled to establish background values for each well. These background values are used for comparison each time each well is sampled to identify above background constituent concentrations.

Figure 2 is a map of the C-746-U Landfill and the MWs located in the vicinity of the C-746-U Landfill. Soil boring logs, MW and piezometer water level records, soil geotechnical tests, and groundwater flow models of the area of the C-746-U Landfill provide sufficient data for the development of a Conceptual Site Model (CSM), described in Section 5. In general, groundwater flow is downward through the silts, clays, and fine sands of the Upper Continental Recharge System (UCRS). In contrast, the underlying Regional Gravel Aquifer (RGA) is conductive and provides the main conduit for lateral groundwater flow. Groundwater flow is in a north-northeasterly direction in the vicinity of the C-746-U Landfill. The Ohio River and lower reaches of Little Bayou and Bayou Creeks are the discharge areas for the RGA flow system.



Figure 2. C-746U Landfill and Wells Located in the Vicinity

Consistent with this CSM, the constituent concentrations in UCRS wells are considered to be representative only of the conditions local to the well or sourced from overlying soils; thus, no discussion of potential "upgradient" sources is presented for the UCRS. The constituent concentrations found in the RGA wells are evaluated while considering the potential for contributions of sources located upgradient (with respect to the landfill) of the monitored well.
2. BASIS FOR CONDUCTING ASSESSMENT/PRELIMINARY SCREENING

2.1 HISTORY

In 1999, the KDWM placed the C-746-U Landfill into groundwater contamination assessment, in accordance with 401 *KAR* 48:300, Section 8. The contaminants that exceeded statistical background that caused the C-746-U Landfill to be placed in assessment were chromium, gross beta, technetium-99 (Tc-99), total dissolved solids, total solids, and turbidity. The 1999 groundwater assessment demonstrated that the gross beta and Tc-99 represented a contaminant plume from PGDP that was migrating into the area of the C-746-U Landfill. Chromium, dissolved and total solids, and turbidity were attributed to corrosion of the stainless steel MW casings. As a result of these findings, the corroded MWs were replaced.

No further groundwater assessment was required until a statistically valid set of data had been collected from the new wells to establish background. DOE received a letter from KDWM on August 29, 2006, that again placed the C-746-U Landfill into groundwater contamination assessment. The letter stated that KDWM determined a statistically valid set of data now existed and contaminants had exceeded either maximum contaminant levels (MCLs) or statistical limits calculated relative to background concentrations (calculated in accordance with the permit from eight consecutive quarters of data from wells considered to be located upgradient of the C-746-U Landfill). The contaminants identified by KDWM for the assessment are those discussed in Section 3. KDWM approved the groundwater assessment plan required in accordance with 401 *KAR* 48:300, Section 8, on February 13, 2008. This approval letter is provided as Appendix A.

The data used to support all of the constituent assessments are groundwater analyses of quarterly and semiannual monitoring for the period 2002 through 2012 including the focused sampling of October 2006. In addition, other data were evaluated to support the assessment. For example, soils analyses from a polychlorinated biphenyls (PCBs) source investigation in 2008 and groundwater analyses from an overpumping test during 2011 support the assessment of Total PCBs.

Beginning in July and October of 2003, groundwater monitoring at the C-746-S, -T, and -U Landfills detected PCBs. Subsequently, PCB levels have declined in general, but continue to be detected in several wells. A July–August 2008 field investigation concluded the PCB contamination is not derived from a local contaminated soils source (Appendix D). The overpumping test conducted in 2011 supported this groundwater assessment that has determined the PCB contamination is likely the result of cross-contamination associated with well rehabilitation efforts conducted in April and May 2003 (Appendix E). With the consent of KDWM, LATA Environmental Services of Kentucky, LLC performed an overpumping test of four C-746-U Landfill MWs beginning in March 2011. The data from the overpumping activity (summarized in Appendix F) indicate the PCBs are associated with suspended solids in the wells and can be removed with effective well development. The fact that all of the RGA wells have been rehabilitated indicates that there is no continuing source of PCBs and is consistent with the determination that the contamination was likely the result of historical cross-contamination that occurred during well rehabilitation.

2.2 PURPOSE OF THE ASSESSMENT

In accordance with 401 *KAR* 48:300, Section 8, after the implementation of the groundwater assessment plan, the operator shall submit a groundwater assessment report containing the new data, analysis of the data, and recommendations on the necessity for abatement. This report fulfills this objective. This assessment seeks to identify the presence of constituents present in C-746-U Landfill wells and attribute the source(s) of those exceedances to determine if there is evidence that the C-746-U Landfill is leaking and is the cause of statistically quantifiable contamination.

Tables 1 through 6 summarize the results of the evaluation of the parameters under this assessment. This assessment includes an evaluation of a broad range of parameters, not just those identified for assessment. The following are the parameters required to be evaluated under the original assessment (August 2006).

- Gross Beta
- Carbon Disulfide
- Chemical Oxygen Demand
- Manganese
- Sulfate
- Tc-99
- Total PCBs
- Radium-228
- Total Organic Carbon (TOC)
- TCE

The following are the additional parameters added to the assessment, identified based on focused sampling conducted in October 2006.³

- Calcium
- Chloride
- Cobalt
- Dissolved Solids
- Iron
- Magnesium
- Sodium
- Uranium

2.3 SUMMARY OF THE ASSESSMENT SCREENING

Tables 1–6 list the parameters whose concentrations were screened against benchmarks (including background) to identify those that needed additional discussion. The current and historical analytical results from RGA wells located in the vicinity of the C-746-U Landfill were screened against benchmarks, including Kentucky MCLs, U.S. Environmental Protection Agency (EPA) MCLs, and EPA regional screening levels (RSLs). Subsequently, those constituents without MCLs or RSLs were evaluated against EPA secondary MCLs (SMCLs) and historical background levels (calculated in

³ The DOE evaluation following the October 2006 focused sampling event also identified additional wells for assessment of some of the original parameters specified by KDWM in August 2006 (manganese, Total PCBs, sulfate, Tc-99, and TOC).

accordance with the C-746-U Landfill permit). Each result was compared against each benchmark concentration and any exceedances of the benchmarks were tabulated with a special focus on those constituents that exceeded the benchmark over the past eight calendar quarters.

The screening results presented in Tables 1–6 document those constituents that have not had a confirmed exceedance of any benchmark in any RGA well. The lack of any current or historical exceedance of a benchmark in an RGA well resulted in this constituent being screened from further evaluation.

The results of the screening are presented in the following sections. The evaluation continued for those constituents that have had an RGA exceedance of a benchmark with particular focus on those constituents that had a confirmed exceedance of a benchmark over the past eight calendar quarters. This subsequent evaluation into the source(s) of these constituents looked at the following:

- Spatial patterns of each constituent present above a benchmark;
- Identified plumes;
- Concentrations of these constituents in other PGDP wells; and
- Concentrations of these constituents in landfill leachate.

2.3.1 Inorganic Parameters Screening against Kentucky or EPA MCL

Table 1 provides a summary of the screening of C-746-U Landfill data for inorganic parameters in RGA wells against Kentucky or U.S. Environmental Protection Agency (EPA) MCL for all inorganic constituents that have an MCL. This table shows that antimony, arsenic, barium, beryllium, cadmium, chromium, copper, lead, mercury, selenium, silver, uranium, nitrate, and fluoride have not exceeded either the Kentucky or EPA MCL since January 2010. Thus, the landfill is in compliance with respect to these parameters.

2.3.2 Radionuclide Parameters Screening against Kentucky or EPA MCL

Table 2 provides a summary of the screening of C-746-U Landfill data for radionuclide parameters in RGA wells against Kentucky or EPA MCLs. This table shows that alpha activity, strontium-90, tritium, iodine-131, radium, radium-224, radium-226, radium-228, and Tc-99 have not exceeded the respective MCL since January 2010. Thus, the C-746-U Landfill is in compliance with respect to these parameters. Additional discussion of beta activity is required because water from some of the wells has exceeded the Kentucky MCL for beta activity of 50 pCi/L since 2010. Some additional discussion of Tc-99 is included because Tc-99 is a source of beta activity and is present in RGA groundwater at locations upgradient to the C-746-U Landfill.

2.3.3 Organic Parameters Screening against Kentucky or EPA MCL

Table 3 provides a summary of the screening of C-746-U Landfill data for organic parameters in RGA wells against Kentucky or EPA MCLs. This table shows that benzene, carbon tetrachloride, 1,2-dicholoroethane, p-dichlorobenzene, 1,1-dichloroethene, 1,1,1-trichloroethane, and vinyl chloride never have been detected in C-746-U Landfill wells and have not exceeded the respective MCL since January 2010. Thus, the C-746-U Landfill is in compliance with respect to these parameters. In addition, none of these constituents ever have been detected in wells located in the vicinity of the C-746-U Landfill.

Parameter	Kentucky MCL	EPA MCL	Maximum Detected Conc.	Units	Number of Samples	Number of Detects	Number Exceed MCL	Exceeds?	Exceed since 2010?	Comments
Antimony	N/A	0.006	ND	mg/L	542	0	0		No	
Arsenic	0.05	0.01	0.0146	mg/L	542	268	12	EPA MCL	No	No exceedance of KY MCL; last EPA MCL exceedance 2004.
Barium	2.0	2.0	1.22	mg/L	555	555	0		No	
Beryllium	N/A	0.004	0.00497	mg/L	542	1	1	EPA MCL	No	Only EPA exceedance 2003.
Cadmium	0.005	0.005	0.00511	mg/L	542	6	1	KY/EPA MCL	No	Only exceedance 2003.
Chromium	0.100	0.100	0.025	mg/L	554	1	0		No	Only detection in 2002; no exceedance.
Copper	N/A	1.3	0.026	mg/L	542	4	0		No	
Lead	0.05	0.015	0.0127	mg/L	542	16	0		No	
Mercury	0.002	0.002	ND	mg/L	540	0	0		No	
Selenium	0.05	0.05	0.0125	mg/L	542	154	0		No	
Silver	0.05	N/A	0.00146	mg/L	542	1	0		No	
Uranium	N/A	0.030	0.029	mg/L	546	9	0		No	
Nitrate (as N)	10.0	10.0	5	mg/L	511	94	0		No	
Fluoride	4.0	4.0	2.6	mg/L	537	511	0		No	

Table 1. Summary of RGA Well Screening of List of Inorganic Parameters with Kentucky or EPA MCL

N/A = not applicable ND = not detected at concentration less than the respective MCL

Parameter	Kentucky MCL	EPA MCL	Maximum Detected Conc.	Units	Number of Samples	Number of Detects	Number Exceed MCL	Exceeds?	Exceed Since 2010?	Comments
Alasha Antinita	15	15	10	лC:/I	549	40	1	KY/EPA	Na	Order encodernes in 2004
Beta Activity	50	4 mrem	137	pCi/L	548	40	60	KY MCL	Yes	12 KY MCL exceedances since 2010; additional evaluation needed to ascertain source(s).
Strontium-90	8	N/A	7.57	pCi/L	537	1	0		No	Only exceedance in 2003.
Tritium	20,000	N/A	641	pCi/L	536	1	0		No	No exceedance; only detection in 2007.
Iodine-131	3	N/A	10.6	pCi/L	514	1	1	KY MCL	No	Only result (7/14/2010) greater than minimum detectable activity (MDA) not confirmed by re- sampling/analysis.
Radium	5	5 ^a	1.04	pCi/L	128	3	0		No	
Radium-224	5	N/A	Below MDA	pCi/L	77	0	0		No	
Radium-226	5	5 ^a	12.4	pCi/L	486	31	1	KY/EPA MCL	No	Exceeded in 2002 only.
Radium-228	5	5 ^a	8.63	pCi/L	302	14	4	KY/EPA MCL	No	Not exceeded since 2008.
Technetium-99	N/A	4 mrem ^b	179	pCi/L	561	352	0		No	

Table 2. Summary of RGA Well Screening of Radionuclide Parameters with Kentucky or EPA MCL

N/A = not applicable ^a EPA MCL for radium-226 and 228 combined. ^b EPA methodology equates 4 mrem to 900 pCi/L.

			Maximum		Number	Number	Number		Exceed	
	Kentucky	EPA	Detected		of	of	Exceed		since	
Parameter	MCL	MCL	Conc.	Units	Samples	Detects	MCL	Exceeds?	2010?	Comments
Benzene	0.005	0.005	ND	mg/L	559	0	0		No	
Carbon Tetrachloride	0.005	0.005	ND	mg/L	559	0	0		No	
1,2-Dichloroethane	0.005	0.005	ND	mg/L	559	0	0		No	
								KY/EPA		Additional evaluation needed to
Trichloroethene	0.005	0.005	0.022	mg/L	559	325	124	MCL	Yes	ascertain source(s).
p-Dichlorobenzene	0.075	0.075	ND	mg/L	554	0	0		No	
1,1-Dichloroethene	0.007	0.007	ND	mg/L	559	0	0		No	
1,1,1-Trichloroethane	0.200	0.200	ND	mg/L	559	0	0		No	
Vinyl Chloride	0.002	0.002	ND	mg/L	559	0	0		No	

Table 3. Summary of RGA Well Screening of Organic Compound Parameters with Kentucky MCL

ND = not detected at concentration less than the respective MCL

Table 4. Summarv	of RGA	Well Screeni	ng of PCBs	against E	PA MCL

	V an 4m alm	EDA	Maximum		Number	Namehan	Number		Exceed	
Parameter	MCL	MCL	Conc.	Units	of Samples	of Detects	MCL	Exceeds?	2010?	Comments
PCB-1016	N/A	0.0005	0.0029	mg/L	465	50	9	EPA MCL	No	
PCB-1221	N/A	0.0005	ND	mg/L	465	0	0		No	
PCB-1232	N/A	0.0005	0.00045	mg/L	465	1	0		No	
										2011 result of 0.000592 in MW363 not confirmed with resampling/
PCB-1242	N/A	0.0005	0.00115	mg/L	465	42	8	EPA MCL	No	reanalysis.
PCB-1248	N/A	0.0005	0.00062	mg/L	465	3	1	EPA MCL	No	
PCB-1254	N/A	0.0005	ND	mg/L	465	0	0		No	
PCB-1260	N/A	0.0005	0.00019	mg/L	465	3	0		No	
PCB-1262	N/A	0.0005	ND	mg/L	11	0	0		No	
PCB-1268	N/A	0.0005	ND	mø/L	46	0	0		No	

N/A = not applicable ND = not detected at concentration less than the respective MCL

				Tapwater			Number	Number			Exceed	
	Parameter	Kentucky MCL	EPA MCL	RSL (No MCL)	Maximum	Units	of Samples	of Detects	Number Exceed RSL	Exceeds?	since 2010?	Comments
	1.2.4-	MCL	MOL		Witter	Cints	Sumples	Dettetts	LACCCU RDL	LACCCU5:	2010.	Comments
	Trimethylbenzene	N/A	N/A	0.015	0.00011	mg/L	7	1	0		No	
	Acetone	N/A	N/A	12	11	mg/L	559	48	0		No	
			SMCL									One RSL exceedance in upgradient
	Aluminum	N/A	only	16	22.7	mg/L	542	108	1	RSL	No	MW373 in 2002.
	Boron	N/A	N/A	3.1	1.95	mg/L	540	154	0		No	
	Carbon Disulfide	N/A	N/A	0.72	0.025	mg/L	559	19	0		No	
			SMCL	NT / A	120		511	511	NT/A		N.	No RSL; no exceedances of SMCL of
ł	Chloride	N/A	only	N/A	130	mg/L	511	511	N/A		No	250 mg/L.
												not above C-746-UL and fill
	Cobalt	N/A	N/A	0.0047	0.86	mg/L	542	341	191	RSL	Yes	background: additional evaluation.
Ì							_	_				No RSL. 500 mg/L (SMCL) exceeded
												only in upgradient MW373 since 2010
E			SMCL								No/	(exceed 10 times); additional
	Dissolved Solids	N/A	only	N/A	1200	mg/L	537	469	N/A	SMCL	Yes	evaluation.
												RSL exceedances, but concentrations
	-	27/1	SMCL		<i>(</i>))	~		4.40		DGI		not above C-746-U Landfill
	Iron	N/A	only	11	62.2	mg/L	542	448	64	RSL	Yes	background; additional evaluation.
												RSL exceedances, but concentrations
	Management	NT/A	SMCL	0.22	24.2		540	520	246	DCI	V	not above C-746-U Landfill
	Manganese	N/A	only	0.32	24.2	mg/L	542	530	246	KSL	Yes	background; additional evaluation.
	Molybdenum	N/A	N/A	0.078	0.00392	mg/L	540	4	0		No	
	Nielial	NI/A	NI/A	0.200	0.0720	m a /I	540	05	0		No	No exceedances of RSL for nickel
	NICKEI	IN/A	IN/A	0.500	0.0729	mg/L	342	95	0		INO	No BSL : last avagadance of SMCI
												(250 mg/L) in 2005: above background
			SMCI									concentrations in ungradient well
	Sulfate	N/A	only	N/A	809.6	mg/L	537	526	N/A	SMCL	No	MW372.
Ì												No exceedance of RSL for vanadium
	Vanadium	N/A	N/A	0.078	0.048	mg/L	542	20	0		No	and compounds.
	Zinc	N/A	N/A	4.700	0.725	mg/L	542	17	0		No	

Table 5. Summary of RGA Well Screening of Parameters without Kentucky or EPA MCL, but with Tapwater RSL or EPA Secondary MCL

N/A = not applicable

			Tapwater			Number	Number	Number		Exceed	
D (Kentucky	EPA	RSL (No	м ·	T T •4	of	of	Exceed	Exceeds	since	
Parameter	MCL	MCL	MCL)	Maximum	Units	Samples	Detects	MCL	?	2010?	Comments
											Above background concentrations only in ungradient wells MW372 and
											MW373 Highest 39 results all in
											MW373 (upgradient); thus, no indication
Calcium	N/A	N/A	N/A	85.2	mg/L	542	542	N/A	N/A	N/A	of a C-746-U Landfill source.
Chemical											
Oxygen											
Demand (COD)	NI/A	NI/A	NI/A	820	ma/I	527	26	NI/A	NI/A	NI/A	No detections of COD since 2006
(COD)	IN/A	IN/A	IN/A	830	nig/L	557	30	IN/A	IN/A	1N/A	Indide infrequently detected No apparent
											pattern to detections. Of 4 detects since
											2010, two in upgradient wells; thus, no
Iodide	N/A	N/A	N/A	3.2	mg/L	537	6	N/A	N/A	N/A	apparent C-746-U Landfill contribution.
											Above background concentrations only
											in upgradient MW372. Highest 37 results
Magnesium	N/A	N/A	N/A	32.9	mg/I	542	542	N/A	N/A	N/A	indication of a C-746-U I and fill source
magnesium	10/11	14/21	10/21	52.9	iiig/12	512	512	14/11	11/21	10/21	Not above C-746-U Landfill background:
											however, 38 of 52 highest results in
											upgradient well MW373; no apparent
	NT / A	NT / A		4.24	π	5.40	515	NT / A		NT/A	pattern to detections that indicates a
Potassium	N/A	N/A	N/A	4.34	mg/L	542	515	N/A	N/A	N/A	C-746-U Landfill contribution .
											Of highest 40 results 28 from upgradient
											(MW373, MW372, MW369) others from
											MW360; thus, no apparent pattern in
											detections that indicates a
Sodium	N/A	N/A	N/A	128	mg/L	542	542	N/A	N/A	N/A	C-746-U Landfill source.
Thorium-	NI/A	NT/A	NI/A	ND	тC:Л	215	0	NI/A	NI/A	NI/A	
Z3Z Total	IN/A	IN/A	IN/A	ND	pC1/L	215	U	IN/A	IN/A	IN/A	
Organic											Values consistent with other PGDP
Carbon	N/A	N/A	N/A	324	mg/L	537	263	N/A	N/A	N/A	areas; additional discussion.

Table 6. Summary of RGA Well Screening of List of Parameters without Kentucky or EPA MCL and without Tapwater RSL or EPA Secondary MCL

N/A = not applicable

Additional discussion of the presence of TCE is required because water from some of the wells has TCE concentrations that have exceeded the MCL of 5 μ g/L since January 2010. Additional discussion is also provided because of the presence of TCE in RGA groundwater at locations upgradient to the C-746-U Landfill.

2.3.4 PCBs Screening against EPA MCL

Table 4 provides a summary of the screening of C-746-U Landfill data for PCBs in RGA wells against the EPA MCL (Kentucky does not have an MCL). This table shows that none of the PCBs have had a confirmed result above the EPA MCL since 2010. In addition, no PCBs have had above background concentrations in RGA wells since January 2010. Thus, the C-746-U Landfill is in compliance with respect to these parameters. However, additional discussion about the presence of PCBs is presented in later sections of the report to explain the historical findings and the continued presence of PCBs above the EPA MCL in a UCRS well located distal to the C-746-U Landfill.

2.3.5 Parameters Screening against EPA RSLs and Secondary MCLs

Table 5 provides a summary of the screening of C-746-U Landfill data for parameters that do not have an MCL but do have an RSL or an SMCL. Of these parameters, only cobalt, iron, and manganese have exceeded the RSL since 2010; however, none of these constituents exhibits a concentration that is above background. There is no indication that the C-746-U Landfill is a source of constituents that cause an exceedance of the RSL for any of these parameters; however, some additional discussion of the source(s) of these constituents is presented.

Dissolved solids concentrations have exceeded the SMCL since January 2010; however, this exceedance is only in an upgradient well (MW373). Dissolved solids exhibit above background concentrations in two upgradient wells, MW372 and MW373. Additional discussion is provided on the potential source(s) of dissolved solids. Concentrations of sulfate have not exceeded the SMCL since January 2010; however, the concentrations of sulfate in several wells approach the SMCL, and upgradient well MW372 has above background concentrations. Chloride concentrations do not exceed the SMCL. Additional discussion is provided to account for the presence of sulfate, chloride, and dissolved solids at found concentrations.

2.3.6 Parameters without MCLs or RSLs

Table 6 provides a summary of the observations for C-746-U Landfill data for parameters that do not have an MCL or RSL for comparison. Because there is no MCL or RSL for comparison, additional discussion of these parameters (e.g., dissolved solids, sulfate, potassium, magnesium, boron, sodium) is included where that discussion potentially informs discussion of other parameters. For example, the highest 39 calcium concentrations all are found in upgradient well MW373. Thus, calcium in MW373 is discussed in the context of other constituents that have high concentrations in that well (e.g., dissolved solids, sulfate, potassium, magnesium, boron, sodium).

Although field parameters (e.g., oxidation reduction potential) are subjected to quarterly statistical analysis, the finding of above background values for field parameters are only discussed in this assessment where these findings are related to potential source(s) of other parameters.

2.3.7 Assessment Parameters Summary

Table 7 provides a summary of the observations of C-746-U Landfill data for parameters that were included for assessment using the screening summarized above to further evaluate these parameters.

Parameter	Wells (Completion Unit)	Conclusion
Gross Beta MW372 [Upgradient Upper Regional Gravel Aquifer (URGA)] and MW373 [Upgradient Lower Regional Gravel Aquifer (LRGA)]		Gross beta exceedances are further discussed in later sections of the report. Elevated beta levels apparently are coming from upgradient sources and are directly related to the Tc-99 that is migrating into the area.
Carbon Disulfide	MW362 (UCRS)	Concentration less than RSL since 2010. No further discussion.
Chemical Oxygen Demand	MW358 (LRGA)	No detectable COD since 2010. No further discussion.
Manganese	MW366 (URGA)	Manganese levels have been below the RSL in this well since 2010; however, other RGA wells have above RSL concentrations that are consistent with results in other PGDP RGA wells. Additional discussion is included.
	MW359 (UCRS), MW365 (UCRS)	Sulfate levels are less than the SMCL. Additional discussion of the source(s) of sulfate is included.
Sulfate	MW362 (UCRS)	Sulfate levels exceeded the SMCL one time (April 2003). Concentrations no longer exceed the SMCL. Additional discussion of the source(s) of sulfate is included.
Technetium-99	MW361 (LRGA), MW372 (URGA), MW373 (LRGA)	Tc-99 levels are less than the constituent- specific derived EPA MCL. Although less than the MCL, Tc-99 is a source of beta activity. Beta levels apparently are related to Tc-99 migrating into the area and are further discussed as a source of beta activity.
PCB, Total	MW361 (LRGA), MW363 (URGA), MW365 (UCRS)	PCB levels in MW361 and MW363 no longer exceed the EPA MCL for total PCBs. Thus, there is no current issue with the PCB concentrations. The historical source of the PCBs may be (2003) cross-contamination related to well rehabilitation activities. MW365 concentrations still exceed the MCL, in part, because the well cannot be completely rehabilitated due to low rates of purging/surging allowed by the UCRS well.
Radium-228	MW362 (UCRS), MW363 (URGA), MW364 (LRGA), MW369 (URGA)	Radium-228 levels do not exceed the MCL in the RGA. The last time MW362 had an exceedance was 1/5/2010; the last eight quarters have not had an exceedance.
ТОС	MW358 (LRGA)	TOC levels no longer exceed background levels and have not since 2010. No further discussion.

Table 7. Summary of Conclusions for Assessment Parameters in All Wells Based on Screening

Table 7. Summary of Conclusions for Assessment Parameters in All Wells Based on Screening (Continued)

Parameter	Wells (Completion Unit)	Conclusion
Trichloroethene	MW370 (LRGA), MW372 (URGA), MW373 (LRGA)	TCE levels exceed the MCL. Additional discussion of the source(s) of TCE is presented that shows TCE in RGA groundwater at locations upgradient of the C- 746-U Landfill is the likely source of the TCE in C-746-U Landfill wells. No UCRS wells have TCE MCL exceedances.

2.3.8 Additional Assessment Parameters Summary

Table 8 provides a summary of the observations of C-746-U Landfill data for parameters that were included for assessment using the screening summarized above to further evaluate these parameters.

2.4 CONCLUSIONS OF THE ASSESSMENT SCREENING

Only two constituents (beta activity and TCE) are present in concentrations in RGA groundwater that are above MCLs and above background, and the exceedances of the MCLs for these parameters are apparently sourced from upgradient of the C-746-U Landfill, as discussed subsequently in this document.

Some additional discussion of the apparent historical source of PCBs (well rehabilitation) is presented; however, irrespective of potential sources, there have been no confirmed exceedances of the EPA MCL in RGA wells since January 2010.

Parameter	Wells (Completion Unit)	Conclusion
	MW371 (UCRS)	Calcium no longer has a statistical exceedance in MW371.
Calcium	MW372 (URGA) and MW373 (LRGA)	Calcium does not have an MCL or RSL. Additional discussion is provided in tables and text in subsequent sections of this report that notes that the highest 39 calcium concentrations are all in upgradient well MW373; thus, the C-746- U Landfill is not the source of the above background calcium. Potential alternate sources are discussed.
Chloride	MW374 (UCRS)	Chloride levels are not above background and are less than the SMCL.
Cobalt	MW369 (URGA)	Cobalt concentrations exceed the RSL; however, additional discussion is provided that indicates that concentrations are consistent with C-746-U background and concentrations of cobalt in other PGDP areas.

Table 8. Summary of Conclusions for Additional Assessment Parameters in All Wells Based on Screening

Table 8. Summary of Conclusions for Additional Assessment Parameters in All Wells Based on Screening (Continued)

Parameter	Wells (Completion Unit)	Conclusion
	MW362 (UCRS)	Dissolved solids levels are below the SMCL since 2010.
Dissolved Solids	MW372 (URGA) and MW373 (LRGA)	Dissolved solids levels in MW372 are less than the SMCL, but exceed the SMCL in MW373 since 2010. Both MW372 and MW373 have dissolved solids concentrations that show concentrations above background. Only upgradient well, MW373, has had levels over the SMCL; thus, the C-746-U Landfill cannot be the primary source of dissolved solids.
	MW374 (UCRS)	Dissolved solids levels have remained below the SMCL since October 2006.
Iron	MW369 (URGA)	Iron levels do not exceed the RSL in this well. Additional discussion of iron is provided to document that iron concentrations are within the range of background, though some alternate sources may be contributing incremental amounts.
Iron	MW374 (UCRS)	Iron levels do not exceed the RSL in this well. Additional discussion of iron is provided to document that iron concentrations are within the range of background, though some alternate sources may be contributing incremental amounts.
Magnasium	MW358 (LRGA) and MW371 (UCRS)	No MCL or RSL. The highest 44 sample concentrations are associated with samples from upgradient wells MW372 and MW373; thus, the C-746-U Landfill is not the likely source of high magnesium levels. Additional discussion of the potential sources of magnesium concentrations in these wells is provided.
Magnesium	MW372 (URGA) and MW373 (LRGA)	No MCL or RSL. The highest 44 concentrations are associated with samples from upgradient wells MW372 and MW373; thus, the C-746-U Landfill is not the likely source of high magnesium levels. Additional discussion of the potential sources of magnesium concentrations in these wells is provided.
Manganese	MW367 (LRGA)	Manganese levels exceed the RSL since 2010. Concentrations in this well are not above background. Some additional discussion of the mechanisms of manganese release from native materials is provided.
manganese	MW369 (URGA)	Manganese levels do not exceed the RSL over the past eight quarters. Concentrations in this well are not above background. Some additional discussion of the mechanisms of manganese release from native materials is provided.

Table 8. Summary of Conclusions for Additional Assessment Parameters in All Wells Based on Screening (Continued)

Parameter	Wells (Completion Unit)	Conclusion
		Manganese levels do not exceed the RSL since
Manganese	MW274 (LICDS)	2010. Some additional discussion of the
(Continued)	MW 3/4 (UCKS)	mechanisms of manganese release from native
		materials is provided.
		Total PCB concentrations do not exceed the
		EPA MCL since 2010. Additional discussion of
PCB, Total	MW366 (URGA)	the likely source of the historical contamination
		is provided. The C-746-U Landfill was not the
		source of the historical contamination.
		No MCL. Higher sodium concentrations more
		frequently seen in upgradient wells MW372 and
		MW373; thus, the pattern of higher
	MW 500 (UKGA)	concentrations is not consistent with a C-746-U
		Landfill source. Additional discussion of the
		potential source(s) is provided.
		No MCL. Higher sodium concentrations are seen
		more frequently in upgradient wells MW372 and
		MW373; thus, the pattern of higher
	MW 369 (UKGA)	concentrations is not consistent with a C-746-U
		Landfill source. Additional discussion of the
		potential source(s) is provided.
Sodium		No MCL. A UCRS source is not needed to
		explain RGA well exceedances. Higher sodium
		concentrations are seen frequently in upgradient
	MW362 (UCRS)	wells MW372 and MW373: thus, the pattern of
	× ,	higher concentrations is not consistent with a C-
		746-U Landfill source. Additional discussion of
		the potential source(s) is provided.
		No MCL. Higher sodium concentrations are seen
		more frequently in upgradient wells MW372 and
		MW373; thus, the pattern of higher
	MW372 (URGA)	concentrations is not consistent with a C-746-U
		Landfill source. Additional discussion of the
		potential source(s) is provided.
		Sulfate levels are less than the SMCL; however,
		concentrations approach the SMCL and the
		highest sulfate concentration in C-746-U
Sulfate	MW372 (URGA)	Landfill wells is associated with upgradient well
		MW373. A major C-746-U Landfill source is
		unlikely: however, additional discussion of the
		potential source(s) of sulfate is provided.
		Tc-99 levels are less than the constituent-
		specific derived EPA MCL. Additional
T 1		discussion is provided because Tc-99 is a source
Technetium-99	MW357 (URGA)	of beta activity, and beta activity may be related
		to the Tc-99 present in upgradient RGA
		locations.

Table 8. Summary of Conclusions for Additional Assessment Parameters in All Wells Based on Screening (Continued)

Parameter	Wells (Completion Unit)	Conclusion
ТОС	MW374 (UCRS) and MW362 (UCRS)	No MCL. Concentrations below background levels. Minimal additional discussion of the TOC levels is provided.
Uranium	MW362 (UCRS)	Uranium levels are less than the MCL.
	MW373 (LRGA)	Uranium levels are less than the MCL.

Some constituents do not have an MCL, but do have an RSL or an SMCL. Of these parameters, only cobalt, dissolved solids, iron, and manganese have exceeded the RSL since January 2010; however, the concentrations of these constituents are consistent with C-746-U background and with concentrations found in the RGA at other locations in the vicinity of PGDP. Thus, there is no indication of a statistically quantifiable C-746-U Landfill source of above RSL concentrations of these parameters; however, additional discussion of the source(s) of these constituents is presented.

Constituents with concentrations that are elevated above the C-746-U Landfill background are associated with upgradient RGA wells MW372 and MW373; thus, non-C-746-U Landfill sources for these constituents are apparent. Additional discussion of parameters without RSLs or MCLs is presented to identify and discuss these alternate sources. For example, the highest 39 calcium concentrations all are found in upgradient well MW373. Thus, calcium in MW373 is discussed in the context of other constituents that have high concentrations in that well (e.g., sulfate, potassium, magnesium, boron, sodium).

Although field parameters are subjected to statistical analysis as part of the quarterly monitoring, the finding of above background concentrations of field parameters only is discussed in this assessment report where their presence sheds light on the source(s) of other parameters.

Dissolved solids concentrations have exceeded the SMCL since January 2010; however, this exceedance is only in an upgradient well. Additional discussion is provided on the potential source(s) of dissolved solids. Concentrations of sulfate have not exceeded the SMCL since January 2010; however, the concentrations of sulfate in several wells have concentrations that approach the SMCL. Chloride concentrations do not exceed the SMCL. Additional discussion is provided as to the potential source(s) of sulfate, chloride, and dissolved solids.

3. DISCUSSION OF CONSTITUENTS NOT ELIMINATED BY SCREENING

This section provides an assessment of constituents present in C-746-U Landfill wells since 2010 at these concentrations:

- Above an MCL, including beta activity and TCE;
- Above an RSL, including cobalt, iron, and manganese;
- Below an RSL, but with concentrations that indicate the potential for a C-746-U Landfill or alternate source, including dissolved solids and sulfate; and
- Other constituents that may indicate the potential for a C-746-U Landfill or alternate source, including, calcium, dissolved solids, magnesium, and sodium.

This section provides a summary of the physical properties of each contaminant, the contaminant concentrations in C-746-U Landfill wells and other sample locations relative to benchmarks, and the conclusions regarding each contaminant.

3.1 BETA ACTIVITY

3.1.1 Physical Properties

Beta particles are subatomic particles ejected from some radioactive atoms. Beta particle emission occurs when a neutron transforms into a proton and a beta particle. The process decreases the number of neutrons in the nucleus by one and increases the number of protons by one. The beta emission changes the radionuclide to a different element. Often, gamma ray emission accompanies the emission of a beta particle.

Gross beta activity has been used as an indicator of beta-emitting radionuclides in water since at least the early 1950s. It is a measurement of all beta activity present, regardless of the specific radionuclide source. Gross measurements are used as a method to screen samples for relative levels of radioactivity.

There are many beta activity emitters. Common sources of beta activity in groundwater are potassium-40 and radium-228. Other beta activity sources include cesium-137, cobalt-60, iodine-129, strontium-90, Tc-99 (the primary source of beta activity in groundwater at PGDP), and tritium.

Direct exposure to beta particles is a hazard; however, emissions from inhaled or ingested beta particle emitters are the greatest concern. Some beta-emitters, such as carbon-14, distribute widely throughout the body. Others accumulate in specific organs and cause chronic exposures. Examples include iodine-131 (concentrates heavily in the thyroid gland) and strontium-90 (accumulates in bone and teeth).

3.1.2 Gross Beta Particle Activity C-746-U Landfill

Gross beta activity exceeds the Kentucky MCL of 50 pCi/L in groundwater samples from MW372—an RGA well located upgradient of the C-746-U Landfill. A plot of data points from this well is shown in Appendix B. The latest quarterly assessments also have identified gross beta activity above the MCL in wells MW366 and MW367. These wells are on the same RGA flow path as MW372. Various graphs of

the gross beta activity of these wells since 2010 are provided in Appendix B. These graphs demonstrate a historical source of beta activity from upgradient of the C-746-U Landfill; in addition, the graphs also show that the activity of that source is decreasing. A graph of beta activity (as a 4-event rolling average) presented in Appendix B on a semilog scale, in conjunction with the other observations, indicates that the C-746-U Landfill is not the source of a statistically quantifiable contribution to the beta activity results.

Tc-99 is a beta emitter. There is Tc-99 present in RGA groundwater at the PGDP site, as shown in Figure 3 (plume map established relative to the derived EPA MCL of 900 pCi/L). NOTE: Figure 3 also demonstrates that the C-746-U Landfill is located well within the boundaries of the Water Policy Box.



Figure 3. Location of C-746-U Landfill Relative to 2009 Tc-99 RGA Plume Map and Water Policy Boundary

Because Tc-99 can contribute to beta activity and also is measured in pCi/L, Figure 4 shows the Tc-99 plume contours to 100 pCi/L.

Figure 4 shows that the C-746-U Landfill RGA wells are located downgradient of a Tc-99 plume that originates from facilities upgradient of the C-746-U Landfill associated with the industrial area of PGDP. The maximum concentration of beta activity was seen in C-746-U Landfill well MW372 in 2007, a well located upgradient of the C-746-U Landfill.

Figures 5 through 15 show that beta activity is well correlated to Tc-99 in wells located in the vicinity of the C-746-U Landfill. These figures show that beta activity increases as Tc-99 concentrations increase for wells located in the vicinity of the C-746-U Landfill. The beta activity source is shown to be completely explained by the upgradient Tc-99 source. Section 4 presents additional information on what is known about the Tc-99/beta activity source(s) that contributes to the beta activity found in the RGA C-746-U Landfill wells.

3.2 TCE

3.2.1 Physical Properties

TCE is a volatile organic compound used mainly as a solvent in industrial degreasing and for cleaning metals, but it also is used as a solvent for waxes, fats, resins, oils, and in numerous other applications. Prior to 1977, TCE had been used as an anesthetic, grain fumigant, disinfectant, and extractant of spice oleoresins in food and of caffeine in the production of decaffeinated coffee.

PGDP used large volumes of TCE, primarily as a degreasing agent; however, use of TCE at PGDP ceased July 1, 1993, a time that is prior to construction of the C-746-U Landfill.

3.2.2 Mobility of TCE in the Terrestrial Environment

TCE is the primary contaminant in RGA groundwater at PGDP. The C-400 Cleaning Building is the largest source area of TCE contamination to groundwater at PGDP. TCE in the vicinity of the C-746-U Landfill migrates with the groundwater.

3.2.3 TCE Concentrations at C-746-U Landfill

Prior to operation of the C-746-U Landfill, TCE contamination was observed in groundwater samples in the vicinity of the C-746-U Landfill; however, the concentrations in just a few of the wells are occasionally just above the MCL of 5 μ g/L. Migration of TCE occurs only in conjunction with migration of groundwater. Plots of TCE data over time for wells with concentrations above the MCL are shown in Appendix B.

3.2.4 TCE at the C-746-U Landfill

Three of the upgradient C-746-U Landfill MWs historically have yielded water with TCE levels greater than the MCL (5 μ g/L): MW370 (LRGA) near the southwest corner of the C-746-U Landfill permit area and MW372 (URGA) and MW373 (LRGA) at the southeast corner of the C-746-U Landfill permit area.



Figure 4. Technetium-99 Contours for the RGA



Figure 5. Technetium-99 versus Beta Activity in MW372 at the C-746-U Landfill



Figure 6. Technetium-99 versus Beta Activity in MW373 at the C-746-U Landfill



Figure 7. Technetium-99 versus Beta Activity in MW358 at the C-746-U Landfill



Figure 8. Technetium-99 versus Beta Activity in MW366 at the C-746-U Landfill



Figure 9. Technetium-99 versus Beta Activity in MW367 at the C-746-U Landfill



Figure 10. Technetium-99 versus Beta Activity in MW369 at the C-746-U Landfill



Figure 11. Technetium-99 versus Beta Activity in MW361 at the C-746-U Landfill



Figure 12. Technetium-99 versus Beta Activity in MW364 at the C-746-U Landfill



Figure 13. Technetium-99 versus Beta Activity in MW365 at the C-746-U Landfill



Figure 14. Technetium-99 versus Beta Activity in MW370 at the C-746-U Landfill



Figure 15. Technetium-99 versus Beta Activity in MW371 at the C-746-U Landfill

TCE levels in MW370 have declined steadily (to 3.2 μ g/L, which is below the MCL of 5 μ g/L) since the baseline samples of March and April 2002.

Both upgradient MW372 and MW373 have experienced above MCL TCE levels (up to 16 μ g/L in MW372 and up to 15 μ g/L in MW373) since the samples were collected to establish C-746-U Landfill background concentrations (see Appendix B for trend plots); however, the TCE concentrations in these wells are now generally decreasing. Subsequent compliance and assessment monitoring identified TCE in a well located in the northwestern portion of the landfill, MW357, with concentrations that have recently exceeded the MCL by a small amount. Concentrations in nearby MW358 and MW361 recently have exceeded the MCL. TCE levels in the C-746-U Landfill untreated leachate are less than the detection limit (1 μ g/L), as shown in Table B.2.

Based upon the known distribution of TCE located upgradient of the C-746-U Landfill and the lack of detection of TCE in the C-746-U Landfill leachate, TCE concentrations in the C-746-U Landfill RGA wells are apparently the result of upgradient sources, as discussed further in Section 4.

3.2.5 Results of Assessment Actions

TCE is present in groundwater contaminant at PGDP, associated with spills related to past industrial practices and disposal in burial grounds (Figure 16 presents a map of the areas where RGA concentrations of TCE are greater than 5 μ g/L). Analyses of untreated leachate of the C-746-U Landfill have not detected TCE.

A 2004 site investigation of the area of the C-746-S&T Landfills (located immediately upgradient of the C-746-U Landfill) identified a small area with TCE levels with a maximum value of 30 μ g/L; however, the source(s) of this contamination were not determined, such that the TCE seen in upgradient C-746-U Landfill wells may be the result of multiple potential upgradient RGA TCE sources.

The plots presented in Appendix B document that the existence and extent of TCE contamination in the C-746-U Landfill wells results from migration of upgradient RGA TCE into the C-746-U Landfill area, as discussed further in Section 4.

The detections of TCE in groundwater in wells MW357, MW358, and MW361 are attributable to TCE migration from upgradient or cross-gradient of the C-746-U Landfill as further discussed in Section 4.



FIGURE No. C746U\20110628_746U_TCE.mxd DATE 06-28-2011

Figure 16. Composite Trichloroethene Contours for the RGA for Calendar Year 2009

THIS PAGE INTENTIONALLY LEFT BLANK

4. ADDITIONAL DISCUSSION OF CONSTITUENTS IN RGA WELL SAMPLES

In accordance with 401 *KAR* 48:300, Section 8, a large number of constituents have been analyzed as part of the groundwater assessment. As noted in the screening section, most of the analyzed constituents are not present above MCLs, RSLs, or above background levels. Constituents that were detected above benchmark concentrations (or approach these levels) have sources that are not related to the C-746-U Landfill. This section presents further discussion on the nature of constituents and their likely sources.

As discussed in Section 3, beta activity and TCE are present in wells located upgradient of the C-746-U Landfill. The presence of these constituents in C-746-U Landfill wells is related to migration of the Tc-99 and TCE from upgradient locations.

As summarized in Tables 1-6 and B.1–B.3, boron, calcium, cobalt, dissolved solids, iron, magnesium, sodium, and sulfate are found more frequently and at higher concentrations in RGA wells located upgradient of the C-746-U Landfill. Figure 17 shows the groundwater monitoring well network for the C-746-S&T Landfills that describes MW372 and MW373 as wells located downgradient of the C-746-S&T Landfills, but upgradient of the C-746-U Landfill. Although groundwater flow direction can be complex in the vicinity of the C-746-U Landfill, the general flow direction in the RGA is to the northeast.

NOTE: Figure 17 is reproduced from the C-746-S&T Landfill Third Quarter report and includes a designation of UCRS wells as "upgradient," sidegradient, and downgradient. As noted earlier, this Groundwater Assessment Report does not consider the characterization of UCRS wells as upgradient/sidegradient/downgradient appropriate, considering the CSM.

As noted in Sections 2 and 3, calcium, conductivity, dissolved solids, magnesium, and sulfate are present at above C-746-U Landfill background for one (or more) C-746-U Landfill upgradient wells (wells located downgradient of the C-746 S&T Landfill); however, none of the constituents present in other RGA wells (upgradient, sidegradient, or downgradient of the C-746-U Landfill) are present above background; thus, there is no C-746-U Landfill source that accounts for the upgradient concentrations and no C-746-S&T source that accounts for them either.

Tables B.1 through B.3 demonstrate that the C-746-U and C-746-S Landfills are not the source of the elevated concentrations discussed above in MW372 and MW373. The following are examples apparent in the review of results:

- The only RGA wells with above C-746-U background concentrations of calcium are MW372 and MW373; calcium concentrations in these wells are more than double the mean value in wells located further upgradient (upgradient of the C-746-S&T Landfill); and the concentrations in other wells located downgradient of the C-746-S&T Landfill do not have above background concentrations of calcium.
- The only RGA wells with above background levels of conductivity are upgradient wells MW372 and MW373; similarly, conductivity levels are more than double the mean for the wells located further upgradient (upgradient of the C-746-S&T Landfill). The concentrations in other wells located downgradient of the C-746-S&T Landfill do not have above C-746-S&T Landfill background conductivity.



- The only RGA well with above background concentrations of magnesium is upgradient MW372, with concentrations more than double the mean value for the wells located further upgradient (upgradient of the C-746-S&T Landfill). MW387, a downgradient C-746-S&T Landfill well, has magnesium values above the C-746-S&T background. The concentrations are lower in this well than in MW372 (located downgradient of the C-746-S&T Landfill), but upgradient of the C-746-U Landfill; thus, the source of the magnesium in the MW372 well cannot be pinpointed and is certainly not attributable to the C-746-U Landfill.
- The wells with dissolved solids levels greater than the SMCL of 500 mg/L are upgradient wells MW372 and MW373. MW387, located downgradient of the C-746 S&T Landfill, shows dissolved solids levels above the C-746-S&T background, but the concentration is less than 500 mg/L; thus the concentrations in this well do not allow the source of the dissolved solids in MW372 and MW373 to be pinpointed.
- The only RGA well with above background concentrations of sulfate is MW372; however, MW373 has "background" concentrations greater than the SMCL (as high as 810 mg/L) and averages more than an order of magnitude higher than the other background well completed at the same elevation. If a UCRS or landfill source close to these wells were causing the high sulfate levels, one would expect to find higher concentrations in the upper RGA well (MW372) rather than the lower RGA well (MW373); however, the lower RGA well has the higher concentration of this sulfate in this cluster, which is not consistent with a UCRS or C-746-U Landfill source.

These constituents are not found at levels above background in UCRS wells except for sulfate; and the sulfate concentrations in the UCRS are an order of magnitude (or more) less than the highest concentrations in RGA wells. Thus, the UCRS cannot account for all of the sulfate present in the RGA because concentrations in the UCRS would have to be 58 times higher than the RGA concentrations to account for all the sulfate. Though the UCRS is difficult to monitor, the UCRS and the C-746-U Landfill do not appear to be a statistically quantifiable source of the exceedances or elevated calcium, conductivity, dissolved solids, magnesium, or sulfate because elevated concentrations are not found in the UCRS, and the RGA concentrations downgradient of the C-746-U Landfill are generally well-explained, except as discussed herein. Thus, although the upgradient C-746-U Landfill wells MW372 and MW373 have higher than expected RGA concentrations of some constituents, no further upgradient sources of these constituents have been pinpointed. There is no indication of a C-746-S&T Landfill source for the constituents discussed above.

Concentrations of cobalt, iron, and manganese remain above the respective RSL, but the values are not above background. These constituents are not elevated above background in any of the C-746-S&T Landfill RGA wells or the C-746-U Landfill RGA wells; thus these constituents are not indicative of contamination from the C-746-U Landfill.

4.1 DISSOLVED SOLIDS

Dissolved solids concentrations exceed the EPA SMCL only in upgradient well MW373; however, concentrations remain high, especially in MW372 and MW373, the upgradient wells. The source(s) of these high concentrations in upgradient wells is not known. The other downgradient C-746-S&T wells do not have concentrations that exceed the SMCL of 500 mg/L; thus, sources located upgradient of the C-746-U Landfill cannot be pinpointed and cannot account for all of the dissolved solids seen in these C-746-U upgradient wells. With the failure to pinpoint an upgradient source, one possible explanation is well fouling that could result in increased dissolved solids concentrations; however, there is no independent evidence that these wells are fouled.

4.2 MAGNESIUM

Magnesium levels show an above background concentration in upgradient well MW372 (URGA). Magnesium levels are higher in MW373 than in other upgradient wells completed at the same elevation; however, the difference is not statistically significant (as determined by the quarterly C-746-U Landfill monitoring program). Similar to the calcium and dissolved solids results described above, the upgradient source(s) of magnesium, if present, cannot be pinpointed, thus, one possible explanation is well fouling that could result in increased dissolved solids concentrations. However, there is no independent evidence that these wells are fouled.

4.3 MANGANESE

Manganese is not found in RGA wells at above background concentrations, but is found at concentrations above its RSL. The concentrations found in the vicinity of the C-746-U Landfill are within the range of C-746-U Landfill background and also concentrations found elsewhere at PGDP as summarized in Appendix B; thus, there is no evidence of a C-746-U Landfill manganese source.

Manganese is a naturally occurring element that is widely distributed in nature. The most abundant manganese minerals are oxide compounds, followed by sulfides and carbonates. In any water sample, the soluble manganese content will depend on the geochemical characteristics of the soil matrix, environmental transformation of manganese compounds, activity of soil microorganisms, and uptake by plants. Some manganese compounds are soluble in water. The RSL for manganese is 0.032 mg/L. The secondary drinking water standard for manganese is 0.05 mg/L. Gradual weathering and conversion to soluble salts account for most of the manganese content in rivers, seawater, and groundwater. Manganese is a common component of well rehabilitation chemicals.

The influence of oxidation reduction potential (Eh) and pH are very important when considering the movement of manganese in the environment. At low Eh and pH, manganese is more likely to be extracted from a solid phase and kept in solution. At high Eh and pH, manganese is more easily precipitated as an oxide. Chelating agents may complex manganous manganese in solution so extensively that precipitation is inhibited, even under conditions favorable for precipitation. Activity of microbial organisms has been shown only to both dissolve and to precipitate manganese under certain environmental conditions.

Concentrations of manganese in C-746-U Landfill RGA wells vary due to the same conditions that affect other RGA PGDP well concentrations.

4.4 SULFATE

Sulfate is commonly found in air, soil, and water. Sulfate is soluble in water and can be found in variable concentrations in environmental media. It is second to bicarbonate as the major anion in hard water supplies. The most common form of sulfur in well-oxygenated waters is sulfate; the only RGA well with above background concentrations of sulfate is MW372—a well identified as upgradient to the C-746-U Landfill; however, MW373 has "background" concentrations greater than the SMCL (as high as 810 mg/L measured during sampling of this upgradient well during the background sampling program) and the concentrations in MW373, measured during the background sampling program for the C-746-U Landfill average more than an order of magnitude higher than the concentrations in the other background well completed at the same depth.

Similar to other constituents, the higher concentrations of sulfate in MW372 and MW373 cannot be attributed to specific upgradient sources. A possible source of sulfate is as a decomposition product of sulfamic acid historically used to rehabilitate wells; however, these wells have not been rehabilitated using sulfamic acid for more than seven years. Thus, one must consider the possibility that the dissolved solids concentrations may result from well fouling or other impacts that prevent obtaining a sample that is representative of the concentrations of these constituents in the RGA in the vicinity of the C-746-U Landfill.

4.5 PCBS

4.5.1 Physical Properties

PCBs refer to a group of chlorinated organic chemicals that are clear to pale yellow in color; odorless to mildly aromatic; and, depending on the percent of chlorination, can take the form of solids, waxy resins, or viscous oily liquids. PCBs were first synthesized in 1881, but did not become widely used until the 1930s.

Production of PCBs in quantity in the United States began in 1929 and continued until 1977. Total production of all forms of PCBs totals about 700,000 tons.

Today, PCBs are listed as a persistent organic pollutant and are found in the environment worldwide. Known mechanisms of transport include vapor transport in air (wind), precipitation as rain or snow, and transport in water when adsorbed on sediments. PCBs are not known to occur naturally in the environment.

If released into soil, PCBs become tightly adsorbed to the soil particles, with adsorption generally increasing with the degree of chlorination of the PCB. PCBs generally do not leach significantly in aqueous soil systems; when leaching does occur, the higher chlorinated congeners have a lower tendency to leach than the lower chlorinated congeners. In the presence of organic solvents, PCBs may leach quite rapidly through the soil. PCBs have superior physical properties that make them attractive for use in industrial products and processes. These include low flammability, high resistivity, low conductance, and high thermodynamic and chemical stability.

PCBs have a low solubility in water, but are known to be soluble in most organic solvents, oils, and fats.

Intentional methods of destruction of PCBs include incineration with high heat and catalytic processes plus certain chemical processes. Destruction by metabolic organisms and by environmental processes proceeds quite slowly.

Due to their unique physical properties, PCBs have been used in a wide variety of applications. These include hydraulic fluids, lubricants, cutting oils, dielectric fluids for transformers and capacitors, electric power cables, paints, inks, sealants, gasket materials adhesives, plasticizers, fire retardants, asphalt, brake linings, heat transfer systems, pesticide extenders, dedusting agents, carbonless reproductive paper, and a fixative for microscopy.

PGDP equipment that contains PCBs includes transformers and capacitors, electric power cables, asphalt roofing materials, hydraulic fluid, paint, and gaskets.

4.5.2 PCBs at C-746-U Landfill

PCBs that are the subject of this assessment are those that have been detected in three wells, MW361 (LRGA), MW363 (URGA), and MW365 (UCRS). The congeners PCB-1016 and PCB-1242 account for 28 of the 30 detections in these three wells. (PCB-1016 and PCB-1242 are similar, have 12 carbon atoms in the biphenyl skeleton and are composed of 42% chlorine by mass.) Single detections of PCB-1248 in wells MW363 and MW365 account for the other two detections.

All of these wells are located considerably north of active landfill phases (and well downgradient of the C-746-U Landfill in the RGA). The wells with detections are located near an abandoned homestead with an open water well that is 36 inches in diameter and approximately 47-ft deep. The top of the open water well has been covered as a safety precaution.

4.5.3 Results of Assessment Actions

The old, brick-lined homestead water well has been sampled and should be abandoned in accordance with regulatory requirements. This is a UCRS well located well distal to the C-746-U Landfill; thus, results from this well do not identify a C-746-U Landfill source. Nevertheless, analytical results for the water sample are reproduced in Appendix C for completeness. Attempts to obtain sediment samples from the homestead well were unsuccessful. The sampling attempts indicate a hard bottom to the well with insufficient sediment to obtain a sample. Appendix E describes the soil sampling event conducted in the vicinity of MW365 that did not identify any PCBs in soils.

4.6 PCB, TOTAL

PCBs (congeners PCB-1016 and PCB-1242) were detected in groundwater of MW366 at levels less than 0.001 mg/L in July 2003, October 2006, and July 2007. As with other C-746-U Landfill MWs, the source of the PCBs appears to be cross contamination associated with well rehabilitation activities. The first well rehabilitation action in MW366 was in April 2003. See Appendix E for further assessment of the source of PCBs in the C-746-U Landfill MWs.

4.7 TOTAL ORGANIC CARBON

EPA defines TOC in terms of mg/L of the amount of carbon that can be converted to carbon dioxide. TOC concentrations are lower than historical background levels; thus, there is no indication of any current above background sources of TOC. TOC concentrations are consistent with those found in RGA wells at PGDP; thus, they do not indicate any incremental contribution to concentrations from the C-746-U Landfill.

4.8 DISCUSSION OF LANDFILL SOURCES

Appendix B provides a review of the concentrations of constituents found in wells located in the vicinity of the C-746-U Landfill, compared to concentrations of constituents found in the C-746-U Landfill leachate. If the C-746-U Landfill had impacts on the RGA groundwater, one principal mechanism would be via migration of the landfill leachate. As shown in that table, there is no "fingerprint" that can be associated with the C-746-U Landfill leachate. Leachate concentrations of many constituents typically are the same order of magnitude as the mean concentrations found in RGA wells at PGDP.

Overall, none of the constituents present in leachate is found at a concentration that accounts for more than a fraction of the concentration of the RGA constituents because the dilution attenuation factor (DAF) estimated for native UCRS materials at PGDP is 58, and the DAF for a lined landfill (like the C-746-U Landfill) is expected to be orders of magnitude higher (DOE 2013, Attachment C2).

Constituents present in the UCRS have the potential to migrate to the RGA. This migration will be attenuated based upon the nature of the materials through which the migration is occurring. For example, migration through a clay matrix will be slower than through a sandy matrix. Modeling performed at PGDP has estimated the attenuation factor at 58 (UCRS concentrations are attenuated by a factor of 58 as the constituent migrates to the RGA). A landfill liner would attenuate the leachate better than the native clay, silt, and sand materials present in the UCRS. For the constituents present in leachate to account for the concentrations found in the RGA, the leachate concentrations would have to be orders of magnitude higher than they are. Thus, the C-746-U Landfill is not the source of the RGA groundwater constituents found in wells in the vicinity of the C-746-U Landfill.

Similarly, concentrations of constituents in the C-746-S leachate are the same order of magnitude as the C-746-U Landfill and the concentrations found in the RGA. Thus, there is no statistically quantifiable information supporting that leachate from the C-746-S Landfill is a source of the RGA groundwater constituents found in wells in the vicinity of the C-746-U Landfill.

There are other potential mechanisms for RGA groundwater to be affected by C-746-U Landfill operations, including migration of landfill gas, spills outside lined landfill areas, and interactions between landfill gas/leachate with native materials. However, the RGA results from the RGA wells do not indicate statistically quantifiable contributions to RGA wells from such C-746-U Landfill operations. Furthermore, it is unlikely that major impacts to RGA groundwater could occur without detection, as follows:

- Direct RGA groundwater contamination by landfill gas without impacts on landfill leachate concentrations would be unlikely because landfill gas is generated in the landfill cells where leachate is also generated. While leachate will flow downward under gravity, landfill gas will expand in all directions. It is unlikely that landfill gas can be so contaminated as to affect the RGA groundwater without it also being detected in the landfill leachate.
- Spills outside the lined area would have to be very large for constituents to migrate through the UCRS and impact the RGA groundwater, given the DAF of the UCRS at 58. Thus, it is unlikely that a large spill could have occurred without being detected. There is no indication of RGA impacts from the C-746-U from any source, including impacts that reasonably could be attributable to spills outside the lined area.
- Landfill gas and leachate can interact with groundwater and aquifer materials to liberate constituents present in native materials; however, there is no indication of RGA impacts from the C-746-U Landfill from any source, including impacts that reasonably could be attributable to impacts of landfill gas/leachate on native materials that are also not detectable in leachate. In addition, even if there are impacts on UCRS matrix materials by landfill gas/leachate, it is unlikely that these impacts would be seen in RGA groundwater due to the significant buffering capacity of the native UCRS materials.

THIS PAGE INTENTIONALLY LEFT BLANK

5. HYDROGEOLOGIC SETTING

PGDP lies in the Jackson Purchase Region of western Kentucky between the Tennessee and Mississippi Rivers, and is bounded on the north by the Ohio River. The confluence of the Ohio and Mississippi Rivers is approximately 35 miles downstream (southwest) from the site, and the confluence of the Ohio and Tennessee Rivers is approximately 15 miles upstream (east) from the site. PGDP is located approximately 3.5 miles south of the Ohio River and 10 miles west of the city of Paducah. Regional groundwater flow discharges to the main surface water features. In the PGDP area, the groundwater systems discharge to Bayou Creek and Little Bayou Creek, perennial watershed streams, and to the Ohio River.

PGDP is located in the northern tip of the Mississippi Embayment portion of the Coastal Plain physiographic province. The Mississippi Embayment is a large north-south tectonic and erosional trough filled with unconsolidated sediments derived from the middle of the North American continent. In the region, the stratigraphic sequence consists of Cretaceous, Tertiary, and Quaternary sediments unconformably overlying Paleozoic bedrock.

5.1 PHYSICAL CHARACTERISTICS OF THE AREA

Locally, PGDP lies within the drainage areas of the Ohio River, Bayou Creek, and Little Bayou Creek. The plant is situated on the divide between the two creeks. Surface flow is east-northeast toward Little Bayou Creek and west-northwest toward Bayou Creek.

Local elevations range from 290 ft above mean sea level (amsl) along the Ohio River to 450 ft amsl in the southwestern portion of PGDP near Bethel Church Road. Generally, the topography in the PGDP area slopes toward the Ohio River at an approximate 27 ft per mile (ft/mile) gradient. The terrain in the vicinity of the plant is slightly modified by the dendritic drainage systems associated with the two principal streams in the area, Bayou Creek and Little Bayou Creek. These streams have eroded small valleys, which are about 20 ft below the adjacent plain.

5.2 GENERAL GEOLOGY

In the immediate vicinity of PGDP, Coastal Plain deposits unconformably overlie Mississippian carbonate bedrock. The full Coastal Plain stratigraphic sequence to the south of PGDP consists of the following units: sands and clays of the Clayton/McNairy Formations; the Porters Creek Clay; and Eocene sand and clay deposits (undivided Jackson, Claiborne, and Wilcox Formations). Continental Deposits unconformably overlie the Coastal Plain deposits, which are, in turn, covered by loess and/or alluvium. Both the loess and alluvium are composed of clayey silt and silty clay.

In the central and northern part of the PGDP site, including the area of the C-746-U Landfill, the lowermost Coastal Plain sediments are composed of unconsolidated, interbedded, fine-grained sand, silt and clay of the Upper Cretaceous-aged McNairy Formation (SAIC 1994; Woolery and Street 2002). The thickness of the McNairy Formation ranges from 221 to 247 ft.

A principal geologic feature in the PGDP area is the Porters Creek Clay Terrace, a subsurface terrace cut that trends approximately east to west across the southern portion of the plant. The Porters Creek Clay Terrace represents the southern limit of erosion or scouring of the ancestral Tennessee River. In the area

north of the subsurface terrace cut, including the C-746-U Landfill, Continental Deposits directly overlie the McNairy Formation.

The Continental Deposits resemble a large, low-gradient alluvial fan that covered much of the region and buried the erosional topography. Thicker sequences of Continental Deposits, as found underlying PGDP, represent valley fill deposits and can be divided informally into a lower unit (gravel facies) and an upper unit (silt facies). The Lower Continental Deposit (LCD) is a Pliocene (?)⁴ to Pleistocene-aged gravel facies consisting of fine to coarse chert gravel in a matrix of very fine to medium sand and silt that rests on an erosional surface representing the beginning of the valley fill sequence. In total, the gravel units average approximately 30-ft thick, but some thicker deposits (as much as 50 ft) exist in deeper scour channels. The LCD is stratigraphically equivalent to the Mounds Gravel as designated by the Illinois Geological Survey (IGS) or the Lafayette Formation (Lafayette gravel) in other parts of the region (Sexton 2006) (Langston and Street 1998).

The alluvial gravels and sands of the LCD are overlaid by Late-Tertiary through Quaternary and Holocene aged sediments including the Plio-Pleistocene Metropolis Formation and Pleistocene loess units. These deposits are cumulatively identified as Upper Continental Deposits (UCD) and range between 30- and 60-ft thick beneath the PGDP site. Investigations conducted at PGDP and the C-746-U Landfill identified at least four separate loess depositional events in the site's stratigraphic sequence (KRCEE 2006).

Subregional studies (McHaffie1983; Nelson et al. 1996; Kiefer 1996; Drahovzal and Hendricks 1997; SAIC 1994) and local studies in the vicinity of PGDP (Langston and Street 1996; Anderson-Blitz 2008) indicate that paleo-tectonic activity has impacted local bedrock and overlying unconsolidated strata upward through the Plio(?)-Pleistocene (late Quaternary) boundary (Langston and Street 1996; Anderson-Blitz 2008). Specifically within the area of the C-746-U Landfill, studies have shown that tectonic activities may have impacted bedrock up through the Pliocene-Late Quaternary Continental Deposits (KRCEE 2006; DOE 2003). These studies report that tectonic activities have not affected the most recent loess units (Late Quaternary-Holocene) and surficial soils in the area.

5.3 SUBSURFACE GEOLOGY AT THE C-746-U LANDFILL

The Holocene Fault Study provides information on the geologic setting in the vicinity of the C-746-U Landfill (KRCEE 2006). Figure 18 is a reproduction of Figure 5 from the Holocene Fault Study that describes the regional geologic map in the vicinity of the C-746-U Landfill. This figure is reproduced without editing to provide context for the discussion of the regional geology. Features marked on the figure may not be representative of current conditions. Figure 19 reproduces Figure 6 from the study that provides a schematic stratigraphic column. Figure 20 reproduces Figure 7 from the study that provides a geologic fault study. Figure 21 provides a copy of the relevant portion of Plate 1 from that study that provides the section location.

The area lithologic logs document that the LCD gravels directly overlie the McNairy Formation beneath the C-746-U Landfill on an erosional surface that occurs at depths of 70 to 86 ft beneath the C-746-U Landfill (elevations of 285 ft to 300 ft amsl). These LCD gravels vary from 29- to 39-ft thick. The depth of the top of the LCD gravels ranges from 42 to 52 ft below the ground surface (bgs). The LCD gravels, in turn, are overlaid by 42 to 52 ft of silts, sands, and clays of the UCD (including the Metropolis Formation and Pleistocene loess units).

⁴ (?) Indicates uncertainty in the age of the geologic unit.


Figure 18. Regional Geologic Map (Modified from Finch 1967)

SYSTEM	SERIES	FORMATION	LITHOLOGY	THICKNESS (IN FT)	DESCRIPTION
RY	HOLOCENE AND PLEISTOCENE	ALLUVIUM		0-40	Brown or gray sand and silty clay or clayey silt with streaks of sand.
TERNAL	PLEISTOCENE	PEORIA LOESS ROXANA SILT LOVELAND SILT		0-43	Brown or yellowish-brown to tan unstratified silty clay.
QUA	PLEISTOCENE	METROPOLIS		2 101	Clay Facies - mottled gray and yellowish brown to brown clayey silt and silty clay, some very fine sand, trace of gravel. Often micaceous.
	PLIOCENE- MIOCENE (?)	MOUNDS GRAVEL		J-141	Gravel Facies - reddish-brown clayey, silty and sandy chert gravel and beds of gray sand.
	EOCENE AND PALEOCENE	JACKSON, CLAIBORNE,		0-200+	Red, brown or white fine to coarse grained sand. Beds of white to dark gray clay are distributed at random.
TERTIARY		AND WILCOX FORMATIONS		0-100+	White to gray sandy clay, clay conglomerates and boulders, scattered clay lenses and lenses of coarse red sand. Black to dark gray lignitic clay, silt or fine grained sand.
		PALEOCENE	PORTERS CREEK CLAY		0-200
		CLAYTON FORMATION		Undetermined	Lithologically similar to underlying McNairy Formation.
UPPER CRETACEOUS		McNAIRY FORMATION		200-300	Grayish-white to dark gray micaceous clay, often silty, interbedded with light gray to yellowish-brown very fine to medium grained sand with lignite and pyrite. The upper part is interbedded clay and sand, and and the lower part is sand.
		RUBBLE ZONE		Undetermined	White, semi-rounded and broken chert gravel with clay.
MISSISSIPPIAN		MISSISSIPPIAN CARBONATES		500+	Dark gray limestone and interbedded chert, some shale.

Figure 19. Schematic Stratigraphic Column of the PGDP Region

(Modified from Nelson et al. 2002 and SAIC 2004)



Figure 20. Geologic Section Depicting Generalized Stratigraphy Underlying C-764-U Landfill (from SAIC 1994)

45



Figure 21. Topographic Map of the Paducah Gaseous Diffusion Plant C-746-U Landfill Expansion

5.4 HYDROGEOLOGY AT PGDP AND THE C-746-U LANDFILL

At PGDP, the unconsolidated sediments overlying the McNairy Formation have been divided into two primary hydrogeologic units, the UCRS and the RGA. In these divisions, the UCRS includes most of the Plio-Pleistocene Metropolis Formation and the Pleistocene loess packages. The RGA hydrogeologic unit consists of a basal sand (where present) of the Metropolis Formation and the gravels and sands of the Lower Continental Deposits. The McNairy Formation underlies the RGA and functions as a bottom aquitard to the RGA throughout the central and northern parts of the PGDP area.

Flow through the UCRS is downward into the RGA. Flow in the RGA is lateral beneath the landfill, toward the north-northeast. RGA groundwater flow from the C-746-U Landfill area ultimately discharges into the lower reaches of Bayou and Little Bayou Creeks and the Ohio River.

The lithologic and well construction logs for MW cluster MW371 (UCRS), MW369 (upper RGA), and MW370 (lower RGA) document the placement of MWs south/upgradient of the C-746-U Landfill and north/downgradient of the present North-South Diversion Ditch (see Figure 2 and Appendix G). At this location, the UCRS consists of a loess sequence, which extends to 19 ft bgs, and silts and fine sands of the Metropolis Formation, that extend to 43 ft bgs. The RGA in this well cluster consists of an upper, sandier member that extends to 54 ft bgs and a lower gravel-dominant member that extends to 76 ft bgs, the erosional surface in the top of the McNairy Formation. The UCRS well, MW371, is screened in a sand unit of the Metropolis Formation; MW369 is screened in the upper, sandier member of the RGA; and MW370 is screened in the lower, gravel-rich member of the RGA.

The logs of MW cluster MW359 (UCRS), MW357 (upper RGA) and MW358 (lower RGA) provide an example of the placement of MWs north/downgradient of the C-746-U Landfill. At this location, the UCRS consists of a loess package that extends to 28 ft bgs and clays and fine sands of the Metropolis Formation that extend to 45 ft bgs. The RGA in this downgradient well cluster consists of an upper sandier member that extends to 55 ft bgs and a lower gravel-dominant member that extends to 84 ft bgs, to the top of the McNairy Formation. (The RGA is 6 ft thicker in the north well cluster.) As in the downgradient well cluster, the UCRS well, MW359, is screened in sand units of the Metropolis Formation; MW357 is screened in sands and gravels of the upper RGA; and MW358 is screened in the lower, gravel-rich member of the RGA.

The logs of Appendix G document the lateral continuity of the primary hydrogeologic units beneath the C-746-U Landfill and consistency in the placement of the MWs.

5.5 CONCEPTUAL SITE MODEL

Soil boring logs, MW and piezometer water level records, soil geotechnical tests, and groundwater flow models of the area of the C-746-U Landfill provide sufficient data for the development of a CSM. In general, groundwater flow is downward through the silts, clays, and fine sands of the UCRS. In contrast, the underlying RGA is highly conductive and provides the main conduit for lateral groundwater flow. Groundwater flow is in a north-northeasterly direction in the vicinity of the C-746-U Landfill. The Ohio River and lower reaches of Little Bayou and Bayou Creeks are the discharge areas for the RGA flow system.

The fine sands, silts, and clays of the upper McNairy Formation in the C-746-U Landfill area have much lower hydraulic conductivity than the gravels and sands of the RGA. Given the slight downward vertical gradient that exists across the contact between the two units, there is little groundwater flow into the McNairy Formation.

The collective information about the units underlying the C-746-U Landfill provides the following conceptual model:

- Waste materials placed in the C-746-U Landfill can be a source of contamination to the landfill leachate. The landfill is designed to collect and treat generated leachate.
- Because water flows downward through the UCRS, the presence of contaminants in UCRS groundwater are related to sources in the immediate vicinity of the well, including soils located above the well screen.
- Naturally-occurring minerals in loess and the Metropolis Formation can be sources of elements and compounds in UCRS water well samples.
- Contaminants found in groundwater from RGA MWs of the C-746-U Landfill can have sources upgradient of the landfill. Potential sources upgradient of the C-746-U Landfill complex include these:
 - Infiltration from the North-South Diversion Ditch,
 - Migration through the RGA from upgradient/PGDP (primarily TCE and Tc-99), and
 - Other PGDP facilities, such as the C-616 lagoons.

6. ENHANCED SAMPLING OF EXISTING WELLS

The presence of PCBs detected in C-746-U Landfill MWs led to enhanced sampling at PGDP to evaluate potential impacts by PCBs on the RGA sitewide as shown in Figure 22.

6.1 EXISTING WELLS

In order to monitor groundwater quality, 35 MWs are sampled annually for geochemical environmental surveillance. Sampling of these wells is not driven by regulation, but is conducted in support of the Federal Facility Agreement investigations and DOE Order 450.1. In September and October 2008, these wells were sampled for PCBs, in addition to the parameters previously sampled in these wells. No PCBs were detected, documenting that there is no large PCB groundwater plume associated with the industrial area of PGDP. Table 9 and Figure 22 summarize the 35 wells and the C-746-U Landfill wells that were sampled for PCBs.



Figure 22. Location of RGA Wells with PCB Analyses to the C-746-S&T and C-746-U Landfill Wells

Wall	Polychlo	Data Collected			
wen	Result	Detection Limit	Detect?	Date Collected	
MW20	0.18	0.18	NO	10-Sep-08	
MW99	0.18	0.18	NO	08-Sep-08	
MW100	0.17	0.17	NO	08-Sep-08	
MW125	0.17	0.17	NO	09-Sep-08	
MW134	0.17	0.17	NO	16-Sep-08	
MW145	0.17	0.17	NO	17-Sep-08	
MW152	0.18	0.18	NO	08-Sep-08	
MW161	0.18	0.18	NO	01-Oct-08	
MW163	0.17	0.17	NO	11-Sep-08	
MW188	0.17	0.17	NO	22-Sep-08	
MW193	0.17	0.17	NO	10-Sep-08	
MW201	0.18	0.18	NO	09-Sep-08	
MW206	0.17	0.17	NO	22-Sep-08	
MW242	0.17	0.17	NO	10-Sep-08	
MW243	0.17	0.17	NO	10-Sep-08	
MW255	0.17	0.17	NO	01-Oct-08	
MW256	0.18	0.18	NO	01-Oct-08	
MW257	0.17	0.17	NO	23-Sep-08	
MW258	0.17	0.17	NO	17-Sep-08	
MW260	0.17	0.17	NO	11-Sep-08	
MW261	0.17	0.17	NO	23-Sep-08	
MW288	0.17	0.17	NO	17-Sep-08	
MW291	0.17	0.17	NO	17-Sep-08	
MW292	0.18	0.18	NO	09-Sep-08	
MW328	0.18	0.18	NO	04-Sep-08	
MW329	0.17	0.17	NO	04-Sep-08	
MW339	0.17	0.17	NO	23-Sep-08	
MW343	0.17	0.17	NO	01-Oct-08	
MW381	0.18	0.18	NO	09-Sep-08	
MW403-PRT3	0.18	0.18	NO	30-Sep-08	
MW404-PRT3	0.17	0.17	NO	30-Sep-08	
MW404-PRT4	0.18	0.18	NO	30-Sep-08	
MW404-PRT5	0.17	0.17	NO	30-Sep-08	
MW409	0.17	0.17	NO	15-Sep-08	
MW414	0.18	0.18	NO	22-Sep-08	

Table 9. PGDP RGA Wells with PCB Analyses and the C-746-S&Tand C-746-U Landfill Wells

THIS PAGE INTENTIONALLY LEFT BLANK

7. CONCLUSIONS

Based on this groundwater assessment, the above-benchmark (including background) constituent results found in groundwater from the vicinity of the C-746-U Landfill are from sources other than the C-746-U Landfill. None of the well samples indicate any landfill-related, above background levels of constituents present above MCLs or other risk-based levels of concern.

Only TCE and beta activity concentrations are persistently above MCLs in a few RGA wells; however, these concentrations are higher in the RGA wells located upgradient of the landfill and are attributable to migration from upgradient TCE and beta activity.

The assessment does identify the following apparent source(s) of the elevated above background levels.

- Constituents (present in the RGA wells) are migrating from upgradient sources, including RGA TCE and beta activity plumes.
- Constituents are present in upgradient to the C-746-U Landfill RGA wells at higher concentrations than in other wells in the vicinity of the landfill, including calcium, conductivity, dissolved solids, magnesium, potassium, and sulfate. Although potential sources of these constituents in the upgradient wells have been discussed, attributing the sources of these constituents continues to be uncertain because the concentrations of these constituents are higher in C-746-U Landfill upgradient wells, but lower in wells located further upgradient (i.e., upgradient of the C-746-S&T Landfill).
- Constituents (PCBs) apparently were introduced as a result of historical well rehabilitation efforts.

The possible sources are summarized in Table 10.

Possible Source	Parameter
Underlying groundwater contamination associated with sources upgradient of C-746-U Landfill	Beta activity, TCE
Upgradient RGA Sources	Potentially calcium, dissolved solids, magnesium, sodium, sulfate, although attribution has uncertainty
Elevated levels resulting from MW fouling	Calcium, dissolved solids, magnesium, sodium, sulfate
Elevated levels resulting from MW contamination during rehabilitation activities	PCBs
Natural variability in the groundwater system	Except as noted, concentrations are within range of concentrations found in PGDP wells; thus, no indication of a C-746-U contribution

Table 10. Summary	of Possible	Contaminant Sources
-------------------	-------------	----------------------------

N/A = not applicable

7.1 RECOMMENDATIONS

Recommendations resulting from this groundwater assessment are as follows:

• Abandon the open well at the old home site. This well presents the potential for a direct contaminant pathway to the UCRS.

No additional assessment, monitoring, or abatement activities have been identified as needed to complete the groundwater assessment process.

8. REFERENCES

- Anderson-Blitz, C. 2008. Integrated Geophysical Imaging of Subsurface Geologic Conditions across a Contaminant Plume, McCracken County, Kentucky, Final Master's Thesis, April.
- BJC (Bechtel Jacobs Company LLC) 2001. C-746-U Solid Waste Landfill Groundwater Monitoring Plan, Paducah Gaseous Diffusion Plant, Paducah, Kentucky, BJC/PAD-205/R1, Bechtel Jacobs Company LLC, Paducah, KY, December.
- Drahovzal, J. A. and T. D. Hendricks 1997. *Geologic Features Relevant to Ground-water Flow in the Vicinity of the Paducah Gaseous Diffusion Plant* Kentucky Geological Survey Open-File Report OF-97-02, Kentucky Geological Survey, University of Kentucky.
- DOE (U.S. Department of Energy) 2001. Feasibility Study for the Groundwater Operable Unit at Paducah Gaseous Diffusion Plant, Paducah, Kentucky, U.S. Department of Energy, Paducah, KY, DOE/OR/07-1857&D2, August.
- DOE 2003. Seismic Investigation Report for Siting of a Potential On-Site CERCLA Waste Disposal Facility at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, DOE/OR/07-2038&D2, U.S. Department of Energy, Paducah, KY, May.
- DOE 2012. C-746-U Contained Landfill Third Quarter Calendar Year 2012 (July-September) Compliance Monitoring Report Paducah Gaseous Diffusion Plant, Paducah, Kentucky, PAD-ENM-0067/V3, U.S. Department of Energy, Paducah, KY, November.
- DOE 2013. Soils Operable Unit Remedial Investigation Report at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, DOE/LX/07-00358&D2/R1, U.S. Department of Energy, Paducah, KY, February.
- EPA (U.S. Environmental Protection Agency). Regional Screening Levels, http://www.epa.gov/region9/superfund/prg/.
- EPA Drinking Water Standard Maximum Contaminant Levels and Secondary Maximum Contaminant Levels, http://water.epa.gov/drink/contaminants/index.cfm#List
- KRCEE (Kentucky Research Consortium for Energy and Environment) 2006. *Investigation of Holocene Faulting at Proposed C-746-Landfill Expansion*, UK/KRCEE Doc #: p.17.6, prepared by William Lettis & Associates, Inc., 2006.
- Kiefer, J. D., J. A. Drahovzal, J. C. Cobb, and D. A. Williams 1996. "Review of Illinois State Geologic Survey Study on Quaternary Faulting in Southern Illinois and Reconnaissance of Neotectonic Structures of Southern Illinois and Western Kentucky," Kentucky Geological Survey Open-File Report OF-97-01.
- Langston, C. and R. Street 1998. "Acquisition of SH-wave Seismic Reflection and Refraction Data in the Area of the Northeastward Trending Contaminant Plume at the PGDP," unpublished Final Report prepared by Langston and Street of the Department of Geological Sciences, University of Kentucky, Lexington, KY.

- Nelson, W. J., B. F. Denny, J. A. Devera, L. R. Follmer, J. M. Masters, and J. Sexton, 1996. *Quaternary Faulting in the New Madrid Seismic Zone in Southernmost Illinois*: Final technical report to the U.S. Geological Survey under the National Earthquake Hazards Reduction Program, Award No. 1434-95-G-2525, 41 p., January.
- PRS (Paducah Remediation Services, LLC) 2008. Groundwater Assessment Plan for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, PRS-PROJ/0006/R2, Paducah Remediation Services, LLC, Kevil, KY, January.
- Sexton, J. L. 2006, Lithologic and Stratigraphic Compilation of Near Surface Sediments for the Paducah Gaseous Diffusion Plant, McCracken County, KY, University of Kentucky Master's Thesis, Paper 295.
- SAIC (Science Applications International Corporation) 1994. Solid Waste Landfill Subsurface Investigation Report, unpublished consultant report prepared for the U.S. Department of Energy, under contract 18B-99069C.
- Woolery, E. W. and R. Street 2002. "Quaternary Fault Reactivation in the Fluorspar Area Fault Complex of Western Kentucky: Evidence from Shallow SH-wave Reflection Profiles, Seismological Research Letters," Volume 73, No. 5, p. 628-639.

APPENDIX A

CORRESPONDENCE

THIS PAGE INTENTIONALLY LEFT BLANK

RECEIVED

FEB 1 9 2008

Robert D. Vance Secretary



ENVIRONMENTAL AND PUBLIC PROTECTION CABINET

Steven L. Beshear Governor DEPARTMENT FOR ENVIRONMENTAL PROTECTION DIVISION OF WASTE MANAGEMENT

14 REILLY ROAD FRANKFORT, KENTUCKY 40601 PHONE (502) 564-6716 FAX (502) 564-4049 www.waste.ky.gov

February 13, 2008

Mr. William E. Murphie, Manager United States Department of Energy Portsmouth Paducah Project Office 1017 Majestic Place, Suite 200 Lexington, Kentucky 40513

Certified Mail Number: 7005 3110 0000 3555 3435

Mr. Russell Boyd, Site Manager Paducah Remediation Services, LLC 761 Veterans Avenue Kevil, Kentucky 42053

Certified Mail Number: 7005 3110 0000 3555 3442

 RE: Groundwater Assessment Plan United States Department of Energy - Paducah Gaseous Diffusion Plant Agency Interest No. 3059 C-746-U Solid Waste Contained Landfill Application No. AIN20070003 Solid Waste Permit # 073-00045 McCracken County

Dear Mr. Murphie and Mr. Boyd:

The Kentucky Division of Waste Management, Solid Waste Branch (Division) has reviewed the Groundwater Assessment Plan received on January 9, 2008 for the C-746-U Solid Waste Contained Landfill. The Groundwater Assessment Plan (AIN20070003) is hereby approved.

Additional copies of the Groundwater Assessment Plan (AIN20070003) were received on January 15, 2008, and an approved copy is enclosed along with a revised TEMPO permit. The Division approved a minor modification for leachate storage (APE20060011) on February 12, 2008. An approved copy of the application for APE20060011 is also enclosed.

Kentud

Mr. William E. Murphie, Mr. Russell Boyd February 13, 2008 Page No. 2 of 2

Be advised, if you consider yourself aggrieved by the issuance of this permit, you have a right pursuant to KRS 224.10-420 (2) and 401 KAR 47:130 Section 2 (3), to file with the cabinet a petition demanding a hearing. This right to demand a hearing shall be limited to a period of thirty (30) days after receipt of this permit.

Should you have any questions regarding this matter, please contact me at (502) 564-6716, extension 240 or Jeff Pratt, P.E. at extension 204.

Sincerely,

R.D. Lugesby

Ronald D. Gruzesky, P.E. Manager, Solid Waste Branch

Enclosures RDG/rth/jnn

c: Reading File

APPENDIX B

TREND CHARTS

THIS PAGE INTENTIONALLY LEFT BLANK

B.1. INTRODUCTION

This Groundwater Assessment Report for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, summarizes results of the activities conducted under the approved Groundwater Assessment Plan for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky, PRS/PROJ/0006/R2. The C-746-U Landfill was placed in assessment in response to the finding that some constituents were found in groundwater samples from monitoring wells (MWs) located in the vicinity of the landfill at above benchmark levels.

Only two constituents [beta activity and trichloroethene (TCE)] are present above their respective maximum contaminant level (MCL) in Regional Gravel Aquifer (RGA) groundwater in the vicinity of the C-746-U Landfill. Additional evaluation of the data for these constituents is presented in this appendix that demonstrates that these constituents are migrating from upgradient sources.

Although polychlorinated biphenyls (PCBs) historically have been present above the U.S. Environmental Protection Agency MCL, they have not exceeded the MCL in RGA wells since 2010. The studies undertaken to evaluate PCBs (as reported elsewhere in this report) demonstrate that the source of the historical PCBs was cross-contamination from well rehabilitation activities.

Other constituents that are present in RGA wells at above background levels or possibly above background levels include calcium, conductivity, dissolved solids, magnesium, sodium, and sulfate. The higher concentrations of these constituents typically are found in upgradient wells MW372 and MW373, and the concentrations in these upgradient wells are higher than concentrations present in wells located further upgradient, sidegradient, or downgradient of the C-746-U Landfill. Thus, there is no identified upgradient source that can account for the concentrations in MW372 and MW373, wells that are located upgradient of the C-746-U Landfill. Migration from potential sources (including migration from UCRS soils near the well, well fouling, compromised well completions, C-746-U Landfill leachate, and C-746-S leachate, etc.) has been considered. This appendix presents a discussion of the extent of constituents in wells located in the vicinity of the C-746-U Landfill and the potential sources for these constituents in RGA wells. Figure B.1 is a map of the wells located in the vicinity of the C-746-S&T Landfills. MW372 and MW373 are shown on both figures in a location that is considered upgradient of the C-746-U Landfills.

B.2. BETA ACTIVITY

Figure B.3 is a graph of the beta activity over time in MW357. Beta activity in this well is above the C-746-U background, but stable to declining. Figure B.4 presents a similar graph for beta activity in MW372. This upgradient well has had beta activity exceeding the MCL during the past two years; however, concentrations appear to be decreasing. It appears that the source of this beta activity in the upgradient well MW372 is an upgradient source, technetium-99 (Tc-99), present at upgradient locations, as shown in Figure B.5.

Figure B.6 is a graph of beta activity concentrations for the three RGA wells with beta activity greater than the MCL over the past two years. MW366 and MW367 are on the same flow path as water passing through MW372. Concentrations in MW366 and MW367 are lower than in upgradient well MW372 and also are decreasing; thus, the concentrations in the downgradient wells are well-explained









Figure B.3. MW357 Beta Activity since 2010



Figure B.4. Beta Activity since 2010 Upgradient Well Exceeds 50 pCi/L



Figure B.5. Technetium-99 Contours for the RGA



Figure B.6. Beta Activity since 2010 Wells with Any Exceedance of 50 pCi/L

by the upgradient source (that accounts for the MW372 concentrations). Consequently, the C-746-U Landfill is not contributing a measurable amount of beta activity to the RGA groundwater.

Figure B.7 graphs beta activity concentrations for all three wells using a four-event rolling average. The information summarized in these graphs indicates that concentrations increased first in the upgradient well and have increased more recently in MW366 and MW367, a pattern fully consistent with migration of beta activity from upgradient of the C-746-U Landfill into the C-746-U Landfill wells.

B.3. TCE CONCENTRATIONS

Figure B.8 is a graph of the TCE concentrations in upgradient wells MW372 and MW373. Concentrations in these wells exceed the MCL, but are stable to declining. TCE concentrations in wells located along the same flow path as MW372 and MW373 (but further downgradient) do not exceed the MCL.

Other TCE concentrations that have exceeded the MCL since 2010 are found in MW357, MW358, and MW361—wells located on a different flow path. Figure B.9 is a graph of recent concentrations. A comparison of this graph to the TCE plume map reproduced as Figure B.10 indicates that the sources of the TCE in both upgradient wells (MW372 and MW373) and other wells (MW357, MW358, and MW359) are migration from upgradient of the C-746-U Landfill.

B.4. SULFATE CONCENTRATIONS

Figures B.11 and B.12 present a summary of the statistical evaluation of sulfate concentrations in Upper RGA and Lower RGA wells in the vicinity of the C-746-U Landfill (DOE 2012). None of the wells has a concentration that exceeds the secondary MCL of 250 mg/L; however, the concentration in MW373 in the Third Quarter 2012 is close. The highest concentrations of sulfate in background wells are noted in upgradient wells MW372 and MW373. MW373 had a concentration of 810 mg/L in 2002. This well has been rehabilitated since then. In part because of the high concentrations in the upgradient well MW373, the only above background concentration of sulfate is at MW372.

Because of the presence of sulfate in upgradient wells, additional evaluation of the concentrations of sulfate in wells located further upgradient was conducted. Figures B.13 and B.14 present a similar statistical summary for wells located in the vicinity of the C-746-S&T Landfills. These figures demonstrate that concentrations of sulfate are much higher in MW372 and MW373 than in other upgradient, downgradient, or sidegradient wells. The C-746-S&T Landfill wells do not help pinpoint a source for the elevated concentrations in MW372 and MW373. The sulfate issues are confirmed to be associated only with the two wells located upgradient of the C-746-U Landfill (MW372 and MW373); no further upgradient source has been found sufficient to explain these concentrations.



Figure B.7. Beta Activity 4 Event Rolling Average, Log Scale



Figure B.8. TCE Concentration, Upgradient Wells



Figure B.9. TCE Concentration, All Wells w/TCE > 0.005



Figure B.10. Composite Trichloroethene Contours for the RGA for Calendar Year 2009

C-746-U Second Quarter 2012 Statistical Analysis Sulfate

URGA UNITS: mg/L

The CV is calculated to determine if background data are normally distributed. If so, the current test well results are compared to the TL. If not, a transformation is performed on the background and test well results, then each transformed test well result is compared to the transformed TL. If the test well result exceeds the TL, that is statistically significant evidence of elevated concentration in that well.

Background I Upgradient W	Data from /ells		Statistics on Background Data	
Well Number:	MW369		X= 45.031	
Date Collected	Result		S= 33.919	
3/18/2002	15.500		CV= 0.753	
4/22/2002	15.800		K factor** = 2.523	
7/15/2002	13.800		TL= 130.609	
10/8/2002	6.900	В	Because CV is less than	or equal to 1.
1/8/2003	10.500	as	ssume normal distribut	e normal distribution and continue
4/3/2003	10.500	W	vith statistical anaylsis.	
7/8/2003	10.900			
10/6/2003	16.300			
Well Number:	MW372			
Date Collected	Result			
3/19/2002	71.700			
4/23/2002	74.700			
7/16/2002	74.100			
10/8/2002	70.500			
1/7/2003	75.800			
4/2/2003	81.800			
7/9/2003	83.600			
10/7/2003	88.100			
Second Quarte April 2012	r 2012 Data (Collected in	n	
Well No. Resu	lt Gradient	Result >7	TL?	
MW357 66.00	00 Downgrad	ient N	OI	
MW360 11.00	00 Downgrad	ient N	OV	
MW363 35.00	00 Downgrad	ient N	O	
MW366 43.00	00 Sidegradie	ent N	O	
MW369 7.600) Upgradien	t N	NO	
MW372 160.0	00 Upgradien	t Y	TES	
Conclusion of	Statistical A	Analysis o	on Data	
The following evidence of ele	test well(s) evated conce	exceeded entration	the Upper Toleran with respect to bac	ce Limit, which is statistically significant kground data.
MW372				

CV Coefficient-of-Variation, CV = S/X If CV is less than or equal to 1 assume normal distribution.

S Standard Deviation, $S = [Sum ([(background result-X)^2]/[count of background results -1])]^{0.5}$

TL Upper Tolerance Limit, TL = X + (K * S)

X Mean, X = (sum of background results)/(count of background results)

^{**} Read from Table 5, Appendix B of *Statistical Analysis of Ground-Water Monitoring Data at RCRA Facilities,* Interim Guidance, EPA, 1989, based on total number of background results

C-746-U Second Quarter 2012 Statistical Analysis Sulfate

The CV is calculated to determine if background data are normally distributed. If so, the current test well results are compared to the TL. If not, a transformation is performed on the background and test well results, then each transformed test well result is compared to the transformed TL. If the test well result exceeds the TL, that is statistically significant evidence of elevated concentration in that well.

Background D Upgradient W	ata from fells	St Ba	atistics on ackground Data			Transformed I Data from Upg	Background gradient Wells
Well Number:	MW370	X=	= 122.381			Well Number:	MW370
Date Collected	Result	S=	= 195.095			Date Collected	LN(Result)
3/17/2002	17.400	C	V= 1.594			3/17/2002	2.856
4/23/2002	37.900		$actor^* = 2.523$			4/23/2002	3.635
7/15/2002	15.700	11	L= 014.000	J		7/15/2002	2.754
10/8/2002	10/8/2002 13.400		Because CV is greater than 1, the natural			10/8/2002	2.595
1/8/2003	14.400	loga	logarithm of background and test well results			1/8/2003	2.667
4/3/2003	18.100	were	were calculated.			4/3/2003	2.896
7/9/2003	9.600	S+	atistics on	1		7/9/2003	2.262
10/6/2003	16.500		ransformed			10/6/2003	2.803
Well Number:	MW373	Ba	ackground Data			Well Number:	MW373
Date Collected	Result	X=	= 3.985			Date Collected	LN(Result)
3/18/2002	163.300	S=	= 1.323			3/18/2002	5.096
4/23/2002	809.600		V- 0 222			4/23/2002	6.697
7/16/2002	109.400	C	v = 0.332			7/16/2002	4.695
10/8/2002	110.600	K	factor** = 2.523			10/8/2002	4.706
1/7/2003	113.700	T	L = 7.322			1/7/2003	4.734
4/2/2003	133.000			-		4/2/2003	4.890
7/9/2003	182.100					7/9/2003	5.205
10/7/2003	193.400					10/7/2003	5.265
Second Quarter 2012 Data Collected April 2012					Transfor Collecte	rmed Second Q d in April 2012	arter 2012 Data
Well No. Resu	lt Gradient	Result >	TL?		Well Nun	nber LN(Resul	t) Result $>$ TL?
MW358 92.0	00 Downgr	adient N	N/A		MW358	4.522	NO
MW361 80.0	00 Downgr	adient N	N/A		MW361	4.382	NO
MW364 62.0	00 Downgr	adient N	N/A		MW364	4.127	NO
MW367 31.0	00 Sidegra	dient N	N/A		MW367	3.434	NO
MW370 18.0	00 Upgradi	ient N	N/A		MW370	2.890	NO
MW373 240.	000 Upgradi	ient N	N/A		MW373	5.481	NO
Conclusion of	Statistical A	nalysis on	Transformed Da	ta			
None of the test that these well	st wells exce s have no el	eded the Uj evated conc	pper Tolerance L centrations with	limit, which is respect to back	statistica ground	ally significant data.	evidence

CV Coefficient-of-Variation, CV = S/X If CV is less than or equal to 1 assume normal distribution.

S Standard Deviation, $S = [Sum ([(background result-X)^2]/[count of background results -1])]^{0.5}$

TL Upper Tolerance Limit, TL = X + (K * S)

X Mean, X = (sum of background results)/(count of background results)

^{**} Read from Table 5, Appendix B of *Statistical Analysis of Ground-Water Monitoring Data at RCRA Facilities,* Interim Guidance, EPA, 1989, based on total number of background results

C-746-S and C-746-T Third Quarter 2012 Statistical Analysis LRGA Sulfate UNITS: mg/L

The CV is calculated to determine if background data are normally distributed. If so, the current test well results are compared to the TL. If not, a transformation is performed on the background and test well results, then each transformed test well result is compared to the transformed TL. If the test well result exceeds the TL, that is statistically significant evidence of elevated concentration in that well.

Background D Upgradient W	ells	Statis Backs	tics on ground Data	
Well Number:	MW395	X= 1	0.756	
Date Collected	Result	S= 2.	147	
8/13/2002	10.300	$CV = V f_{CV}$	0.200	
9/16/2002	9.100		16173	
10/16/2002	8.800	TL-	10.175	
1/13/2003	9.000	Because	e CV is less than	or equal to 1,
4/10/2003	8.300	assume	normal distribut	ion and continue
7/16/2003	8.200	with Sta	uisucai allayisis.	
10/14/2003	8.300			
1/13/2004	8.200			
Well Number:	MW397			
Date Collected	Result			
8/13/2002	14.000			
9/16/2002	12.800			
10/17/2002	12.300			
1/13/2003	12.700			
4/8/2003	12.800			
7/16/2003	13.100			
10/14/2003	12.100			
1/13/2004	12.100			
Third Quarter July 2012	· 2012 Data C	ollected in]	
Well No. Resu	ılt Gradient	Result > TL?	-	
MW370 18.0	00 Downgrad	lient YES		
MW373 230.	00 Downgrad	lient YES		
MW385 18.0	00 Sidegradie	ent YES		
MW388 20.0	00 Downgrad	lient YES		
MW392 6.20	0 Downgrad	lient NO		
Conclusion of	Statistical A	nalysis on Dat	a	
The following evidence of ele	test well(s) e vated concer	xceeded the U ntration with	pper Tolerand respect to bacl	ee Limit, which is statistically significant sground data.
MW370				
MW373				
MW385				
MW388				

CV Coefficient of Variation, CV = S/X If CV is less than or equal to 1 assume normal distribution.

S Standard Deviation, $S = [Sum ([(background result-X)^2]/[count of background results -1])]^{0.5}$

TL Upper Tolerance Limit, TL = X + (K * S)

X Mean, X = (sum of background results)/(count of background results)

^{**} Read from Table 5, Appendix B of *Statistical Analysis of Ground-Water Monitoring Data at RCRA Facilities,* Interim Guidance, EPA, 1989, based on total number of background results

C-746-S and C-746-T Third Quarter 2012 Statistical Analysis URGA Sulfate UNITS: mg/L

The CV is calculated to determine if background data are normally distributed. If so, the current test well results are compared to the TL. If not, a transformation is performed on the background and test well results, then each transformed test well result is compared to the transformed TL. If the test well result exceeds the TL, that is statistically significant evidence of elevated concentration in that well.

Backgrou Upgradie	nd Data nt Wells	a from s		Statist Backg			
Well Numb	ber: N	1W220	X= 10.481				
Date Collec 10/14/20 1/15/200	cted 02 3	Result 10.400 9.800		S= 2.0 CV= 0 K fact	648 0.253 tor** = 2.523		
4/10/200	3	15.400		TL=	17.161		
7/14/200	3	14.900		Because	e CV is less than	or equal to 1,	
10/13/20	03	13.500		assume	normal distribution and continu		
1/13/200	4	10.300		with statistical anaylsis.			
4/13/200	4	14.300					
7/21/200	4	10.500					
Well Numb	ber: N	1W394					
Date Colled	cted	Result					
8/13/200	2	11.200					
9/16/200	2	8.300					
10/16/20	02	8.000					
1/13/200	3	8.500					
4/10/200	3	7.900					
7/16/200	3	8.400					
10/14/20	03	8.200					
1/13/200	4	8.100					
Third Qu July 2012	arter 2(012 Data (Collected	in]		
Well No.	Result	Gradient	Resul	t > TL?			
MW221	13.000	Sidegradi	ent	NO			
MW222	11.000	Sidegradi	ent	NO			
MW223	12.000	Sidegradi	ent	NO			
MW224	15.000	Sidegradi	ent	NO			
MW369	9.400	Downgra	dient	NO			
MW372	160.00	Downgra	dient	YES			
MW384	21.000	Sidegradi	ent	YES			
MW387	37.000	Downgra	dient	YES			
MW391	12.000	Downgra	dient	NO			
Conclusion	1 of Sta	atistical A	analysis	on Dat	a		

The following test well(s) exceeded the Upper Tolerance Limit, which is statistically significant evidence of elevated concentration with respect to background data.
MW372
MW384
MW387

CV Coefficient of Variation, CV = S/X If CV is less than or equal to 1 assume normal distribution.

S Standard Deviation, $S = [Sum ([(background result-X)^2]/[count of background results -1])]^{0.5}$

TL Upper Tolerance Limit, TL = X + (K * S)

X Mean, X = (sum of background results)/(count of background results)

** Read from Table 5, Appendix B of *Statistical Analysis of Ground-Water Monitoring Data at RCRA Facilities,* Interim Guidance, EPA, 1989, based on total number of background results
B.5. OTHER CONSTITUENTS

Other constituents present in upgradient wells MW372 and MW373 that do not have an identified upgradient source that accounts for these concentrations include boron, calcium, conductivity, dissolved solids, magnesium, manganese, potassium, sodium, and sulfate. Table B.1 compares mean concentrations of constituents present in RGA wells at Paducah Gaseous Diffusion Plant (PGDP) with the following:

- The mean concentrations for those same constituents in RGA wells identified as upgradient to the C-746-S&T Landfills (per the permit-required statistical evaluation for the C-746-S&T Landfill). These locations are further upgradient of the C-746-U Landfill.
- The mean background concentrations for RGA wells located upgradient of the C-746-U Landfill (from 2002-2003), without including the contribution of upgradient wells MW372 or MW373.
- The mean background concentrations for RGA wells located upgradient of the C-746 Landfill (from 2002-2003), including MW372 and MW373 [i.e., as reported in the 3rd C-746-U Landfill report, *C-746-U Contained Landfill Third Quarter Calendar Year 2012 (July–September) Compliance Monitoring Report Paducah Gaseous Diffusion Plant, Paducah, Kentucky*, PAD-ENM-0067/V3].
- The mean 3rd Quarter 2012 concentrations for all RGA wells in the vicinity of the C-746-U Landfill, except MW372 and MW373.
- The mean 3rd Quarter 2012 concentrations for MW372 and MW373.

The comparisons provided in this table demonstrate that concentrations of these constituents are similar for all of these groups, except the concentrations of boron, calcium, conductivity, dissolved solids, magnesium, potassium, sodium, and sulfate in MW372 and MW373 are above levels seen in other locations in the vicinity of the C-746-U and C-746 S&T Landfills. Some conditions local to MW372 and MW373 may be the source(s) of these constituents.

The mean concentrations of these constituents are within the range of concentrations found at RGA wells in the vicinity of PGDP. The number of samples and number of nondetects also are presented to demonstrate that most of these constituents are routinely detected in RGA groundwater at PGDP and the range of concentrations found at PGDP is sufficient to explain most of the variability without any C-746-U Landfill source.

Observations that highlight these trends include the following.

- Most constituents are frequently detected in the vast majority of RGA samples at PGDP, indicating their typical presence in native waters.
- Calcium concentrations average 28 mg/L at PGDP. Concentrations in C-746-S&T upgradient wells and concentrations in C-746-U upgradient wells (without MW372 and MW373) average 25 mg/L. The third quarter concentrations for MW372/MW373 average 73 mg/L; thus, concentrations in MW372/MW373 are above what might be considered typical for PGDP RGA groundwater. There is no indication of a measurable source of calcium from the C-746-U Landfill.
- Chloride concentrations average 46 mg/L for all PGDP RGA well samples, but also average 50 mg/L in the C-746-S&T Landfill upgradient RGA wells—wells located upgradient of the C-746-S&T

Constituent (units)	Number of Nondetects	Number of Samples	Mean Conc. Detect ^a	Mean Conc. S&T Upgradient ^b	Mean Conc. Upgradient ^c w/o MW372/373	Mean Conc. U (as reported) Upgradient ^d	Mean 3 rd Qtr. Conc. U w/o 372/373 ^e	3 rd Qtr. Mean Conc. of MW372 and MW373 ^f	Comments
Boron ^g (mg/L)	1050	1211	0.69	0.54	0.88	1.1	0.23	1.6	Concentrations below RSL. Elevated concentrations in MW372/MW373 not above background. Detection limit has impact on statistics. Detection limit decreased by order of magnitude since 2003.
Calcium (mg/L)	1	1929	28	25	28	38	25	73	No RSL. Concentration similar for all PGDP groups without 372/373; above background in 372/373. C-746-U Landfill not the source. Upgradient concentrations of S&T not elevated.
Chloride (mg/L)	3	1957	46	50	50	45	29	46	Concentrations below SMCL. Concentration similar for all wells at PGDP. No C-746-U-Landfill impact.
Cobalt (mg/L)	1030	1719	0.019	0.016	0.032	0.026	0.0041	0.0010	Some concentrations above low RSL. Not above background; concentrations consistent with PGDP-wide variation; historical concentrations higher. No apparent C-746-U source.
Dissolved Solids (mg/L)	173	1583	262	226	255	321	253	590	SMCL of 500 mg/L exceeded only in MW373 since 2010. Above background in MW372/MW373. No apparent C-746-U impact.
Iron (mg/L)	599	2038	3.7	0.65	9.3	8.3	1.7	0.3	Mean Concentration less than RSL. Concentration lower in MW372/373. Concentrations within range at PGDP. No apparent C-746-U impact.
Magnesium (mg/L)	1	1929	11	9.9	11	15	10	29	No RSL. Above background in MW372 only. Markedly higher concentration in MW372/MW373, but not in S&T upgradient wells.
Manganese (mg/L)	428	1952	0.85	0.21	0.82	0.75	0.27	0.035	Above RSL of 0.32 mg/L. Historical issue at MW372/MW373 has subsided. Concentrations within range typical of PGDP. No apparent C-746-U source.
Sodium (mg/L)	2	1880	41	33	47	48	40	65	No RSL. Not above background. Somewhat higher concentration in MW372/MW373. No apparent C-746-U source.
Sulfate (mg/L)	50	1684	33	11	15	84	43	195	SMCL of 250 mg/L last exceeded in 2005. Above background in MW372. Markedly higher concentration in MW372/MW373.
Conductivity: fie (µmho/cm)	ld parameter	N/A	N/A	380	455	546	427	903	Above background in MW372. Markedly higher values in MW372/MW373.

Table B.1. Summary of Concentrations in RGA PGDP Wells

^aMean concentration of parameter from database for RGA wells; detected quantities only. ^bMean concentration of parameter for upgradient (background) C-746-S&T Landfills wells (includes nondetect values). ^cMean concentration of parameter for upgradient C-746-U Landfill wells without MW372 or MW373 concentrations included in mean calculation. ^dMean concentration of parameter for upgradient C-746-U Landfill wells including MW372 and MW373 concentrations. ^eMean concentration of parameter in third quarter 2012 in C-746-U Landfill well except for MW372 and MW373. ^fMean concentration of parameter in third quarter 2012 in MW372 and MW373. ^gThird quarter boron results include results only from LRGA wells MW358, MW361, MW364, MW367, MW370, and MW373.

- Landfills and further upgradient of the C-746-U Landfill. Concentrations upgradient of the C-746-U Landfill (whether they include MW372/MW373) also average 46 mg/L. Thus, it appears that chloride in the vicinity of the C-746-U Landfill has a similar concentration profile to chloride found elsewhere in the RGA at PGDP, including concentrations of chloride located further upgradient of the C-746-U Landfill and MW372 and MW373.
- Dissolved solids concentrations average 262 mg/L at PGDP. Concentrations in C-746-S&T upgradient wells and concentrations in C-746-U upgradient wells (without MW372 and MW373) average 226-255 mg/L. The third quarter concentrations for MW372/MW373 average 590 mg/L. Table B.1 compares mean concentrations of constituents present in RGA wells at PGDP with the following: thus, concentrations in MW372/MW373 are above what might be considered typical for PGDP RGA groundwater. These elevated concentrations are not the result of contributions from the C-746-U Landfill because MW372 and MW373 are located upgradient of the landfill.
- Conductivity in the upgradient C-746-S&T Landfill and C-746-U Landfill samples (not including MW372/MW373) average 380–455 µmhos/cm; but, conductivity in MW372/MW373 averages 893 µmhos/cm for the 3rd Quarter 2012. The third quarter concentrations for MW372/MW373 are above what might be considered typical for PGDP RGA groundwater. These elevated concentrations are not the result of contributions from the C-746-U Landfill because MW372 and MW373 are located upgradient of the landfill.
- Magnesium concentrations average 11 mg/L at PGDP. Concentrations in C-746-S&T upgradient wells and concentrations in C-746-U upgradient wells (without MW372 and MW373) average 9.9-11 mg/L. The third quarter concentrations for MW372/MW373 average 29 mg/L; thus, concentrations in MW372/MW373 are above what might be considered typical for PGDP RGA groundwater. These elevated concentrations are not the result of contributions from the C-746-U Landfill because MW372 and MW373 are located upgradient of the landfill.
- Sulfate concentrations average 33 mg/L in the RGA at PGDP. Concentrations upgradient to the C-746-S&T and C-746-U Landfills (not including MW372/MW373) average somewhat lower than that (12 mg/L); however, concentrations in MW372 and MW373 average 195 mg/L. This average is well above the slight concentration variations noted in other C-746-U Landfill wells that appear to be within the range of PGDP background. These elevated concentrations are not the result of contributions from the C-746-U Landfill because MW372 and MW373 are located upgradient of the landfill.

The source(s) of the elevated levels of calcium, conductivity, dissolved solids, magnesium, and sulfate in MW372 and MW373 are uncertain; however, due to the location of these wells upgradient of the C-746-U Landfill, the C-746-U Landfill is not the source. Additional evaluation of these two wells may identify the sources of these constituents.

B.6. COMPARISON TO CONSTITUENTS PRESENT IN LEACHATE

Table B.2 provides a review of the concentrations of constituents that have been evaluated in connection with the C-746-U Landfill, compared to concentrations of constituents found in the C-746-U Landfill leachate. If the C-746-U Landfill had impacts on the RGA groundwater, one principal mechanism would be via migration of the landfill leachate through the liner and through the underlying UCRS.

Constituent (units)	Mean Conc. C-746-U Landfill Leachate (mg/L) ^a	Mean Conc. RGA Wells at PGDP (mg/L) ^a	3 rd Quarter 2012 Mean Conc. of MW372 and MW373 (mg/L) ^a	Comments
Beta activity (pCi/L)	79	291	43	Kentucky MCL of 50 pCi/L. Although landfill leachate could be a source of some beta activity, it cannot fully account for an exceedance of the Kentucky MCL in RGA groundwater. The DAF estimated for the PGDP UCRS is approximately 58 (and would be orders of magnitude higher for migration through the landfill liner). Thus, beta activity would have to be orders of magnitude higher than it is to account for the RGA concentrations The wells with the beta activity exceedance (MW372 and MW373) are upgradient-to-the-C-746-U Landfill wells. Based on the DAF and leachate concentrations, there is no measurable C-746-U Landfill source of beta activity.
Beta activity (since January 2010) (pCi/L)	38	562	43	Kentucky MCL of 50 pCi/L. Beta activity not elevated above the MCL in leachate over last 2+ years. As above, there is a DAF of 58+; thus, no measurable C-746-U source.
Boron	0.43	0.69	1.6	Concentrations below RSL in RGA wells. Boron concentrations are lower in leachate than in RGA wells; however, concentrations are higher in upgradient wells MW372/MW373. Thus, no apparent C-746-U source, but some indication of higher concentrations in upgradient wells from other sources.
Calcium	107	28	73	No RSL. Although calcium is elevated in leachate, the leachate concentrations would have to be orders of magnitude higher to account for the RGA concentrations given the DAF of 58+. There are non-C-746-U Landfill leachate sources contributing to upgradient MW372 and MW373 concentrations. Note: the leachate sample may have suspended calcium (whole leachate sample). No apparent C-746-U source.
Chloride	39	46	46	Concentrations below SMCL of 250 mg/L. Chloride concentrations lower in in leachate than in RGA wells. Thus, the leachate is not the source of the RGA groundwater chloride concentrations. No apparent C-746-U source.
Cobalt	0.015	0.019	0.0010	Although cobalt concentrations exceed RSL, cobalt concentrations in leachate not elevated above concentrations typical of PGDP RGA groundwater. No apparent C-746-U source.
Dissolved solids	622	262	590	Dissolved solids concentrations above SMCL in leachate but at levels that are comparable to concentrations in upgradient wells MW372 and MW373. The DAF estimated for PGDP is 58+; thus, the dissolved solids concentrations in leachate would have to be orders of magnitude higher than observed to account for the RGA concentrations found in downgradient wells. No apparent C-746-U source.
Iron	2.8	0.65	0.3	Iron concentration mean below RSL in leachate. The DAF estimated for PGDP is 58+; thus, the dissolved solids concentrations in leachate would have to be orders of magnitude higher than observed to account for the RGA concentrations found in downgradient wells. Concentrations in C-746-U Landfill wells within range typical of PGDP. No apparent C-746-U source.

Table B.2. Comparison of Leachate Concentrations to RGA Groundwater Concentrations

Table B.2. Comparison of Leachate Concentrations to RGA Groundwater Concentrations (Continued)

Constituent (units)	Mean Conc. C-746-U Landfill Leachate (mg/L) ^a	Mean Conc. RGA Wells at PGDP (mg/L) ^a	3 rd Quarter 2012 Mean Conc. of MW372 and MW373 (mg/L) ^a	Comments
Magnesium	28	11	29	No RSL. Magnesium concentrations in leachate comparable to levels in upgradient wells MW372/MW373. The DAF for native materials is approximately 58. No apparent C-746-U
				source.
Manganese	1.2	0.85	0.035	
Sodium	63	41	65	No RSL. Sodium concentrations in leachate comparable to concentrations found in upgradient wells MW372/MW373. DAF for native materials of 58. No apparent C-746-U source.
Sulfate	198	33	195	Mean concentrations below SMCL. Sulfate concentrations in leachate comparable to concentrations found in upgradient wells MW372/MW373. No apparent C-746-U source.
Trichloroethene	Nondetect	N/A	0.0054	MCL of 0.005 mg/L. No detectable TCE in leachate. Upgradient MW372/MW373 average just above MCL of 0.005 mg/L. C-746-U is not a source of TCE to the RGA.

^aExcept for beta activity

As shown in Table B.2, there is no "fingerprint" that can be associated with the C-746-U Landfill leachate. Leachate concentrations of many constituents typically are the same order of magnitude as the mean concentrations found in RGA wells in the vicinity of PGDP [as identified in the Groundwater Operable Unit Feasibility Study (2001) and updated in the Soils Operable Unit Remedial Investigation (2013)].

Mean calcium and sulfate concentrations in leachate are higher than the other constituents, but the concentrations in leachate may have contributions from suspended solids (because leachate samples are not collected using low-flow-equivalent techniques). Overall, none of the constituents present in leachate is found at a concentration that accounts for more than a fraction of the concentration of the RGA constituents because the dilution attenuation factor (DAF) estimated for native UCRS materials at PGDP is 58, and the DAF for a lined landfill (like the C-746-U Landfill) is expected to be orders of magnitude higher.

Constituents present in the UCRS have the potential to migrate to the RGA. This migration will be attenuated based upon the nature of the materials through which the migration is occurring. For example, migration through a clay matrix will be slower than through a sandy matrix. Modeling performed at PGDP has estimated the UCRS attenuation factor at 58 (UCRS concentrations are attenuated by a factor of 58 as the constituent migrates to the RGA). A landfill liner would attenuate the leachate better than the native clay, silt, and sand materials present in the UCRS. Thus, for the constituents present in C-746-U Landfill leachate to account for the concentrations found in the RGA in the vicinity of the C-746-U Landfill, the leachate concentrations would have to be orders of magnitude higher than they are to migrate through both the landfill liner and UCRS. Thus, the C-746-U Landfill is not the source of the RGA groundwater constituents found in wells in the vicinity of the C-746-U Landfill.

B.7 COMPARISON TO CONSTITUENTS IN C-746-S LEACHATE

Table B.3 provides a review of the concentrations of constituents that have been evaluated in association with the C-746-U Landfill Groundwater Assessment and compares them to concentrations found in the S-Landfill leachate. As shown in that table, there is no "fingerprint" that can be associated with the S-Landfill leachate. The table compares S-Landfill leachate concentrations to U-Landfill leachate concentrations, mean concentrations of constituents in PGDP wells, and the mean concentrations in MW372/MW373 (wells located upgradient of the U-Landfill but downgradient of the S-Landfill) for the 3rd Quarter 2012. With the exception of manganese, concentrations of constituents are typically within an order of magnitude as the mean concentrations found in the U-Landfill leachate and do not indicate a source to RGA wells at the C-746-U Landfill.

As in the discussion for the C-746-U-Landfill presented above, constituent concentrations may be more elevated in leachate because the concentrations in leachate may have contributions from suspended solids (as leachate samples are not collected using low-flow-equivalent techniques). Overall, none of the constituents present in leachate is found at a concentration that accounts for more than a fraction of the concentration of the RGA constituents in that the DAF estimated for native materials underlying the S-Landfill is 58.

Constituent (units)	Mean Conc. S-Landfill Leachate (mg/L) ¹	Mean Conc. U-Landfill Leachate(mg/L) ¹	Mean Conc. RGA Wells at PGDP (mg/L) ¹	Mean Conc. of MW372 and MW373(mg/L) ¹	Comments
Beta Activity	24	79	29	43	S-Landfill leachate has beta activity at levels similar to RGA well
(pCi/L)					activity. S-Landfill beta activity has not exceeded the MCL since
					2005. The S Landfill is not a source of gross beta to RGA
E					groundwater at measurable levels.
Boron ³	Nondetect at 0.2-2.0	0.43	0.75	1.6	No detectable boron in S-Landfill leachate. The S Landfill is not a source of boron to the RGA, including MW372 or MW373.
Calcium	167	98	28	73	Although calcium is elevated in leachate, there are nonleachate
					sources contributing to MW372 and MW373 concentrations
					because the DAF is 58, minimum. Leachate sample may have
					suspended calcium (whole leachate sample).
Chloride	22	39	46	46	Chloride concentration not elevated in leachate. Chloride
					concentrations lower in S-Landfill leachate than typical RGA well
	0.0002	0.015	0.010	0.0010	concentrations.
Cobalt	0.0082	0.015	0.019	0.0010	Cobalt concentration not elevated in leachate. Cobalt
					concentrations less in S-Landfill and below typical RGA
Dissolved Solida	692	622	262	595	Dissolved solids concentrations only slightly elevated in leachete
Dissolved Solids	005	022	202	385	The DAE estimated for PGDP is approximately 58 Little
					evidence that S-L andfill is source of significant dissolved solids
Iron	8.3	2.8	28	0.3	Iron concentration not elevated in leachate
Magnesium	23	28	11	28	Magnesium concentration slightly elevated in leachate, but not at
magnesium	20	20		20	levels that contribute to RGA concentrations as DAF for native
					materials, is approximately 58.
Manganese	1.1	1.2	0.091	0.035	
0					
Sodium	27	63	41	65	S-Landfill leachate sodium concentrations below background
					levels in RGA wells.
Sulfate	181	198	33	195	Sulfate concentrations lower in S-Landfill; sulfate slightly
					elevated in leachate, but not at levels that contribute significantly
					to RGA concentrations.
Trichloroethene	Nondetect	Nondetect	N/A	0.0054	No detectable TCE in leachate. The S-Landfill is not a source of
					TCE to the RGA in the vicinity of the U-Landfill.

Table B.3 Comparison of S-Landfill Leachate Concentrations to RGA Groundwater Concentrations

¹Except for beta activity, however, additional discussion is provided because levels are approaching the SMCL.

Although concentrations of manganese are much higher in the S-Landfill leachate, the concentrations of manganese are not elevated in wells, like MW372 and MW373, that are located downgradient of the S-Landfill. Thus, there is no indication that manganese (or any other S-Landfill leachate constituent) is migrating from the S-Landfill at concentrations that cause a measurable impact on the RGA groundwater in the vicinity of the C-746-U Landfill.

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX C

HOME SITE WELL WATER SAMPLE ANALYTICAL RESULTS

ABW2W12-08		from: AB	W2	on 1/	23/2009	Media:	WG	SmpMethod: GR	
Comments:									
Analysis	Results	Counting Error	Units	Result Qual	Foot R Note	eporting Limit	TPU	Method	V/V/A*
				–	_			014/040.00	
	2		mg/L	U	2			SVV846-9056	/ X
Chloride	14		mg/L		2			SW846-9056	/ X
Fluoride	0.16		mg/L	±1.1	0.	1		9214	/ X /
Nitrate as Nitrogen	1		mg/L	*U	1			SVV846-9056	/ X / R-
Sunate	2		mg/∟	U	2			311840-9056	/ X
METAL									
Aluminum	3.76		mg/L		0.	2		SW846-6010B	I / X
Antimony	0.005		mg/L	UB	0.	005		SW846-6020	/ X
Arsenic	0.0691		mg/L		0.	01		SW846-6020	I / X
Barium	0.25		mg/L		0.	005		SW846-6020	/ X
Beryllium	0.001		mg/L	U	0.	001		SW846-6020	/ X
Boron	0.2		mg/L	U	0.	2		SW846-6010B	/ X
Cadmium	0.001		mg/L	U	0.	001		SW846-6020	/ X
Calcium	36.4		mg/L		1			SW846-6010B	/ X
Chromium	0.1		mg/L	UN	0.	1		SW846-6020	/ X
Cobalt	0.0129		mg/L		0.	01		SW846-6020	/ X
Copper	0.2		mg/L	U	0.	2		SW846-6020	/ X
Iron	14.3		mg/L		0.	1		SW846-6010B	I / X
Lead	0.0147		mg/L		0.	0013		SW846-6020	/ X
Magnesium	7.77		mg/L		0.	025		SW846-6010B	/ X
Manganese	2.28		mg/L		0.	05		SW846-6020	I / X
Mercury	0.0002		mg/L	U	0.	0002		SW846-7470A	/ X
Molybdenum	0.00115		mg/L		0.	001		SW846-6020	/ X
Nickel	0.05		mg/L	U	0.	05		SW846-6020	/ X
Potassium	19.4		mg/L		0.	2		SW846-6010B	/ X
Rhodium	0.005		mg/L	UB	0.	005		SW846-6020	/ X
Selenium	0.005		mg/L	UBX	0.	005		SW846-6020	/ X
Silver	0.001		mg/L	UB	0.	001		SW846-6020	/ X
Sodium	14.2		mg/L		1			SW846-6010B	/ X
Tantalum	0.005		mg/L	U	0.	005		SW846-6020	/ X
Thallium	0.002		mg/L	UX	0.	002		SW846-6020	/ X
Uranium	0.00608		mg/L		0.	001		SW846-6020	/ X
Vanadium	0.2		mg/L	U	0.	2		SW846-6020	/ X
Zinc	0.2		mg/L	U	0.	2		SW846-6020	/ X
METAL-D									
Barium, Dissolved	0.0637		mg/L		0.	005		SW846-6020	/ X
Chromium, Dissolved	0.01		mg/L	U	0.	01		SW846-6020	/ X
Uranium, Dissolved	0.001		mg/L	U	0.	001		SW846-6020	/ X
PHYSC									
Dissolved Solids	238		mg/L		45	5		EPA-160.1	/ X
РРСВ									
PCB-1016	0.17		ug/L	U	0.	17		SW846-8082	/ X
PCB-1221	0.18		ug/L	U	0.	18		SW846-8082	/ X
PCB-1232	0.14		ug/L	U	0.	14		SW846-8082	/ X
PCB-1242	0.1		ug/L	U	0.	1		SW846-8082	/ X
PCB-1248	0.12		ug/L	U	0.	12		SW846-8082	/ X
PCB-1254	0.07		ug/L	U	0.	07		SW846-8082	/ X
PCB-1260	0.05		ug/L	U	0.	05		SW846-8082	/ X
PCB-1268	0.09		ug/L	U	0.	09		SW846-8082	/ X

Paducah OREIS Report for EMSPGW09-01

*Verification/Validation/Assessment

		Paducah (DREIS Re	port for	EMSPGV	V09-01		
Polychlorinated biphenyl	0.18		ug/L	UX	0.18		SW846-8082	/ X /
RADS								
Alpha activity	-5.83	3.89	pCi/L	U	33.3	4.05	SW846-9310	/ X /
Beta activity	51.8	7.79	pCi/L	U	63	9.88	SW846-9310	/ X /
Iodine-131	-2.93	5.85	pCi/L	U	6.43	5.85	RL-7124	/ X /
Radium-226	0.325	0.404	pCi/L	U	0.68	0.449	RL-7129	/ X /
Strontium-90	1.13	0.154	pCi/L	U	1.27	0.191	RL-7140	/ X /
Technetium-99	1.83	10.1	pCi/l	U	15.2	10.1	RI -7100	/ X /
Thorium-230	0.0906	0.154	pCi/l	U	1.81	0.766	RI -7128	/ X /
Tritium	220	736	pCi/L	UB	264	736	RL-7155	/ X /
SVOA								
1.2-Dichlorobenzene	5		ua/L	U	5		SW846-8260B	/ X /
1,4-Dichlorobenzene	5		ug/L	U	5		SW846-8260B	/ X /
VOA								
1.1.1.2-Tetrachloroethane	5		ua/L	U	5		SW846-8260B	/ X /
1.1.1-Trichloroethane	1		ug/L	U	1		SW846-8260B	/ X /
1 1 2 2-Tetrachloroethane	5		ug/l	U	5		SW846-8260B	/ X /
1 1 2-Trichloroethane	1		ug/L		1		SW846-8260B	/ X /
1 1-Dichloroethane	1		ug/L		1		SW846-8260B	1 × 1
	1		ug/∟	0	1		SW040-0200B	
	I F		ug/L	0	1 E		SW040-0200D	
1,2,3-Thenoropropane	5		ug/L	0	5		SVV640-6200B	
	5		ug/L	0	5		SVV846-8260B	/ X /
1,2-Dichloroethane	1		ug/L	U	1		SW846-8260B	/ X /
1,2-Dichloropropane	5		ug/L	U	5		SW846-8260B	/ X /
2-Butanone	10		ug/L	U	10		SW846-8260B	/ X /
2-Hexanone	10		ug/L	U	10		SW846-8260B	/ X /
4-Methyl-2-pentanone	10		ug/L	U	10		SW846-8260B	/ X /
Acetone	10		ug/L	U	10		SW846-8260B	/ X /
Acrolein	25		ug/L	UJ	25		SW846-8260B	/ X /
Acrylonitrile	10		ug/L	U	10		SW846-8260B	/ X /
Benzene	5		ug/L	U	5		SW846-8260B	/ X /
Bromochloromethane	5		ug/L	U	5		SW846-8260B	/ X /
Bromodichloromethane	5		ug/L	U	5		SW846-8260B	/ X /
Bromoform	5		ug/L	U	5		SW846-8260B	/ X /
Bromomethane	5		ug/L	U	5		SW846-8260B	/ X /
Carbon disulfide	5		ug/L	U	5		SW846-8260B	/ X /
Carbon tetrachloride	1		ug/L	U	1		SW846-8260B	/ X /
Chlorobenzene	5		ua/L	U	5		SW846-8260B	/ X /
Chloroethane	5		ua/L	U	5		SW846-8260B	/ X /
Chloroform	5		ug/l	Ū	5		SW846-8260B	/ X /
Chloromethane	5		ug/l	U	5		SW846-8260B	/ X /
cis-1 2-Dichloroethene	1		ug/L	U U	1		SW846-8260B	/ X /
cis-1 3-Dichloropropene	5		ug/L		5		SW846-8260B	/ X /
Dibromochloromothono	5		ug/L		5		SW040-0200D	
Dibromocritoromethane	5 F		ug/L	0	5		SW040-0200D	
	5		ug/L	0	5		SVV640-6200B	
Ethylbenzene	5		ug/L	0	5		SVV846-8260B	/ X /
Iodometnane	5		ug/L	UJ	5		SVV846-8260B	/ X /
Methylene chloride	5		ug/L	U	5		SW846-8260B	/ X /
Styrene	5		ug/L	U	5		SW846-8260B	/ X /
Tetrachloroethene	1		ug/L	U	1		SW846-8260B	/ X /
Toluene	9.6		ug/L		5		SW846-8260B	/ X /
Total Xylene	15		ug/L	U	15		SW846-8260B	/ X /
trans-1,2-Dichloroethene	1		ug/L	U	1		SW846-8260B	/ X /
trans-1,3-Dichloropropene	5		ug/L	U	5		SW846-8260B	/ X /
Trans-1,4-Dichloro-2-butene	5		ug/L	U	5		SW846-8260B	/ X /
Trichloroethene	1		ug/L	U	1		SW846-8260B	/ X /

*Verification/Validation/Assessment

2/24/2009 Page 2 of 4

	Paducah O	REIS Re	eport f	or E	CMSPGW0	9-01		
5		ug/L	U		5		SW846-8260B	/ X /
10		ug/L	U		10		SW846-8260B	/ X /
2		ug/L	U		2		SW846-8260B	/ X /
73		mg/L			25		EPA-410.4	/ X /
0.05		mg/L	JU		0.05		SW846-9010C	/ X /
77		mg/L			20		EPA-160.2	/ X /
24.7		mg/L	D		1		SW846-9060	/ X /
21.3		ug/L			10		SW846-9020B	/ X /
	from: AB	W2	on 2	/12/20	09 Media:	WG	SmpMethod: GR	
Results	Counting Error	Units	Result Qual	Foot Note	Reporting Limit	TPU	Method	V/V/A*
5.6		mg/L			1		SW846-9056	/ X /
	5 10 2 73 0.05 77 24.7 21.3 Results 5.6	Paducah O 5 10 2 73 0.05 77 24.7 21.3 from: AB ¹ Results Counting Error 5.6	5 ug/L 10 ug/L 2 ug/L 73 mg/L 0.05 mg/L 77 mg/L 24.7 mg/L 21.3 ug/L	Paducah OREIS Report f 5 ug/L U 10 ug/L U 2 ug/L U 73 mg/L U 73 mg/L JU 73 mg/L JU 74 mg/L D 9005 mg/L D 9005 mg/L D 10 ug/L D 11.3 ug/L D Results Counting Error Units Result Qual 5.6 mg/L U D	Paducah OREIS Report for E 5 ug/L U 10 ug/L U 2 ug/L U 73 mg/L U 73 mg/L JU 73 mg/L JU 74 mg/L JU 75 mg/L D 24.7 mg/L D 21.3 ug/L On 2/12/20 From: ABW2 Results Counting Error Units Result Foot 5.6 mg/L Visit Note	S ug/L U 5 10 ug/L U 10 2 ug/L U 10 2 ug/L U 2 73 mg/L JU 0.05 0.05 mg/L JU 0.05 77 mg/L D 1 24.7 mg/L D 1 21.3 ug/L IO 1 From: ABW2 on 2/12/2009 Media: Results Counting Error Units Foot Reporting Limit 5.6 mg/L 1 1 1	Paducah ORE IS Report for EMSPGW09-01 5 ug/L U 5 10 ug/L U 10 2 ug/L U 2 73 mg/L U 25 0.05 mg/L JU 0.05 77 mg/L D 1 24.7 mg/L D 1 21.3 ug/L D 1 From: ABW2 on 2/12/2009 Results Counting Error Units Foot Reporting Limit TPU 5.6 mg/L 1 1 1 1	Paducah ORE IS Report for EMSPG W09-01 5 ug/L U 5 SW846-8260B 10 ug/L U 10 SW846-8260B 2 ug/L U 2 SW846-8260B 73 mg/L U 2 SW846-8260B 73 mg/L U 2 SW846-8260B 73 mg/L JU 0.05 SW846-9010C 77 mg/L JU 0.05 SW846-9010C 77 mg/L D 1 SW846-9020B 24.7 mg/L D 1 SW846-9020B 21.3 ug/L D 1 SW846-9020B From: ABW2 on 2/12/2009 Media: WG SmpMethod: GR Results Counting Error Units Foot Reporting Limit TPU Method 5.6 mg/L 1 SW846-9056 SW846-9056 SW846-9056

TBABW212-08	_	from: QC			on 1/23/2009		٧Q	SmpMethod:	
Comments:									
Analysis	Results	Counting Error	Units	Result Qual	Foot Note	Reporting Limit	TPU	Method	V/V/A*
SVOA									
1,2-Dichlorobenzene	5		ug/L	U		5		SW846-8260B	/ X
1,4-Dichlorobenzene	5		ug/L	U		5		SW846-8260B	/ X
VOA									
1,1,1,2-Tetrachloroethane	5		ug/L	U		5		SW846-8260B	/ X
1,1,1-Trichloroethane	1		ug/L	U		1		SW846-8260B	/ X
1,1,2,2-Tetrachloroethane	5		ug/L	U		5		SW846-8260B	/ X
1,1,2-Trichloroethane	1		ug/L	U		1		SW846-8260B	/ X
1,1-Dichloroethane	1		ug/L	U		1		SW846-8260B	/ X
1,1-Dichloroethene	1		ug/L	U		1		SW846-8260B	/ X
1,2,3-Trichloropropane	5		ug/L	U		5		SW846-8260B	/ X
1,2-Dibromoethane	5		ug/L	U		5		SW846-8260B	/ X
1,2-Dichloroethane	1		ug/L	U		1		SW846-8260B	/ X
1,2-Dichloropropane	5		ug/L	U		5		SW846-8260B	/ X .
2-Butanone	10		ug/L	U		10		SW846-8260B	/ X
2-Hexanone	10		ug/L	U		10		SW846-8260B	/ X .
4-Methyl-2-pentanone	10		ug/L	U		10		SW846-8260B	/ X .
Acetone	10		ug/L	U		10		SW846-8260B	/ X .
Acrolein	25		ug/L	UJ		25		SW846-8260B	/ X .
Acrylonitrile	10		ug/L	U		10		SW846-8260B	/ X .
Benzene	5		ug/L	U		5		SW846-8260B	/ X .
Bromochloromethane	5		ug/L	U		5		SW846-8260B	/ X .
Bromodichloromethane	5		ug/L	U		5		SW846-8260B	/ X .
Bromoform	5		ug/L	U		5		SW846-8260B	/ X .
Bromomethane	5		ug/L	U		5		SW846-8260B	/ X .
Carbon disulfide	5		ug/L	U		5		SW846-8260B	/ X .
Carbon tetrachloride	1		ug/L	U		1		SW846-8260B	/ X /
Chlorobenzene	5		ug/L	U		5		SW846-8260B	/ X .
Chloroethane	5		ug/L	U		5		SW846-8260B	/ X
Chloroform	5		ug/L	U		5		SW846-8260B	/ X
Chloromethane	5		ug/L	U		5		SW846-8260B	/ X
cis-1,2-Dichloroethene	1		ug/L	U		1		SW846-8260B	/ X
cis-1,3-Dichloropropene	5		ug/L	U		5		SW846-8260B	/ X
Dibromochloromethane	5		ug/L	U		5		SW846-8260B	/ X
Dibromomethane	5		ug/L	U		5		SW846-8260B	/ X
Ethylbenzene	5		ug/L	U		5		SW846-8260B	/ X
lodomethane	5		ug/L	UJ		5		SW846-8260B	/ X
Methylene chloride	5		ug/L	U		5		SW846-8260B	/ X
Styrene	5		ug/L	U		5		SW846-8260B	/ X
Tetrachloroethene	1		ug/L	U		1		SW846-8260B	/ X
Toluene	5		ug/L	U		5		SW846-8260B	/ X
Total Xylene	15		ug/L	U		15		SW846-8260B	/ X
trans-1,2-Dichloroethene	1		ug/L	U		1		SW846-8260B	/ X
trans-1,3-Dichloropropene	5		ug/L	Ű		5		SW846-8260B	/ X
Trans-1,4-Dichloro-2-hutene	5		ua/L	Ŭ		5		SW846-8260B	/ X
Trichloroethene	1		ua/l	IJ		1		SW846-8260B	/ X
Trichlorofluoromethane	5		ua/l	U		5		SW846-8260B	/ X
Vinyl acetate	10		ua/l	Ű		10		SW846-8260B	/ X
Vinvl chloride	2			U U		2		SW846-8260B	/ X
	-		ug/L	0		-		311340 0200D	1 1

Paducah OREIS Report for EMSPGW09-01

Laboratory Footnotes and Qualifiers

Footnote

- A. Insufficient uranium present in the sample to determine an assay.
- B. Maximum assay was used to calculate the MDA for total uranium activities.
- C. Normal assay was used to calculate the MDA for total uranium activities.
- D. The relative bias for the LCS is greater than 25%.
- E. Gross activities are a calculated value. Gamma activity is converted to the corresponding gross alpha/beta measurement.
- F. Insufficient sample available/provided for gross beta analysis.
- G. TIMS assay used to calculate total uranium activity.
- H. No nuclide meet criteria for gross gamma.
- I. The MDA of all principle nuclide not identified and nuclide identified were summed to provide max, reportable activity
- J. No analysis result available. Sample signal too weak.
- K. No analysis result available. Total U below reporting limit.
- L. No minor isotope determination available. Signal strength insufficient.
- M. Result is biased high and MDA is biased low due to interfering lines and/or increases in BKG due to sample activity.
- N. Measured U-235 act/mass was below MDA therefore all other cal. U isotopes & U-total will be rpt as below their resp. MDAs.
- O. Gross Gamma has no output error.
- P. The max plant assay was assumed since the calculated assay was not within the range of the plant cascade assays.
- Q. Mass of U-235 is < or = MDM, thus mass of total U/U isotopes won't be reported. Total U/U isotopes will be < their MDAs Asbestos Not Detected
- R. Cs-134 activity will be understated due to the short half-life and will exclude any previous site induced Cs-134.
- S. Gross gamma is a Cs-137 equivalence. Activity assumes branch yield and det eff of Cs-137 for all line in spectrum.
- T. Analyte is a common volatile laboratory contaminant
- T1. Sample analysis is below LCR for concent., however above report. limit for assay.
- T1Z1. Samp analysis below LCR concent, above report.limit assay/.05wt% = or >2 sigma?
- V. Method 5030A (Purge & Trap)
- W. Analyte is present at the LCR.
- X. See comments for explanation
- Y. U/U-234 act are estimated. Assay used was determined by gamma. U/U-234 results can't be used for any NCS/NMC&A purposes. Uranium
- Z. Std Dev is calculated based on controls (SRM) prepared and analyzed with each sample batch. SRM is ~0.711 wt% U-235.
- Z1. This 0.05 wt% value equal to or > 2 sigma for controls associated w/data.

Inorganic Qualifiers

- * Duplicate analysis not within control limits.
- + Method of standard additions (MSA) correlation coefficient less than 0.995.
- A Indicates that a TIC is suspected aldol-condensation product.
- B Applies when the analyte is found in the associated blank
- D All compounds identified in the analysis at the secondary dilution factor.
- E Result estimated due to interferences.
- J Indicates an estimated value
- M Duplicate injection precision not met.
- N Sample spike recovery not within control limits.
- Q No analytical result available or not required because total analyses< PQL.
- R QC indicates that data are not usable. Resampling and re-analysis are necessary for verification.
- S Result determined by method of standard additions (MSA).
- U Analyte analyzed for but not detected at or below the lowest concentration reported.
- W Post-digestion spike recovery out of control limits.
- X Other specific flags and footnotes may be required to properly define the results.

Organic Qualifiers

- A Tentatively identified compound (TIC) is suspected aldol-condensation product.
- B Compound found in blank as well as sample.
- C Compound presence confirmed by GC/MS (GC/MS flag).
- D Compounds identified in an analysis at a secondary dilution filter.
- E Result exceeds calibration range (GC/MS flag).
- J Indicates an estimated value.
- N Presumption evidence of a compound GC/MS flag).
- P Difference between results from two GC columns unacceptable.
- U Compound analyzed for but not detected at or below the lowest concentration reported.
- X Other specific flags and footnotes may be required to properly define the results.
- Y MS, MSD recovery and/or RPD failed acceptance criteria.
- Z (Reserved by CLP for a laboratory-defined organic date qualifier.)

Rad Qualifiers

- A Analyzed but not detected at the analyte quantitation limit.
- B Method blank not statistically different from sample at 95% level of confidence.
- D Sample is statistically different from duplicate at 95% level of confidence.

- J Indicates an estimated value.
- L Expected and measured value for LCS is statistically different at 95% level of confidence.
- M Expected and measured value for MS is statistically different at 95% level of confidence.
- R QC indicates that data are not usable. Resampling and reanalysis are necessary for verification.
- T Tracer recovery is < or equal to 30% or > or equal to 105%.
- U Value reported is < the MDA and/or < 2 sigma TPE.
- X Other specific flags and footnotes may be required to properly define the results.

PEMS/OREIS CODES

ΤB

Trip Blank

<u>Media C</u>	odes
AG	Soil Gas
AQ	Air Quality Control Matrix
DC	Drill Cuttings
FR	Filter Residue
FT	Filter
GR	Grout
LD	Drilling Fluid
LF	Floating/Free Product on Groundwater Table
LO	Oil, All Types (Transformer, Waste, Motor, Mineral)
LT	Liquid from tank
MD	Meteorological
MS	Metal Shavings
NA	Not Available
NW	Non-Water Liquid
QA	Aquatic Animal
QB	Aquatic Bird
QC	Aquatic (Some combination of at least 2) of bird,
	plant, animal; Excludes benthic organism
QN	Benthic Organism
QP	Aquatic Plant
SC	Cement
DIL	Laboratory dilution
SE	Sediment (associated with surface water)
SF	Filter Sandpack
SL	Sludge
SO	Soil
SP	Floor Sweepings
SQ	Soil/Solid Quality Control Matrix
SS	Scrapings
SW	Swab or Wipe
SZ	Solid Waste
TB	Terrestrial Bird
TC	Terrestrial (Some combination at least 2) of bird,
	plant, or animal.
TW	Treated Water
WC	Wall corings
WG	Groundwater
WL	Water that has leached through waste
WQ	Water Quality Control Matrix
WS	Surface Water
WW	Waste Water
WZ	Special Water Control Matrix

Smp Method Codes

?	Other, defined in COMMENTS column
CSF	Continuous Sample Flow
ES	Estimate
FPC	Flow Proportional Composite
GR	Grab
NA	Not Applicable
SC	Spatial Composite
SPLT	Split
TC	Temporal Composite

Sample Type Codes

Sample '	Type Codes		diluted with drilling fluid due to insufficient
?	Other, defined in COMMENTS column		purging prior to sampling
DI	Dejonized Water used for preparing blanks, etc.	BH-QC	Result may be biased high based upon lab
FB	Field Blank		QC (i.e. surrogate, MS/MSD, etc.)
FR	Field Replicate (Code used for Field Duplicate)	BH-RB	Result may be biased high; chemical
FTB	Filter Blank		detected in associated refrigerator blank
PRBL	Preservative blank	BH-RI	Result may be biased high, chemical
RB	Refrigerator blank		detected in associated equipment rinsate.
REG	Regular	BH-SOLID	Result biased high due to sampling
REG2	Regular sample secondary analysis		containing a large amount of solids
RED RED	Replicate	BH-SS	Results may be biased high; sample may
DED1	Replicate 1		contain particles of the acetate sampling
DI	C E-minment Directo /Decen		sleeve
KI	QC Equipment Kinsate/Decon	C 0	

Verification Codes ? Other, defined in COMMENTS column B Result exceeds background criteria I Result exceeds established criteria S Result exceeds statistical controls based on historical data T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes Particulation Codes = Validated result, which is detected and unqualified ? Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes EBH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate	TLC	Toxicity Laboratory Control Sample
 ? Other, defined in COMMENTS column B Result exceeds background criteria I Result exceeds established criteria S Result exceeds statistical controls based on historical data T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes = Validated result, which is detected and unqualified ? Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	Verifica	tion Codes
 B Result exceeds background criteria I Result exceeds established criteria S Result exceeds statistical controls based on historical data T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes = Validated result, which is detected and unqualified ? Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	?	Other, defined in COMMENTS column
 I Result exceeds established criteria S Result exceeds statistical controls based on historical data T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes Validated result, which is detected and unqualified Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information 	В	Result exceeds background criteria
 S Result exceeds statistical controls based on historical data T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes Validated result, which is detected and unqualified Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	Ι	Result exceeds established criteria
T Holding time exceeded for this analysis X Result exceeds permit limits Validation Codes	S	Result exceeds statistical controls based on historical
 X Result exceeds permit limits X Result exceeds permit limits X Result exceeds permit limits Yalidation Codes Validated result, which is detected and unqualified Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	Т	Holding time exceeded for this analysis
Validation Codes = Validated result, which is detected and unqualified ? Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes E BH-ER Result may be biased high; chemical detected in associated equipment rinseate	X	Result exceeds permit limits
 Validated result, which is detected and unqualified Other, defined in COMMENTS column Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information 	Vəlidəti	ion Codes
 ? Other, defined in COMMENTS column D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	=	Validated result, which is detected and unqualified
 D Analyte, compound or nuclide detected above the reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate 	?	Other, defined in COMMENTS column
reported detection limit, and the reported detection limit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate	D	Analyte, compound or nuclide detected above the
Imit is approximated due to quality deficiency. J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate		reported detection limit, and the reported detection
 J The analyte was positively identified; the associated numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT 		limit is approximated due to quality deficiency.
numerical value is the approximate concentration of the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high the to	J	The analyte was positively identified; the associated
the analyte in the sample. N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high the to		numerical value is the approximate concentration of
N The analysis indicates the presence of an analyte for which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high the to		the analyte in the sample.
which there is presumptive evidence to make a "tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high due to	Ν	The analysis indicates the presence of an analyte for
"tentative identification." R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high due to		which there is presumptive evidence to make a
R Result rejected by validator. U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high due to	_	"tentative identification."
U The analyte was analyzed for, but was not detected above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high due to	R	Result rejected by validator.
above the reported sample quantitation limit. X Not validated; Refer to the RSLTQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH CONT Result may be biased high due to	U	The analyte was analyzed for, but was not detected
X Not variated; Refer to the RSLIQUAL field for more information Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH-CONT Result may be biased high due to	V	above the reported sample quantitation limit.
Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH-CONT Result may be biased high due to	Χ	Not validated; Refer to the RSLIQUAL field for
Assessment Codes BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH-CONT Result may be biased high due to		more information
BH-ER Result may be biased high; chemical detected in associated equipment rinseate BH-CONT Result may be biased high due to	Assessm	ient Codes
BH CONT Result may be biased high due to	BH-ER	Result may be biased high; chemical
BH_CONT Result may be based high due to	D G O	detected in associated equipment rinseate
DIFCONT Result may be blased high due to	BH-CO	NT Result may be biased high due to
contamination of the sample from the field		contamination of the sample from the field
OF IADORATORY		or laboratory
detected in associated field blank	ри-гр	detected in associated field black
BH-FB BH-RI Result may be biased high: chemical	BH-FR	RH-RI Result may be biased high: chemical

detected in associated field blank and Result may be biased high, chemical detected in

detected in associated field blank and result may be biased high; chemical detected in

Result may be biased high; compound is a known or probable lab contaminant.

Result biased high due to laboratory process

Result may be biased high; sample may be

associated equipment rinsate.

associated trip blank.

qualifiers

Result may be biased high; chemical

Result may be biased high; chemical detected in associate field blank. See comments for additional assessment

BH-FB BH-TB

BH-FB.&

BH-LAB

BH-LABPR

BH-PURGE

PEMS/OREIS CODES

Assessment Code	es (cont.)	J	Result estimated			
		KYRHTAB-50	Kentucky Radiation Health and Toxic			
BH-IB	Result may be biased high, chemical		Agents Branch (KYRHTAB) has performed			
DU TEMD	Result biased bigh due to a temperature		an independent data evaluation (not to be			
DII-I EIVIF	exceedance		validation) and the rad error accounts for			
BL-AIR	Biased low due to air rotary drilling method		greater than 50% of the results			
BL-AIR &	Biased low due to air rotary drilling method.	KYRHTAB-ER	Kentucky Radiation Health and Toxic			
,	See comments for additional assessment		Agents Branch (KYRHTAB) has performed			
	qualifiers.		an independent data evaluation (not to be			
BL-HS	Biased low due to headspace in sample		confused with data verification and			
	container		validation) and the data presents error			
BL-HS, BL-TEM	P Biased low due to headspace in sample		problems (ie., no counting uncertainty or			
	container & result biased low due to a		zero counting uncertainty).			
	temperature exceedance.	KYRHIAB-LI	Kentucky Radiation Health and Toxic			
DL-LAD	known or probable lab contaminant		Agents Branch (KTKHTAB) has performed			
RI -LARPR	Result may be biased low due to laboratory		an independent data evaluation (not to be			
DE ENDIR	process		confused with data verification and			
BL-PRES	Result may be biased low due to improper		validation) and the results are less than (LT)			
	preservative added.		the maximum detectable activity (MDA) or			
BL-PURGE	Result may be biased low; sample may be		detection limit and should not be plotted.			
	diluted with drilling fluid due to the	KYRHTAB-NE	Kentucky Radiation Health and Toxic			
	insufficient purging prior to sampling		Agents Branch (KYRHTAB) has performed			
BL-PURGE,&	Result may be biased low; sample may be		an independent data evaluation (not to be			
	diluted with drilling fluid due to insufficient		confused with data verification and			
	additional assassment qualifiers		validation) and the ad error exhibits a			
BL-OC	Result may be biased low based upon lab	KYRHTAB-OK	Kentucky Radiation Health and Toxic			
DL-QC	OC (i e surrogate MS/MSD etc.)	KIRIIIID-OK	Agents Branch (KYRHTAB) has performed			
BL-T	Result may be biased low: sample holding		an independent data evaluation (not to be			
	time exceeded		confused with data verification and			
BL-T,J	Result may be biased low; sample holding		validation) and the data is acceptable for			
	time exceeded, estimated		use.			
BL-TEMP	Result may be biased low due to	LAB-PREP	Prep method used by the lab valid but not			
	temperature exceedance.		proceduralized.			
BL-TEMP, J	Result biased low due to a temperature	LCSEXP	LCS Expired			
BI TEMD II	exceedance, estimated. Result biased low due to a temperature	LCSNA L CSNI	Laboratory control sample not analyzed.			
DL-TEMIT, U	exceedance not detected	MDA-METHOD	The recalculated MDA is considered a			
CCCSEXP	Continuous Calibration Check Standard	MD/T METHOD	method-wide MDA. Batch specific MDAs			
	Expired		were not calculated.			
DIL	Result is obtained from dilution	MDA-RECALC	The original MDA of 21.4 pCi/L was			
DIS-EDDF1	Discrepancies between the EDD and the		calculated incorrectly and was recalculated			
	Form 1. Form 1s are generated by		during the Field Laboratory evaluation. The			
	instrument software that automatically		recalculated MDA is 24.7 pCi/L.			
	reports all detected compounds. It is the	MSMSDEXP	Matrix Spike/Matrix Spike Duplicate			
	lab's policy to not report quantities below	NT/A	Standard Expired			
	data are correct. However, the EDD format		Validation requested but qualifier not			
	data which feeds OREIS will be used for	NOVAL	provided due to missing Form I			
	reporting.	NOVAL-FLAB	Validation targeted for this project but not			
DR	Discrepancy between summary data report		required for field laboratory data.			
	and raw data.	NR	Assessment question not resolved.			
FDUP-OUT	Field duplicate exceeds the RPD criterion	PENP	PE Sample Not Performed			
ICSEXP	Initial Calibration Standard Expired	QUAL	This data should be considered qualitative			
IN-LAB	Result should be considered information		due to the sampling process, the variability			
	only. Compound is a known or probable lab		in the medium sampled or issues with the			
	contaminant. Result should be considered information	D	anarytical process.			
IIN-LAD,&	only Compound is a known or probable lab	r R-C	Result questionable credibility at issue			
	contaminant. See comments for additional	R-C?	Result questionable, credibility at issue			
	assessment qualifiers		other defined in COMMENTS column			
IN-LABQC	Result should be considered information	R-C, &	Result questionable, credibility at issue. See			
-	only. Quality control requirements of the		comments for additional assessment			
	laboratory method were not met.		qualifiers			
IN-METH	Result should be considered information					
	only. Lab utilized a modified method.					

PEMS/OREIS CODES

Assessment Codes (cont.)

R-DUPVAR	Result questionable, measured variability of the field duplicate is outside PARCC
	parameter expectations, therefore population
	estimates of variability may be off by
	several orders of magnitude.
R-H	Result unusable due to historical trending
	(i.e., other)
R-HSS	Rejected due to high suspended solids
R-MTRX	Result rejected due to matrix interference.
R-NORAD	Result unusable: Uranium-235 portion of
	calculation is below reliable detection limits.
R-NORAD.&	Result unusable: Uranium-235 portion of
, - ,	calculation is below reliable detection limits.
	See comments for additional assessment
	qualifiers
R-NTRS	Result rejected; not a true representative
	sample
R-NTRSFW	Result rejected; not a true representative
	sample of formation water.
R-PRES	Result rejected due to improper preservative added.
R-RERUN	Result unusable; results for re-analysis
	should be used
R-T	Result rejected due to missing holding time
U	Not detected
U,J	Not detected and result estimated
U-RAD	Result considered a non-detect; instrument
	measurement error is equal to or greater than
	the reported result
U-RAD,&	Result considered a non-detect; instrument
	measurement error is equal to or greater than
	the reported result, see comments for
	additional assessment qualifiers
?	Other, defined in COMMENTS column

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX D

PCB IN SOILS FIELD INVESTIGATION SUMMARY

THIS PAGE INTENTIONALLY LEFT BLANK

PCB IN SOILS FIELD INVESTIGATION

INTRODUCTION

The Groundwater Assessment Plan for the C-746-U Landfill at the Paducah Gaseous Diffusion Plant, Paducah, Kentucky (PRS/PROJ/006/R2) identified polychlorinated biphenyls (PCBs) as a potential contaminant related to the C-746-U Landfill. Planned assessment actions related to the PCBs included a field investigation to sample and analyze shallow soils for PCB contamination to evaluate if PCB soil contamination areas exist at the northern portion of the landfill permit area. This field investigation focused on the vicinity of monitoring well cluster MW363, MW364, and MW365. PCBs historically have been detected in water samples from Upper Continental Recharge System (UCRS) well MW365, screened over the depth interval 32–42 ft below ground surface (bgs), and in the Regional Gravel Aquifer (RGA) wells, MW363, screened over the depth interval 55–65 ft bgs, and MW364, screened over the depth interval 73–83 ft bgs.

SAMPLING AND ANALYSIS PLAN SUMMARY

Data quality objectives (DQOs) for the PCB field investigation, incorporated as Appendix C of the Groundwater Assessment Plan, were used to develop the sampling and analysis approach. The intent of the investigation was to collect a limited number of soil samples in the vicinity of well MW365 and, if PCB contamination of soils was present, to "step out" within the investigation area until "clean" [less than 1 part per million (ppm) PCBs] soils were encountered. Per the DQOs, sampling was to begin on north-south and east-west axes passing through MW365. The initial sample borings were to be located 30 ft from MW365, with samples collected from the depth intervals 2–4 ft, 4–6 ft, 6–8 ft,10–12 ft, 14–16 ft, and 18–20 ft. Samples were to be analyzed using the Hach PCB in Soil Pocket Colorimeter II Test Kit, using a 5 ppm detection limit. Follow-on confirmatory sampling, to be initiated once the area of PCBs greater than (or equal to) 5 ppm was established, would identify the PCB congeners present in the source zone and refine the defined extent of PCB contamination, to a level of 1 ppm or less, using fixed-base laboratory analysis.

FIELD INVESTIGATION

Following a Paducah Remediation Services, LLC, quality assessment briefing on July 24, 2008, the field collection of samples occurred from July 30–August 5, 2008. (Table D.1 summarizes key dates of the field investigation.) Samplers used a van-mounted Model 5400 Geoprobe, with a dual-tube sampling system, to collect soil samples from the six specified depth intervals (2–4 ft, 4–6 ft, 6–8 ft, 10–12 ft, 14–16 ft, and 18–20 ft) in the first four locations (30 ft from MW365 in north, south, east, and west directions), in accordance with procedures *Direct Push Technology Sampling* (PRS-ENR-0020) (currently PAD-ENR-0020); and *Decontamination of Sampling Equipment and Devices* (PRS-ENM-2702) (currently PAD-ENM-2702).

The dual tube sampling system recovered the cores from each sampling interval in 1.125-inch diameter plastic liners. In the field, the samplers trimmed the 4-ft long liners to the length of the recovered core (frequently greater than each 2-ft sample interval) and capped each end with plastic caps prior to field preservation with ice. Procedure PRS-ENM-2708 (currently PAD-ENM-2708), *Chain-of-Custody Forms, Field Sample Logs, Sample Labels, and Custody Seals,* governed the documentation of sample custody. The samples were placed in a secured refrigerator at the end of each day until the sample liners could be

BORING	SAMPLE INTERVAL (ft bgs)	SAMPLE ID NUMBER	DATE SAMPLED	DATE CORE DESCRIBED AND SAMPLE PREPARED	DATE OF FIELD TEST KIT ANALYSIS	DATE OF SAMPLE SUBMITTAL TO PGDP LABORATORY
W-30	2-4	UGAPCB08-01	07/30/08	07/31/08	07/31/08	07/31/08
	4-6	UGAPCB08-02	07/30/08	07/31/08	07/31/08	07/31/08
	6-8	UGAPCB08-03	07/30/08	07/31/08	07/31/08	07/31/08
	10-12	UGAPCB08-04	07/30/08	07/31/08	07/31/08	07/31/08
	14-16	UGAPCB08-05	07/30/08	07/31/08	07/31/08	07/31/08
	18-20	UGAPCB08-06	07/30/08	07/31/08	07/31/08	07/31/08
N-30	2-4	UGAPCB08-07	08/04/08	08/05/08	08/06/08	08/06/08
	4-6	UGAPCB08-08	08/04/08	08/05/08	08/06/08	08/06/08
	4-6	DUGAPCB08-01	08/04/08	08/05/08	NA*	08/06/08
	6-8	UGAPCB08-09	08/04/08	08/05/08	08/06/08	08/06/08
	10-12	UGAPCB08-10	08/04/08	08/05/08	08/06/08	08/06/08
	14-16	UGAPCB08-11	08/04/08	08/05/08	08/06/08	08/06/08
	18-20	UGAPCB08-12	08/04/08	08/05/08	08/06/08	08/06/08
S-30	2-4	UGAPCB08-13	08/04/08	08/05/08	08/06/08	08/06/08
	4-6	UGAPCB08-14	08/04/08	08/05/08	08/06/08	08/06/08
	6-8	UGAPCB08-15	08/04/08	08/05/08	08/06/08	08/06/08
	10-12	UGAPCB08-16	08/04/08	08/05/08	08/06/08	08/06/08
	14-16	UGAPCB08-17	08/04/08	08/05/08	08/06/08	08/06/08
	18-20	UGAPCB08-18	08/04/08	08/05/08	08/06/08	08/06/08
E-30	2-4	UGAPCB08-19	08/05/08	08/05/08	08/06/08	08/06/08
	2-4	DUGAPCB08-02	08/05/08	08/05/08	NA	08/06/08
	4-6	UGAPCB08-20	08/05/08	08/05/08	08/06/08	08/06/08
	6-8	UGAPCB08-21	08/05/08	08/05/08	08/06/08	08/06/08
	10-12	UGAPCB08-22	08/05/08	08/05/08	08/06/08	08/06/08
	14-16	UGAPCB08-23	08/05/08	08/05/08	08/06/08	08/06/08
	18-20	UGAPCB08-24	08/05/08	08/05/08	08/06/08	08/06/08

 Table D.1. Summary of Dates of Field Investigation and Sample Analyses

*NA = not analyzed, duplicate for fixed-base lab analysis only

opened, the sample cores described (see Tables D.2 through D.5), and samples prepared for analysis at an on-site sample preparation facility.

The investigation used a field laboratory trailer located at the C-755 sample preparation facility to perform the test kit analyses. Table D.6 documents the test kit analyses. For each soil sample of the initial four soil borings, a sample also was collected for analysis by the PGDP Analytical Services using Method 8082. Table D.7 summarizes the fixed-base laboratory analyses. Both field analysis and fixed-base laboratory analysis document that PCBs are not present in the soil samples at detectable levels.

Table D.2.	Sample	Lithologic	Descriptions	for West	Boring
------------	--------	------------	--------------	----------	--------

BORING	SAMPLE INTERVAL (ft bgs)	DATE SAMPLED	LITHOLOGIC DESCRIPTION
	2-4	07/30/08	SILT, 10YR4/6 (dark yellowish brown), soft, plastic, and moist, grading downward to CLAY, 10YR6/6 (brownish yellow), soft, plastic, and moist
	4-6	07/30/08	SILT, 10YR6/6 (brownish yellow), soft, plastic, and moist
	6-8	07/30/08	SILT as above
W-30	10-12	07/30/08	SILT, 7.5YR6/6 (reddish yellow), moderately soft, low plasticity, and moist
	14-16	07/30/08	SILT, 10YR7/1 (light gray) mottled with 10YR7/6 (yellow), moderately soft, low plasticity, and moist
	18-20	07/30/08	SILT as above but with lesser mottling

Table D.3. Sample Lithologic Descriptions for North Boring

BORING	SAMPLE INTERVAL (ft bgs)	DATE SAMPLED	LITHOLOGIC DESCRIPTION
	2-4	08/04/08	SILT, 10YR7/1 (light gray) mottled with 10YR7/6 (yellow), soft, plastic, and wet
	4-6	08/04/08	SILT, 10YR7/1 (light gray) mottled with 10YR6/6 (brownish yellow), moderately soft, medium plasticity, and moist. Trace 10YR2/1 (black) flecks (manganese?)
N-30	6-8	08/04/08	SILT, 10YR5/6 (yellowish brown) with little mottling by 10YR7/1 (light gray), moderately soft, low to medium plasticity, and moist
	10-12	08/04/08	SILT, 10YR5/6 (yellowish brown), moderately soft, low to medium plasticity, and moist
	14-16	08/04/08	SILT, 10YR7/1 (light gray) with some 10YR6/8 (brownish yellow) mottling, moderately soft, medium plasticity, and moist
	18-20	08/04/08	SILT as above

Table D.4. Sample Lithologic Descriptions for South Boring

BORING	SAMPLE INTERVAL (ft bgs)	DATE SAMPLED	LITHOLOGIC DESCRIPTION				
	2-4	08/04/08	SILT, 10YR6/3 (pale brown), soft, medium plasticity, and moist, grading downward to SILT, 10YR6/6 (brownish yellow) mottled with 10YR7/1 (light gray), moderately soft, medium plasticity, and moist				
	4-6	08/04/08	3/04/08 SILT, 10YR6/6 (brownish yellow), moderately soft, low plasticit and slightly moist (grading downward to moist)				
S 20	6-8	08/04/08	SILT, 10YR6/6 (brownish yellow), moderately soft, low plasticity, and moist				
5-30	10-12	08/04/08	SILT, 7.5YR5/6 (strong brown), moderately soft, low plasticity, and moist				
	14-16	08/04/08	SILT, 10YR7/1 (light gray) mottled with 10YR6/6 (brownish yellow), moderately soft to hard, low plasticity, and moist				
	18-20 08/04/08		18.0-19.6': SILT as above. 19.6-19.9': SILT with some sand, 7.5YR5/6 (strong brown), hard, low plasticity, and moist. Trace pebbles. 19.9-20.0': SILT, 10YR7/1 (light gray), soft, medium plasticity, and moist				

Table D.5. Sample Lithologic Descriptions for East Boring

BORING	SAMPLE INTERVAL (ft bgs)	DATE SAMPLED	LITHOLOGIC DESCRIPTION
	2-4	08/05/08	SILT, 10YR7/1 (light gray) mottled with 10YR6/6 (brownish yellow), moderately soft to hard, low plasticity, and moist
	4-6	08/05/08	SILT as above but moderately soft
E-30	6-8	08/05/08	SILT, 10YR6/6 (brownish yellow) with some mottling by 10YR7/1 (light gray), moderately hard, low plasticity, and moist. Trace pebble, rounded 10YR2/1 (black), approximately 4 mm diameter
	10-12	08/05/08	SILT, 10YR7/1 (light gray) mottled with 10YR6/6 (brownish yellow), moderately soft, low plasticity, and moist
	14-16	08/05/08	SILT as above
	18-20	08/05/08	SILT as above

				JLTS (AI	BSORBA	PCBs (ppm) (as PCB-1248)			
DATE	BORING/ CALIBRATOR	SAMPLE INTERVAL (ft bgs)	1 st READING	2 nd READING	3 rd READING	4 th READING	1 st READING	2 nd READING	3 rd READING
7/31/2008	DI Water*		0.000	0.008	0.009		NA	NA	NA
7/31/2008	1 ppm PCBs*		1.773	2.018	1.946		1.0	1.0	1.0
7/31/2008	5 ppm PCBs*		1.395	1.370	1.371		5.0	5.0	5.0
7/31/2008	W-30	2-4 ft	1.772	1.670	1.643		1.0	3.1	3.1
7/31/2008	W-30	4-6 ft	1.220	2.753	2.474		6.9	<1	<1
7/31/2008	W-30	6-8 ft	0.954	2.145	2.152		9.7	<1	<1
7/31/2008	W-30	10-12 ft	Cuvette	e broken-	could not	complet	e analysi	8	
7/31/2008	W-30	14-16 ft	4.717	4.717	4.717		<1	<1	<1
7/31/2008	W-30	18-20 ft	1.736	1.714	1.718		1.4	2.9	2.6
8/6/2008	DI Water*		0.000	0.001	0.002	0.004	NA	NA	NA
8/6/2008	1 ppm PCBs*		1.571	1.536	1.502		1.0	1.0	1.0
8/6/2008	5 ppm PCBs*		0.989	0.962	0.944		5.0	5.0	5.0
8/6/2008	N-30	2-4 ft	2.563	2.533	2.479		<1	<1	<1
8/6/2008	N-30	4-6 ft	2.019	1.981	1.941		<1	<1	<1
8/6/2008	N-30	6-8 ft	3.931	3.864	3.864		<1	<1	<1
8/6/2008	N-30	10-12 ft	1.964	1.914	1.879		<1	<1	<1
8/6/2008	N-30	14-16 ft	2.435	2.381	2.326		<1	<1	<1
8/6/2008	N-30	18-20 ft	2.877	2.812	2.751		<1	<1	<1
8/6/2008	S-30	2-4 ft	2.143	2.126	2.085		<1	<1	<1
8/6/2008	S-30	4-6 ft	2.489	2.457	2.413		<1	<1	<1
8/6/2008	S-30	6-8 ft	2.514	2.464	2.415		<1	<1	<1
8/6/2008	S-30	10-12 ft	1.695	1.655	1.618		<1	<1	<1
8/6/2008	S-30	14-16 ft	2.582	2.522	2.489		<1	<1	<1
8/6/2008	S-30	18-20 ft	1.778	1.749	1.706		<1	<1	<1
8/6/2008	E-30	2-4 ft	2.653	2.586	2.528		<1	<1	<1
8/6/2008	E-30	4-6 ft	2.306	2.246	2.192		<1	<1	<1
8/6/2008	E-30	6-8 ft	2.474	2.411	2.357		<1	<1	<1
8/6/2008	E-30	10-12 ft	2.371	2.315	2.266		<1	<1	<1
8/6/2008	E-30	14-16 ft	2.442	2.391	2.348		<1	<1	<1
8/6/2008	E-30	18-20 ft	1.698	1.660	1.627		<1	<1	<1

Table D.6. Results of the PCB Test Kit Analyses

7/31/08 1st Readings were affected by improper orientation of the cuvette within the colorimeter. These data are invalid. * = calibration standards NA = not analyzed

			RESULTS (µg/kg)								
BORING	SAMPLE INTERVAL (ft bgs)	SAMPLE ID NUMBER	Aroclor-1268	PCB, Total	PCB-1016	PCB-1221	PCB-1232	PCB-1242	PCB-1248	PCB-1254	PCB-1260
W-30	2-4	UGAPCB08-01	<80	<120	<90	<120	<90	<60	<90	<80	<90
	4-6	UGAPCB08-02	<80	<120	<90	<120	<90	<60	<90	<90	<90
	6-8	UGAPCB08-03	<80	<130	<100	<130	<100	<60	<100	<90	<100
	10-12	UGAPCB08-04	<80	<130	<100	<130	<100	<60	<100	<90	<100
	14-16	UGAPCB08-05	<70	<120	<90	<120	<90	<60	<90	<80	<90
	18-20	UGAPCB08-06	<70	<120	<90	<120	<90	<60	<90	<80	<90
N-30	2-4	UGAPCB08-07	<80	<120	<100	<120	<100	<60	<100	<90	<100
	4-6	UGAPCB08-08	<80	<130	<100	<130	<100	<60	<100	<90	<100
	4-6	DUGAPCB08-01	<80	<120	<90	<120	<90	<60	<90	<80	<90
	6-8	UGAPCB08-09	<80	<120	<90	<120	<90	<60	<90	<80	<90
	10-12	UGAPCB08-10	<80	<120	<90	<120	<90	<60	<90	<90	<90
	14-16	UGAPCB08-11	<80	<120	<90	<120	<90	<60	<90	<80	<90
	18-20	UGAPCB08-12	<80	<120	<90	<120	<90	<60	<90	<80	<90
S-30	2-4	UGAPCB08-13	<80	<130	<100	<130	<100	<60	<100	<90	<100
	4-6	UGAPCB08-14	<80	<130	<100	<130	<100	<60	<100	<90	<100
	6-8	UGAPCB08-15	<80	<130	<100	<130	<100	<60	<100	<90	<100
	10-12	UGAPCB08-16	<80	<120	<90	<120	<90	<60	<90	<90	<90
	14-16	UGAPCB08-17	<80	<120	<100	<120	<100	<60	<100	<90	<100
	18-20	UGAPCB08-18	<80	<130	<100	<130	<100	<60	<100	<90	<100
E-30	2-4	UGAPCB08-19	<80	<130	<100	<130	<100	<60	<100	<90	<100
	2-4	DUGAPCB08-02	<80	<120	<90	<120	<90	<60	<90	<80	<90
	4-6	UGAPCB08-20	<80	<130	<100	<130	<100	<60	<100	<90	<100
	6-8	UGAPCB08-21	<70	<120	<90	<120	<90	<60	<90	<80	<90
	10-12	UGAPCB08-22	<80	<120	<100	<120	<100	<60	<100	<90	<100
	14-16	UGAPCB08-23	<80	<130	<100	<130	<100	<60	<100	<90	<100
	18-20	UGAPCB08-24	<80	<130	<100	<130	<100	<60	<100	<90	<100

Table D.7. Fixed-base Laboratory (PGDP Analytical Services) Analyses

CONCLUSIONS

None of the soil samples collected from the vicinity of the well cluster containing MW363, MW364, and MW365 contained detectable levels of PCBs. Although the field test kits identified two soil samples from boring W-30 with PCBs below the 5 ppm level of detection (from 2 to 4 ft and 18 to 20 ft), fixed-base laboratory analyses of these same samples reported no detectable level of PCBs (with detection limits of approximately 0.1 ppm). All fixed-base laboratory analyses had reporting limits less than 1 ppm PCBs. The investigation had achieved its data quality objective.

A subsequent assessment of historical monitoring well maintenance activities identified that PCBs were introduced into C-746-U Landfill wells as a result of cross-contamination from an off-site facility (associated with previous well rehabilitation activities). This cross-contamination is sufficient to explain all the PCBs found in C-746-U Landfill well monitoring well groundwater samples. No further investigation of the shallow soils was required.

APPENDIX E

ASSESSMENT OF WELL REHABILITATION SOURCE OF PCB CONTAMINATION IN C-746-U LANDFILL WELLS

THIS PAGE INTENTIONALLY LEFT BLANK

ASSESSMENT OF WELL REHABILITATION SOURCE OF PCB CONTAMINATION IN C-746-U LANDFILL WELLS

The occurrence of polychlorinated biphenyls (PCBs) in monitoring wells at the C-746-S&T and C-746-U Landfills has been noted. As described in Appendix D, no PCB source associated with soils has been found. A soil source would be unlikely to be the source of groundwater contamination because PCBs have low solubility in water and high soil organic carbon partition coefficients (K_{oc}); thus, PCBs have a tendency for adsorption of soil and organic matter and would be unlikely to migrate at detectable levels to RGA groundwater units. PCBs are not mobile in soil and are rarely observed migrating in groundwater. When PCBs do migrate in groundwater, it's usually as colloids or attached to suspended particles. Therefore, they are not readily mobile for transport through the Upper Continental Recharge System (UCRS) and Regional Gravel Aquifer (RGA) as a dissolved phase.

The nature of PCBs and the empirical association of the timing of detections of PCBs with previous well rehabilitation events suggest that the source of PCBs is cross-contamination of the wells from materials used in well rehabilitation.

Five monitoring wells of the C-746-S&T and C-746-U Landfills—MW360, MW363, MW365, MW369, and MW372—produced water with detectable PCB levels in 2008.¹ Table E.1 summarizes the PCB detections in the landfill monitoring wells.

Well ID	# Samples	# Detects	Min (µg/L)	Max (µg/L)	Unit
MW220	10	1	0.78	0.78	URGA
MW221	10	1	7.87	7.87	URGA
MW222	12	5	0.19	3.1	URGA
MW223	12	3	0.25	1.61	URGA
MW224	10	2	0.24	0.359	URGA
MW263	2	1	0.092	0.092	RGA
MW357	15	1	0.43	0.43	URGA
MW360	15	4	0.18	0.23	URGA
MW361	15	2	0.3	0.443	LRGA
MW363	16	11	0.48	2.84	URGA
MW364	16	2	0.3	0.35	LRGA
MW365	14	8	1.17	3.69	UCRS
MW366	15	3	0.2	0.79	URGA
MW367	15	1	0.17	0.17	LRGA
MW368	2	1	1.47	1.47	UCRS
MW369	16	12	0.2	1.15	URGA
MW370	15	1	0.188	0.188	LRGA
MW372	15	3	0.2	0.38	URGA
MW373	15	1	0.19	0.19	LRGA

 Table E.1. Monitoring Wells with PCB Detections (January 2000 to June 2008)

Note: Bold indicates monitoring wells that produced water with detectable PCB levels in 2008.

None of the wells in the current monitoring network, however, yielded water with detectable levels of PCBs in 2002. Then in July and October of 2003, 13 wells had samples with detections of PCBs. The

¹ PCBs continue to be detected in samples from MW363, MW365, and MW369.

PCB-contaminated wells monitor the UCRS, upper RGA (URGA), and lower RGA (LRGA). The maximum PCB detection was 7.87 μ g/L (in MW221). The nearly simultaneous identification of PCBs in all three landfills and in monitoring wells that monitor multiple hydrogeologic units (UCRS, URGA, and LRGA) describes a broad occurrence that suggests a pathway other than migration via a groundwater plume. Table E.2 illustrates the PCB detections over time.

Well ID	CY 2002	CY 2003	CY 2004	CY 2005	CY 2006	CY 2007	CY 2008	Unit
MW220								URGA
MW221								URGA
MW222								URGA
MW223								URGA
MW224								URGA
MW357								URGA
MW360								URGA
MW361								LRGA
MW363								URGA
MW364								LRGA
MW365								UCRS
MW366								URGA
MW367								LRGA
MW368								UCRS
MW369								URGA
MW370								LRGA
MW372								URGA
MW373								LRGA
Shading sig	nifies a PCB de	tection						LICON

 Table E.2. Current Landfill Monitoring Wells with PCB Detections (by Calendar Year)

In April and May of 2003, PGDP rehabilitated 19 monitoring wells prior to the July 2003 sampling event (see Table E.3).

Well ID	Rehab Completed	Unit	PCB Detected?
MW365	4/17/2003	UCRS	Yes
MW363	4/22/2003	RGA	Yes
MW368	4/25/2003	UCRS	Yes
MW367	4/28/2003	RGA	Yes
MW364	4/29/2003	RGA	Yes
MW366	4/29/2003	RGA	Yes
MW357	5/1/2003	RGA	Yes
MW358	5/1/2003	RGA	Yes
MW360	5/2/2003	RGA	Yes
MW361	5/2/2003	RGA	Yes
MW369	5/7/2003	RGA	Yes
MW370	5/7/2003	RGA	Yes

Table E.3. Monitoring Well Rehabilitation,April and May 2003

	Rehab		РСВ
Well ID	Completed	Unit	Detected?
MW372	5/15/2003	RGA	Yes
MW373	5/15/2003	RGA	Yes
MW221	5/19/2003	RGA	Yes
MW222	5/22/2003	RGA	Yes
MW224	5/22/2003	RGA	Yes
MW223	5/23/2003	RGA	Yes
MW220	5/30/2003	RGA	Yes
MW242	5/30/2003	RGA	No

Table E.3. Monitoring Well Rehabilitation,April and May 2003 (Continued)

The potential for soils to the source of the occurrence of PCBs was that PCBs in soil around the landfills (and in the sediments of the former North-South Diversion Ditch) had been dragged down the borehole during well installation and then the rehabilitation process (heat and chemical treatment) mobilized the PCBs, allowing them to move into the monitoring wells. This explanation, however, does not account for PCBs found in the C-746-U Landfill monitoring wells. A similar explanation that would account for PCB detections in RGA wells would be if PCB-contaminated isolation casing was used in the construction of the wells, resulting in contamination of the well materials during construction. (Some of the isolation casing arrived with an oily coating and paint and had to be decontaminated.) This explanation, however, does not account for PCBs found in UCRS wells.

The explanation that best addresses all information is that PCBs had been introduced inadvertently into the landfill monitoring wells during rehabilitation. The well rehabilitation equipment had been used at previous sites and had not been decontaminated thoroughly. The PGDP wells became cross-contaminated. PCBs introduced by the rehabilitation became attached to sediment or colloid particles in the PGDP monitoring wells which then were captured during sampling events.

Figures E.1 through E.18 illustrate the relationship of the PCB occurrences to well rehabilitation events for the 18 wells with PCB detections. Where the first detection of PCBs is delayed after well rehabilitation, plots of turbidity support the association of PCBs with suspended solids in the wells.





Figure E.1. PCB Analyses and Well Rehabilitation in MW220



Figure E.2. PCB Analyses and Well Rehabilitation in MW221



Figure E.3. PCB Analyses and Well Rehabilitation in MW222



Figure E.4. PCB Analyses and Well Rehabilitation in MW223



Figure E.5. PCB Analyses and Well Rehabilitation in MW224


Figure E.6. PCB Analyses and Well Rehabilitation in MW357



Figure E.7. PCB Analyses and Well Rehabilitation in MW360



Figure E.8. PCB Analyses and Well Rehabilitation in MW361



Figure E.9. PCB Analyses and Well Rehabilitation in MW363



Figure E.10. PCB Analyses and Well Rehabilitation in MW364



Figure E.11. PCB Analyses and Well Rehabilitation in MW365



Figure E.12. PCB Analyses and Well Rehabilitation in MW366



Figure E.13. PCB Analyses and Well Rehabilitation in MW367



Figure E.14. PCB Analyses and Well Rehabilitation in MW368



Figure E.15. PCB Analyses and Well Rehabilitation in MW369



Figure E.16. PCB Analyses and Well Rehabilitation in MW370



Figure E.17. PCB Analyses and Well Rehabilitation in MW372



Figure E.18. PCB Analyses and Well Rehabilitation in MW373

THIS PAGE INTENTIONALLY LEFT BLANK

APPENDIX F

ASSESSMENT OF IMPACTS OF OVERPUMPING MONITORING WELLS IN THE VICINITY OF THE C-746-U LANDFILL ON POLYCHLORINATED BIPHENYLS CONCENTRATIONS

THIS PAGE INTENTIONALLY LEFT BLANK

INTRODUCTION

With the approval of the Kentucky Division of Waste Management, LATA Environmental Services of Kentucky, LLC, (LATA Kentucky) performed overpumping tests of four C-746-U monitoring wells (MW360, MW363, MW365, and MW369), beginning in March 2011, to assess impacts to the concentration of polychlorinated biphenyl (PCB)-contamination in the wells (see Appendix E for additional discussion of the suspected PCB source). MW360, MW363, and MW369 are upper Regional Gravel Aquifer (RGA) wells. MW365 is an Upper Continental Recharge System (UCRS) well. Each of these wells produced water with detectable PCB concentrations in 2008; thus, these wells were subjected to the overpumping test. NOTE: PCB levels in MW360 declined to below the laboratory detection limits beginning in 2009; however, this well still was subjected to overpumping.

The field effort for the overpumping assessment consisted of the following steps:

- 1. Collecting an initial groundwater sample for PCB analysis from each of the four wells by the low-flow sampling method, using the existing dedicated pumps (bladder pumps);
- 2. Removing the dedicated bladder pumps from the four wells followed by sustained pumping in the wells (at a rate of 1 to 2 gpm) using an inertial pump with a one-way foot valve; this pumping was accomplished using a reciprocal pumping action to more-aggressively agitate the well water;
- 3. Collecting groundwater samples for PCB analysis using the inertial pump one hour after sustained pumping began and again at the completion of overpumping (when monitoring of extracted water indicated turbidity had significantly declined);
- 4. Installing a new dedicated bladder pump (with new pump tubing) in each of the four wells; and
- 5. Collecting seven groundwater samples for PCB analysis from each of the four wells using the low flow method (and the new pumps), with a period of at least one day between the collection of samples in each well.

As discussed below, a review of the PCB analyses of samples collected from this effort in conjunction with the field monitoring parameters documents that PCB concentration levels are directly correlated to the turbidity of the groundwater samples. The implication of this relationship is that the PCB contamination is affixed to suspended solids within the well and well sand pack and not representative of the monitored groundwater unit.

QUALITY CONTROL

The field effort used personnel from LATA Kentucky's environmental sampling group, LATA Kentucky sampling and sample management procedures, and LATA Kentucky work control. The laboratory that performed the PCB analyses was certified by the U.S. Department of Energy Consolidated Audit Program to ensure the quality of the chemical analyses. LATA Kentucky performed data review and assessment of the completed analyses to further ensure quality control.

The groundwater analyses used a standard four-week laboratory analysis and reporting period except for the samples collected at the conclusion of overpumping (the last samples to be collected). LATA Kentucky required a two-week analysis and reporting period for these samples to provide for timely assessment of the effectiveness of early and final stages of the overpumping test.

The laboratory analysis reported the following PCB Aroclors (each with a lower reporting limit of $0.4 \,\mu g/L$):

- PCB-1016
- PCB-1221
- PCB-1232
- PCB-1242
- PCB-1248
- PCB-1254
- PCB-1254
- PCB-1260
- PCB-1268

PCB-1016 and PCB-1242 are the only Aroclors that commonly have been reported in analyses of groundwater samples from these wells.

In the overpumping test, only PCB-1242 was detected in samples from any of the wells. A discussion of the results of the individual well overpumping tests is presented below.

MW360 (UPPER RGA)

Samplers overpumped MW360 on March 7 and March 8, 2011, extracting approximately 1,175 gal of water (all wastewater generated by the overpumping test was treated for PCBs before final disposition). The overpumping continued until the water turbidity was reduced to 4.1 nephelometric turbidity units (NTUs). Post-overpumping sampling occurred April 5 through April 27, 2011.

A total of 20 previous analyses of groundwater samples from MW360 have been conducted. These include five analyses with detections of PCBs ($0.18-0.52 \mu g/L$) (samples collected during the 2006–2008 period). None of the eight analyses of 2009 and 2010 groundwater samples from MW360 reported detectable levels of PCBs.

Of the 10 samples (plus 2 duplicate samples) collected during the overpumping test, only 1 analysis (sample of April 11, 2011) reported a PCB detection, estimated at 0.272 μ g/L for PCB-1242. This concentration is below the U.S. Environmental Protection Agency's (EPA's) Drinking Water Standard maximum contaminant level (MCL) of 0.0005 mg/L and subsequent sampling (after April 11, 2011) did not detect PCBs. The finding that there are no longer detections of PCBs in groundwater at MW360 is consistent with the established PCBs trend, indicating that PCBs are no longer a problem in MW360. This finding indicates that there is no continuing source of PCBs to the water in this well.

MW363 (UPPER RGA)

Overpumping of MW363 occurred on March 9 and March 10, 2011, producing 1,600 gal of water with a final turbidity of 1.8 NTUs. Post-overpumping sampling was performed over the period April 6 through April 27, 2011.

The previous analyses of groundwater samples from MW363 (total of 24) reported 19 detections of PCBs (maximum of 2.84 μ g/L in 2004, declining to 0.25–0.3 μ g/L in 2009 and 2010). Three analyses of samples collected in 2010 did not detect PCBs (with detection limits of 0.17–0.18 μ g/L).

Analyses of the overpumping test documents the presence of PCB-1242 in 9 of 10 samples (and in the 2 duplicate samples), at estimated levels of 0.257–0.592 μ g/L. With the exception of 1 outlier, the PCB-1242 levels closely relate to turbidity of the samples (Figure F.1).



Figure F.1. PCB-1242 versus Turbidity in MW363

Except for the one sample with a concentration of 0.592 μ g/L (collected in early April 2011), none of the other samples had concentrations that exceed the EPA MCL of 0.5 μ g/L. Thus, the overpumping reduced concentrations of PCBs to less than the EPA MCL where they have remained since. The finding that there no longer are concentrations of PCBs in groundwater at MW363 that exceed the EPA MCL is consistent with the established PCBs trend, indicating that PCBs no longer are a problem in MW363. This finding indicates that there is no continuing source of PCBs to the water in this well.

MW365 (UCRS)

As is common with UCRS wells, MW365 was unable to sustain continued pumping. The samplers pumped the well five times (it was dry on a sixth attempt) during the period March 14 through April 7, 2011, producing a total of approximately 33 gal. Post-overpumping sampling occurred April 7 through May 3, 2011. Turbidity levels remained high in spite of development efforts under this task and were above 10 NTUs in three of the seven follow-on samples.

The 24 previous analyses of MW365 groundwater samples include 18 detections of PCBs (3.63--0.44 μ g/L). PCB-1242 was detected in all 10 of the overpumping test samples and in the two duplicate samples (ranging from 0.50-21.2 μ g/L). As in MW363, PCB-1242 levels closely relate to the turbidity of the groundwater in all but one sample (Figure F.2).



Figure F.2. PCB-1242 versus Turbidity in MW365

Sample concentrations continue to exceed the EPA MCL of 0.5 μ g/L in this UCRS well. Except as noted above, none of the RGA wells has PCB concentrations that exceed the EPA MCL. Concentrations of PCBs in MW365 have decreased in response to the overpumping but remain above the EPA MCL. MW365 is located approximately 1,000 ft north of the C-746-U Landfill. This area was subjected to the soils investigation summarized in Appendix D.

The body of investigation on the PCB issue is consistent with the inadvertent introduction of PCBS into this well by cross-contamination during well rehabilitation. Unfortunately, the well overpumping did not produce enough water to remove PCBs from this well to below EPA MCLs.

MW369 (UPPER RGA)

The samplers overpumped MW369 on March 2 and March 3, 2011, with follow-on sampling from April 6 through May 3, 2011. Overpumping produced a total of 950 gal of wastewater, reducing the turbidity of the water to 0.8 NTUs.

There are 27 analyses of MW369 groundwater for PCBs, prior to the overpumping test, with 22 detections of PCBs (0.19–1.15 μ g/L). Of the 10 groundwater samples and 2 duplicate samples collected for the overpumping test, PCBs were detected only 2 times at an estimated level of 0.112 μ g/L for PCB-1242 (the March 3 sample at the end of overpumping and the April 11 sample). The April 11 estimated PCB-1242 occurrence was in the groundwater sample with highest turbidity.

The post-overpumping results are below the EPA MCL of $0.5 \mu g/L$ and subsequent sampling (after April 11, 2011) did not detect PCBs. The finding that there are no longer detections of PCBs in groundwater at MW369 is consistent with the established PCBs trend, indicating that PCBs are no longer a problem in MW369. This finding indicates that there is no continuing source of PCBs to the water in this well.

ANALYSIS

Table F.1 summarizes the results of the overpumping test by comparing PCB analytical results with turbidity measurements. Tables F.2 through F.5 document the field measurements of groundwater at the time of sample collection. The field measurements include the following:

- Depth-to-water (from the top of well casing),
- pH,
- Temperature,
- Conductivity,
- Turbidity,
- Dissolved oxygen, and
- Oxidation/reduction potential.

With the exception of depth-to-water, the field parameters were measured in a flow-through cell to minimize the impact of atmospheric conditions on the sample quality.

			MW360				MW363	
Sample	PCB-124	42 (µg/L)	T 1.1.1.4		PCB-12	42 (µg/L)	T	
Date	Results ^a	Data Qualifier	(NTU)	Sample ID	Results ^a	Data Qualifier	(NTU)	Sample ID
3/2/2011								
3/3/2011								
	0.4	ND	11	MW360E1UG2-11				
2/7/2011	0.4	ND	4.1	MW360E2UG2-11				
5/7/2011	0.4	ND		MW360DE2UG2-11				
	0.4	ND		MW360E3UG2-11				
2/0/2011					0.378	J	7.2	MW363E1UG2-11
5/9/2011					0.291	J		MW363E2UG2-11
2/10/2011					0.291	J		MW363E3UG2-11
3/10/2011					0.276	J	1.8	MW363DE3UG2-11
3/14/2011								
4/5/2011	0.4	ND	58.9	MW360E4UG2-11				
4/5/2011	0.4	ND	58.9	MW360DE4UG2-11				
4/6/2011					0.592	J	21.7	MW363E4UG2-11
4/7/2011								
4/11/2011	0.272	J	3.8	MW360E5UG2-11	0.294	J	1.4	MW363E5UG2-11
4/11/2011					0.37	J	1.4	MW363DE5UG2-11
4/12/2011								
4/14/2011	0.4	ND	3.5	MW360E6UG2-11	0.4	ND	1	MW363E6UG2-11
4/19/2011	0.4	ND	2.6	MW360E7UG2-11	0.257	J	0.3	MW363E7UG2-11
4/18/2011								
4/21/2011	0.4	ND	2.2	MW360E8UG2-11	0.294	J	1	MW363E8UG2-11
4/25/2011	0.4	ND	7.5	MW360E9UG2-11	0.297	J	4.5	MW363E9UG2-11
4/27/2011	0.4	ND	8.5	MW360E10UG2-11	0.3	J	4.8	MW363E10UG2-11
5/3/2011								

Table F.1. PCB-1242 Analyses Results and Turbidity for Overpumping Test

^a lower detection limit or estimated quantity except when highlighted

ND = not detected

J = estimated

			MW365				MW369	
Sample	PCB-12	42 (µg/L)	Trackiditar		PCB-12	42 (µg/L)	Trackidia	
Date	Results ^a	Data Qualifier	(NTU)	Sample ID	Results ^a	Data Qualifier	(NTU)	Sample ID
					0.4	ND	7.8	MW369E1UG2-11
3/2/2011					0.4	ND		MW369E2UG2-11
					0.4	ND		MW369DE2UG2-11
3/3/2011					0.112	J	0.8	MW369E3UG2-11
3/7/2011								
5/7/2011								
3/9/2011								
3/ 7/ 2011								
3/10/2011								
5/10/2011								
	2.19		21.3	MW365E1UG2-11				
3/14/2011	1.6		21.3	MW365DE1UG2-11				
	4.39							
4/5/2011								
4/3/2011								
4/6/2011					0.4	ND	14.5	MW369E4UG2-11
4/7/2011	<mark>10.6</mark>		64.8	MW365E3UG2-11				
4/11/2011					0.112	J	14.8	MW369E5UG2-11
4/11/2011								
4/12/2011	21.2		56.6	MW365E4UG2-11				
4/14/2011	2.21		10.9	MW365E5UG2-11	0.4	ND	5.8	MW369E6UG2-11
4/18/2011	0.64		4.1	MW365E6UG2-11	0.4	ND	9.2	MW369E7UG2-11
4/16/2011	0.562		4.1	MW365DE6UG2-11	0.4	ND	9.2	MW369DE7UG2-11
4/21/2011	0.506		3.7	MW365E7UG2-11				
4/25/2011	0.626		7	MW365E8UG2-11	0.4	ND	9	MW369E8UG2-11
4/27/2011	0.535		9.6	MW365E9UG2-11	0.4	ND	10.3	MW369E9UG2-11
5/3/2011	0.502		7.9	MW365E10UG2-11	0.4	ND	7.5	MW369E10UG2-11

 Table F.1. PCB-1242 Analyses Results and Turbidity for Overpumping Test (Continued)

^a lower detection limit or estimated quantity except when highlighted

ND = not detected

J = estimated

				MW360				
Sample Date	Sample ID	Conductivity	Depth	Dissolved Oxygen	Oxidation/ Reduction Potential	рН	Temperature	Turbidity
		(umho/cm)	(ft)	(mg/L)	(mV)	(SU)	(°F)	(NTU)
3/2/2011								
3/3/2011								
	MW360E1UG2-11	395	41.55	4.44	501	6.83	56.5	11
2/7/2011	MW360E2UG2-11		41.35					4.1
5/7/2011	MW360DE2UG2-11							
	MW360E3UG2-11							
2/0/2011								
5/9/2011								
2/10/2011								
5/10/2011								
3/14/2011								
4/5/2011	MW360E4UG2-11	445	38.1	2.11	493	6.23	55.1	58.9
4/3/2011	MW360DE4UG2-11	445	38.1	2.11	493	6.23	55.1	58.9
4/6/2011								
4/7/2011								
4/11/2011	MW360E5UG2-11	452	37.79	0.86	476	6.19	59.7	3.8
4/11/2011								
4/12/2011								
4/14/2011	MW360E6UG2-11	450	37.95	1.11	648	6.15	58.8	3.5
4/19/2011	MW360E7UG2-11	458	37.41	1.26	291	6.1	61.3	2.6
4/10/2011								
4/21/2011	MW360E8UG2-11	448	37.53	1.56	482	6.1	59	2.2
4/25/2011	MW360E9UG2-11	459	36.21	1.55	399	6.18	61.2	7.5
4/27/2011	MW360E10UG2-11	463	35.48	1.14	262	6.18	60.6	8.5
5/3/2011								

Table F.2.	Field	Measurement	s in	the	MW360	Overpumping	g Test

				MW363				
Sample Date	Sample ID	Conductivity	Depth	Dissolved Oxygen	Oxidation/ Reduction Potential	рН	Temperature	Turbidity
		(umho/cm)	(ft)	(mg/L)	(mV)	(SU)	(°F)	(NTU)
3/2/2011								
3/3/2011								
3/7/2011								
3/7/2011								
2/0/2011	MW363E1UG2-11	357	47.48	3.74	748	6.75	57	7.2
5/9/2011	MW363E2UG2-11							
2/10/2011	MW363E3UG2-11							
3/10/2011	MW363DE3UG2-11							1.8
3/14/2011								
4/5/2011								
4/3/2011								
4/6/2011	MW363E4UG2-11	411	44.53	2.43	116	6.31	60.8	21.7
4/7/2011								
4/11/2011	MW363E5UG2-11	397	44.38	1	449	6.22	60.1	1.4
4/11/2011	MW363DE5UG2-11	397	44.38	1	449	6.22	60.1	1.4
4/12/2011								
4/14/2011	MW363E6UG2-11	392	44.6	0.86	544	6.2	60	1
4/19/2011	MW363E7UG2-11	403	43.97	1.18	326	6.49	60.4	0.3
4/18/2011								
4/21/2011	MW363E8UG2-11	396	44.13	1.19	521	6.15	58	1
4/25/2011	MW363E9UG2-11	397	42.8	1.32	347	6.26	60.9	4.5
4/27/2011	MW363E10UG2-11	394	42	1.03	154	6.25	59.2	4.8
5/3/2011								

Table F.3.	Field	Measuremen	ts in the N	AW363 (Overpumping	Test

				MW365				
Sample Date	Sample ID	Conductivity	Depth	Dissolved Oxygen	Oxidation/ Reduction Potential	pH	Temperature	Turbidity
		(umho/cm)	(ft)	(mg/L)	(mV)	(SU)	(°F)	(NTU)
3/2/2011								
3/3/2011								
3/7/2011								
5/7/2011								
2/0/2011								
3/9/2011								
2/10/2011								
3/10/2011								
	MW365E1UG2-11	397	38.45	6.7	790	6.52	55.1	21.3
3/14/2011	MW365DE1UG2-11	397	38.45	6.7	790	6.52	55.1	21.3
	MW365E2UG2-11							
4/5/2011								
4/3/2011								
4/6/2011								
4/7/2011	MW365E3UG2-11	398	43.74	7.36	293	6.37	68.6	64.8
4/11/2011								
4/11/2011								
4/12/2011	MW365E4UG2-11	414	42.58	9.73	583	6.84	58.2	56.6
4/14/2011	MW365E5UG2-11	418	42.08	7.6	582	6.22	60.4	10.9
4/10/2011	MW365E6UG2-11	438	38.98	11.37	295	6.5	61.5	4.1
4/18/2011	MW365DE6UG2-11	438	38.98	11.37	295	6.5	61.5	4.1
4/21/2011	MW365E7UG2-11	423	39.36	7.92	525	6.1	59.8	3.7
4/25/2011	MW365E8UG2-11	424	35.7	6.96	547	6.17	62	7
4/27/2011	MW365E9UG2-11	422	33.48	5.83	366	6.17	60.6	9.6
5/3/2011	MW365E10UG2-11	428	30.22	4.98	460	6.15	59	7.9

Table F.4. Field Measurements in the MW365 Overpumping 7	'est
--	------

		MW369										
Sample Date	Sample ID	Conductivity	Depth	Dissolved Oxygen	Oxidation/ Reduction Potential	рН	Temperature	Turbidity				
		(umho/cm)	(ft)	(mg/L)	(mV)	(SU)	(°F)	(NTU)				
	MW369E1UG2-11	359	43.2	1.9	673	6.35	59	7.8				
3/2/2011	MW369E2UG2-11											
	MW369DE2UG2-11											
3/3/2011	MW369E3UG2-11		43					0.8				
2/7/2011												
5/7/2011												
2/0/2011												
5/9/2011												
2/10/2011												
3/10/2011												
3/14/2011												
4/5/2011												
4/3/2011												
4/6/2011	MW369E4UG2-11	238	40.35	2.89	199	6.76	62.4	14.5				
4/7/2011												
4/11/2011	MW369E5UG2-11	214	39.89	8.09	531	7.52	61.3	14.8				
4/11/2011												
4/12/2011												
4/14/2011	MW369E6UG2-11	254	40.13	8.18	533	7.32	62.8	5.8				
4/10/2011	MW369E7UG2-11	244	39.72	5.16	250	6.96	62.1	9.2				
4/18/2011	MW369DE7UG2-11	244	39.72	5.16	250	6.96	62.1	9.2				
4/21/2011												
4/25/2011	MW369E8UG2-11	335	38.94	9.11	519	7.38	62.4	9				
4/27/2011	MW369E9UG2-11	347	38.34	8.24	310	7.37	61.4	10.3				
5/3/2011	MW369E10UG2-11	348	37.97	8.7	402	7.48	60.5	7.5				

Table F.5. Field Measurements in the MW369 Overpumping Test

Figures F.1 and F.2 illustrate the relationship between PCB-1242 levels and turbidity. Because turbidity is a measure of suspended solids, PCB concentrations could be reduced further by surging and aggressive overpumping. However, PCB concentrations in RGA wells are below the EPA MCL, such that additional well development efforts are not needed to confirm that there is no above-MCL PCB contamination present in these wells and thus no C-746-U Landfill source of above-MCL contamination.

The lone UCRS well (MW365) is not purged effectively. Even with additional purging of this well, concentrations may remain above the EPA MCL. given its location, approximately 1,000 ft north of the C-746-U Landfill, and with a negative finding of soil contamination in the vicinity (see Appendix D), no additional steps are needed to confirm that the C-746-U Landfill is not a source of this PCB contamination.

APPENDIX G

LITHOLOGIC LOGS/WELL LOGS

THIS PAGE INTENTIONALLY LEFT BLANK

The logs of Appendix G document the lateral continuity of the primary hydrogeologic units beneath the C-746-U Landfill. The map showing well locations is from *C-746-U Solid Waste Landfill Groundwater Monitoring Plan, Paducah Gaseous Diffusion Plant, Paducah, Kentucky,* BJC/PAD-205/R1, December 2001. The lateral continuity of tops of the individual well clusters is summarized in Table G.1.

Monitoring Well Cluster	Average Surface Elevation (ft amsl)	UCD/UCRS (Hydrogeologic Units 1–3)	LCD/RGA (Hydrogeologic Units 4–5)	McNairy (Hydrogeologic Unit 6)	Dominant Lithology	Approximate Depth (ft below grade)	Thickness	Approximate Elevation of Top of Unit
	367	HU1			CL	0-28	28 ft	367
357		HU2			CL, Sdy CL	28-30	12 ft	339
358		HU3			Sd, Sdy CL	30-45	15 ft	337
359			HU4		Sd	45-55	10 ft	322
			HU5		GR, Sdy GR	55-84	29 ft	312
				HU6	CL	84		283
	359	HU1			SiCl, CL	0-19	19 ft	359
360		HU2			Sdy CL, SiCl, ClSi	19-37	18 ft	340
361		HU3			Sd, SiSd	37-45	8 ft	322
362			HU4		Sdy GR, Sd	45-60	15 ft	314
			HU5		GR	60-70	10 ft	299
				HU6	CL, ClSi	70 ft		289
	366	HU1			Silt, SiCl, ClSi	0-19	19 ft	366
363		HU2			Cl, SiCl	19-No Sample	?	347
364		HU3			SiCl	No Sample	?	?
365			HU4		Sd	52-56.5	4.5 ft	314
			HU5		Sdy GR	56.5-86	29.5 ft	309.5
				HU6	CL	86		280

Table G.1. Table of Averaged Stratigraphic Tops (Elevations) and Thicknesses for the C-746-U Landfill Well Clusters

Monitoring Well Cluster	Average Surface Elevation (ft amsl)	UCD/UCRS (Hydrogeologic Units 1–3)	LCD/RGA (Hydrogeologic Units 4–5)	McNairy (Hydrogeologic Unit 6)	Dominant Lithology	Approximate Depth (ft below grade)	Thickness	Approximate Elevation of Top of Unit
	367	HU1			SiCl, ClSi	0-20	20 ft	367
366		HU2			ClSi	20-35	15 ft	347
367		HU3			SiCl, Sdy CL	35-50	15 ft	332
368			HU4		Sd	50-55	5 ft	317
			HU5		GR, Sdy GR	55-80	25 ft	312
				HU6	CL	80		287
	362	HU1			ClSi, Silt	0-19 ft	19 ft	362
369		HU2			Silty Sd	19-34 ft	15 ft	343
370		HU3			Sdy Silt	34-43	9 ft	328
371			HU4		Sd	43-54 ft	11 ft	319
			HU5		GR & Sd	54-76	22 ft	308
				HU6	CL	76 ft		286
	358	HU1			CL, SiCl	0-14 ft	14 ft	358
372		HU2			Silt	?	?	?
373		HU3			SiCl	?	?	?
374			HU4		Sd, Sd & GR	42-50	8 ft	316
			HU5		GR	50-71	21 ft	308
				HU6	CL	71		287

Table G.1. Table of Averaged Stratigraphic Tops (Elevations) and Thicknesses for the C-746-U Landfill Well Clusters (Continued)

Cl = clay ClSi = clayey silt GR = gravel

Sdy = sandy SiCl = silty clay SiSd = silty sand

SD = sand? = no or poor sample

HU = hydrogeologic unit

LCD = Lower Continental Deposits

RGA = Regional Gravel Aquifer

UCD = Upper Continental Deposits UCRS = Upper Continental Recharge System



G-5

		C-7	46 S,T, AND U LANDF Monitoring Wells	ILL		
Mon.Well	Plant Coord.'s @ Casing	Plant Coord.'s @ Brass Cap	KY. State Plane Coord.'s @ Casing	KY. State Plane Coord.'s @ Brass Cap	Elevation @ Top of Outer Casing	Elevation @ Brass Cap
357	N. 6451.80 W. 2829.58	N. 6450.32 W. 2829.36	N. 1946219.62 E. 752317.96	N. 1946218.17 E. 752317.61	368.98	366.86
358	N. 6444.38 W. 2851.93	N. 6442.95 W. 2851.77	N. 1946221.07 E. 752294.45	N. 1946219.69 E. 752294.07	369.13	366.62
359	N. 6448.02 W. 2840.71	N. 6446.67 W. 2840.63	N. 1946220.27 E. 752306.22	N. 1946218.99 E. 752305.79	369.21	366.65
360	N. 6467.64 W. 2627.14	N. 6466.58 W. 2626.04	N. 1946158.88 E. 752511.71	N. 1946157.49 E. 752512.34	362.27	360.03
361	N. 6487.36 W. 2617.48	N. 6486.83 W. 2616.48	N. 1946173.57 E. 752528.03	N. 1946172.71 E. 752528.76	361.55	359.46
362	N. 6477.31 W. 2621.41	N. 6476.47 W. 2620.31	N. 1946165.71 E. 752520.64	N. 1946164.52 E. 752521.35	362.11	359.63
363	N. 6521.42 W. 2392.05	N. 6522.63 W. 2392.78	N. 1946121.16 E. 752749.91	N. 1946122.56 E. 752749.69	368.78	366.25
364	N. 6535.89 W. 2373.54	N. 6536.81 W. 2374.75	N. 1946127.69 E. 752772.48	N. 1946128.99 E. 752771.70	368.39	365.95
365	N. 6528.32 W. 2383.31	N. 6529.57 W. 2384.10	N. 1946124.31 E. 752760.60	N. 1946125.76 E. 752760.33	368.23	366.00
366	N. 6121.18 W. 2246.10	N. 6121.53 W. 2244.89	N. 1945695.37 E. 752736.17	N. 1945695.23 E. 752737.42	369.16	366.87
367	N. 6145.28 W. 2247.09	N. 6145.57 W. 2245.97	N. 1945718.09 E. 752744.24	N. 1945717.94 E. 752745.38	369.48	367.37
368	N. 6134.00 W. 2247.27	N. 6134.21 W. 2246.03	N. 1945707.70 E. 752739.86	N. 1945707.43 E. 752741.09	369.23	367.07
369	N. 4564.73 W. 2957.51	N. 4564.54 W. 2958.42	N. 1944516.22 E. 751495.91	N. 1944516.37 E. 751495.00	364.32	362.02
370	N. 4589.20 W. 2957.40	N. 4589.07 W. 2958.26	N. 1944538.88 E. 751505.13	N. 1944539.08 E. 751504.28	365.20	362.95
371	N. 4576.61 W. 2957.43	N. 4576.41 W. 2958.30	N. 1944527.20 E. 751500.41	N. 1944527.35 E. 751499.53	364.75	362.56
372	N. 4817.24 W. 2486.89	N. 4816.20 W. 2487.00	N. 1944575.12 E. 752026.73	N. 1944574.19 E. 752026.25	359.60	357.33
373	N. 4823.14 W. 2509.92	N. 4822.16 W. 2509.96	N. 1944589.18 E. 752007.56	N. 1944588.28 E. 752007.16	359.88	357.72

T. 7.,

۰,

(

G-6

Mon.Well	Plant Coord.'s @ Casing	Plant Coord.'s @ Brass Cap	KY. State Plane Coord.'s @ Casing	KY. State Plane Coord.'s @ Brass Cap	Elevation @ Top of Outer Casing	Elevation @ Brass Cap
374	N. 4819.04 W. 2497.62	N. 4818.02 W. 2497.78	N. 1944580.80 E. 752017.44	N. 1944579.90 E. 752016.92	359.58	357.53
375	N. 5886.80 W. 2907.85	N. 5887.13 W. 2906.55	N. 1945724.51 E. 752034.75	N. 1945724.33 E. 752036.08	370.47	368.73
376	N. 5816.34 W. 2403.20	N. 5815.10 W. 2402.97	N. 1945471.04 E. 752476.77	N. 1945469.80 E. 752476.53	370.55	368.22
377	N. 5318.27 W. 2280.94	N. 5319.38 W. 2280.09	N. 1944963.29 E. 752404.58	N. 1944964.00 E. 752405.78	365.81	363.58
384	N. 3828.36 W. 3121.20	N. 3828.26 W. 3119.62	N. 1943893.91 E. 751069.56	N. 1943893.23 E. 751070.98	365.36	363.07
385	N. 3804.81 W. 3119.46	N. 3804.72 W. 3117.96	N. 1943871.41 E. 751062.40	N. 1943870.77 E. 751063.76	365.77	363.18
386	N. 3816.45 W. 3120.54	N. 3816.28 W. 3119.05	N. 1943882.61 E. 751065.73	N. 1943881.90 E. 751067.05	365.46	363.11
387	N. 4188.73 W. 3073.18	N. 4187.91 W. 3074.44	N. 1944210.41 E. 751248.43	N. 1944210.12 E. 751246.95	363.60	361.46
388	N. 4197.35 W. 3080.77	N. 4196.58 W. 3082.14	N. 1944221.24 E. 751244.60	N. 1944221.04 E. 751243.04	363.57	360.98
389	N. 4258.20 W. 2913.08	N. 4256.78 W. 2912.00	N. 1944215.21 E. 751422.89	N. 1944213.48 E. 751423.36	364.26	361.92
390	N. 4394.91 W. 2713.38	N. 4396.00 W. 2712.24	N. 1944267.64 E. 751659.15	N. 1944268.22 E. 751660.62	360.57	358.23
391	N. 4557.92 W. 1993.30	N. 4558.12 W. 1994.68	N. 1944150.52 E. 752388.11	N. 1944151.21 E. 752386.90	366.80	364.39
392	N. 4582.37 W. 1994.30	N. 4582.74 W. 1995.72	N. 1944173.58 E. 752396.29	N. 1944174.45 E. 752395.11	366.00	363.73
393	N. 4571.03 W. 1993.08	N. 4571.17 W. 1994.65	N. 1944162.60 E. 752393.20	N. 1944163.31 E. 752391.79	366.94	364.25
394	N. 3460.44 W. 1895.64	N. 3460.79 W. 1896.92	N. 1943095.72 E. 752069.68	N. 1943096.52 E. 752068.63	378.55	376.57
395	N. 3484.23 W. 1894.71	N. 3484.66 W. 1895.93	N. 1943117.44 E. 752079.41	N. 1943118.30 E. 752078.44	379.22	377.50
396	N. 3471.98 W. 1894.83	N. 3472.47 W. 1896.14	N. 1943106.13 E. 752074.74	N. 1943107.06 E. 752073.70	378.93	376.77
397	N. 3138.13 W. 2509.29	N. 3136.78 W 2508 90	N. 1943025.35 E 751380.12	N. 1943023.94 E. 751379.98	387.17	385.00

•

(

(

G-7

•

Mon.Well	Plant Coord.'s @ Casing	KY. State Plane Coord.'s @ Casing	Elevation @ Top of Inner Casing
380	N. 7205.26 W. 5190.31	N. 1947798.67 E. 750408.15	368.64
381	N. 7745.84 W. 4892.90	N. 1948189.46 E. 750885.61	369.30

This is to certify to Science Applications International Corporation, that the information shown hereon was obtained under my personal supervision. The coordinates shown were calculated from a primary traverse using Accu-Air monuments A-2 and A-20 as the reference baseline. The mathematical error of closure obtained was calculated to be greater than 1: 97,989. Elevations for monitoring wells were calculated using the method of differential leveling and based upon an elevation at Accu-Air monument A-20 of 373.60 feet above Mean Sea Level. Kentucky State Plane Coordinates were calculated from data received from Bechtel Jacobs Inc. for monument locations. This information meets or exceeds the minimum standards of practice for land surveying in Kentucky.

tornal, the trates as a

1.111

Joel T. Prine, PLS No. 3367

<u>6/20/07</u> June 20, 2002





- Andrews	<i>`</i>					C-17			Pagecf_	3
	LITHOI	LOGIC	LOC	BOR	ING/WELL	NO.:	MW 35	8		
	Facility:	PGDP						Sit	e: 0-746 U	
	Client:	Eı	nvironr	nental	Restoration	n Divisio	on BJC,)			
	Contractor:	(SAI	С	Drill C	Contractor: M	165		Driller: 5	- File for ey	
	Start Time:	14 39	2/8	102	End Time:	715	2/8/02	Borchole	Dia(s): 8.5	
	Drill Method	/Rig Type:	87 Hote	w stan	Augors, CME	5' 5 100 1	CIME75>	Total Dep	pth: 39	
	Logged By:	WayDH	hu					Protectio	n Level: D	
ł	Depth (ft) Interv	SAMPLE al Number	Recovery (ft)	MEA ALPEAIBet	FIELD SUREMENTS a/Gamma VOC'	LITHOL	OGIC DESCRIPTION	GRAPH LOG	COMMENTS	
	1 2 3 4 5 6 7 8 C	pun th 1 2014 2	36,40			CLAY dark y ioyn mt m brown Sibang from Sibang from CLAY Little Very K orange Small orange Mois	Trace sund pellowish oran 6/6 oderite yelle 1042 5/4 gular, moist sund 240 orange 107 ark yellowish e 1042 6/6 zones off ics, 5 doing t firm			
A1182A/26JUL **	7 10 11 12 3	Run 3	4.9			CLAY 10412 401100 10412 50104 mo15	trice sand Rule orange g/z, Mt durk uish orange 6/b ngular, firm			
	DATA REVOR	ÆBIÆ:	• •••	··· / m	ØTE:	DATA CHE	CKED & REVIEWEI) EY:	DATE:	

G-11

· •j							0	C-17		Page 2 of 3
	LITH	HOLO	OGIC	LOO	G BC	DRING/WE	LL I	NO .: MW 358		
	Facility	r: PG,	DP						Site:	
	Client:		Er	nvironi	nenta	al Restore	tion	Division BTC .)		
	Contra	ctor: (SAIC	2	Dril	I Contractor			Driller: 6.6	autney
	Start 1	lime:		·		End Tir	ne:		Borehole Dia	a(s): 8.5
+	Drill M	lethod/R	tig Type:	8 4 Hollo	w 5+	PAN AUGERS,	LME	5' Spoon, CME 75 >	Total Depth:	39
-	Logged	Ву: //	Compt -	fre					Protection 1	evel: D
	Depth (ft)	Interval	SAMPLE	Recovery		FIELD MEASUREMENT Beta/Gamma	S VOC's	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS
	14 15 16 17 18		punt y	5.0 5.0		(com)		CLAY, 1. He silt trace sand, very pale orange loyiz 8/2 mt david ypilowish orange to yiz 6/6 subongular, firm moist. Loss clay at bottom of sipoon		
	70 21 22 23		Run 5	5.0				CLAY, Some Silt trace Sand, durk yollowishovang loge 6/6 mt Ligh gray N7, Subangu firm, moist		
A1182A/26JUL"	29 25 26 27		Runt 6					24.0 - 27.8 Same 45 4boul 27.8 - 28.0 SAND, little clay dark yellowish orang, iby12 6/6 Medium Graint B Subagular to Subrow Moist	e	
	DATA	RECORDE	81 59 :	• • •		" DATE:		DATA CHECKED & REVIEWED) EY:	DATE:

~	1	7
5	1	1

*										
LITI	HOLO	DGIC	LO	G B(DRING/WE	LL I	NO .: MW 358			
Facilit	y: P	GDP							Site:	
Client	:	E:	nviron	ment	al Restora	tion	Division BTC)			
Contra	ctor: (SAI	C	Dri	Il Contractor	:: M	65	Drille	: 5.6	Gutney
Start '	Time:		,		End Tir	ne:		Borch	ole Di	a(s): 8.5
Drill 4	lethod/R	Rig Type:	84 461	low 5	From Augars	CIME	=5' 5,000 , LME75 >	Total	Depth	: 39
Logged	By:	Wast	H.M				, 	Prote	ction .	Level: D
SAMPLE FIELD										
(ft)	Interval	Number	Recovery (ft)	ALPHA (cpm)	Beta/Gamma (cpm)	VOC's	LITHOLOGIC DESCRIPTION	10	G	COMMENTS
78 -		Cont.	~							
29 30 31 32 33 34 35 36 37		2007 7 7 200 8	5.0				24.0-24.8 SAND, 1.++1e gravel Frace Chy dark ypllowish oringt 1042646, Coarse Sudangular, Sylwryth Poorly Sorted 24.8-34.0 CLAY 1.++1e Silt Trice Sand dark ymllowish orning 1042616 Mt Light Gray N7 Sudangular SHIFF Moist 34.0-36.0 Same as about 36.8-36.2 CLAY Some S.14 trace Same inderate reddish brown 102416, Subang firm moist SFLTY CLAY			Ibindan t MNO Stain S
38							Trice sand, Mockerat Yallowish Brown 10412 514, firm Moist		7	rice MWO Stains
DATA	RECORDE	BLÆ¥:	·	••••	DATE:		DATA CHECKED & REVIEWED	BY:		DATE:

			C-17		Page L cf 4			
LITHOLC	GIC LOG	BORING/WELL	NO .: MW 358	3				
Facility: P	GDP			Site: (146ULAND			
Client:	Environme	ntal Restoration	Division ' ,)					
Contractor: E	SAIC I	orill Contractor:	465	Driller: H. Di	PCHDRCH			
Start Time:	337 3/5/0	Z. End Time:	0950 3/6/02	Borchole Dia(s):	8/11			
Drill Method/Ri	ig Type: 81/4 14	SA, CHE 5	SPOON, CHiz 85	Total Depth: S	15			
Logged By:	V. MULLINS	2		Protection Level	: D			
Depth	SAMPLE	FIELD		GRAPH				
(ft) Interval	Number (ft) (cr	AlBeta/Gamma VOC's	LITHOLOGIC DESCRIPTION	100	COMMENTS			
41 42 42 43 43 44 45 44 45 44 45 46 47 50 48	2.5 5.0 33"" 5.0		<u>CLAY</u> , SANDY (40) SUBBNGNLAZ TO SUBBNNDID; ST. MOIST, FINIL SAN TRACE MICH YELLOWISH REID (5YR, 4/6). SOND FINIL FRI (5YR, 4/6). MUD REATELY SUR MUD REATELY SUR MUD REATELY SUR MUCH, YELLOWISH RICH, YELLOWISH RICH, YELLOWISH RICH, YELLOWISH					
49-50-51-50-55	1:0		SAND FINE GRUNNID, WELL SORTED; SUBROWN STRONG BROWN (7.5 425/0) (CROER 4" SAND MEDIUM GRAND SUBROWN GRAND SUBROWN	200				
DATA REVORDET	<u> N</u>	judáte:	DATA CHECKED & REVIEWED	<u>аў — 1 — т</u> ву:	DATE:			
z	2			(C-17		Page 2 cl 4	-
-------------	---	------------------------------------	------------------------------	--	--	-----------------------	-----------------	------
	LITHOLC	GIC LO	G BORI	NG/WELL I	NO .: MW 358	3]
	Facility: P	GDP				Si	te: (-7460 Lawo	FILL
	Client:	Environ	nental I	Restoration	Division ' ,)			1
	Contractor: E	SAIC	Drill Ca	ontractor:	465	Driller: H. DPCHDRCH		
	Start Time:	337 3/5	5/02	End Time: 2	1950 3/6/02	Borehole	Dia(s): 8/1	
	Drill Method/R	ig Type: 8/4	17SA	, CME 5'	SPACE, CHE 85	Total De	pth: 85	
	Logged By:	V. MULLI	uS			Protectio	n Levei: D	^
	Depth (ft) Interval	SAMPLE Recovery Numberi (ft)	MEAS ALPHAI Beta (tpm)	FIELD UREMENTS /Gamma VOC's (symi (ypm)	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS	×
	54 -					$\left \right\rangle$		
	55 56 57 57 60 60 61	3.5			ARAVEL, UNEY POULLY SOLTED, CHA VARY COALSE PRESSIE GALASE PRESSIE GRAUEL TRACE COBISLE GRA STRONG BZONN (R.S VR 4/6) LOWER 10" SAND FINGTO MEDIUM GEARDED SUBDUNDED, WILL SOLTED, REDDISH BZOWN (7.5426/2 SAND AS DEOVE SAND AS DEOVE			
A1982/A2810	62 60 63 64 65 65 67 67 67	22 " 22 " 5.0			VERY CONTRE SAUD MEDUM TO VIRY CONTRE SAUP, SUBZOUNDED, VARY PEDBLE TO VERY CON GRAVEL (3590), FAR PEDBLE TO VERY CON STEDNG BECON (2.5 VR 5/6) CHERT, CON TRACE MICH GRAVELLY (YO'Z), FINE PEDBLE TO VERY CONTRE PEDBLE TO VERY		•	
6	DAIA RECORDED	-BI:	A STA	11	DATA CHECKED & REVIEWED	d1:	DATE:	[

C-17

(

Page 3 cl 4

LITHOLOGIC L	OC BORING/WELL	NO.: MI.1358							
Facility: PGDP		1100000	Site: (74601000					
Client: Envire	onmental Restoration	Division ')							
Contractor: BAIC	Drill Contractor:	465	Driller: H. L	PLHDRCH					
Start Time: 1337 3/5/02 End Time: 0950 3/6/02 Borehole Dia(s):									
Drill Method/Rig Type: 8	SPOON, CHE 85	Total Depth: 85							
Logged By: V. MULL	Protection Lev	el: D							
Depth SAMPLE (ft) Interval Number (ft)	FIELD MEASUREMENTS ery ALPHA Beta (Gamma VOC's (cpm) (ppm)	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS					
69		· ·							
70 30'		GIRAVIEL SANDY (4070) MEDINMITA VENY COARSE SHOD, GIRAVIEL TO VIE24 COARSE INERSIE GIRAVIEL, TRACE	>						
73 - 77		COBBLE GROUPS SVB2002020, VERY POULLY SUITED, WEF, STEAMS BROWN (7.5425/6)							
75 76 77 75 77 75 75 78		NO ZECOVERY							
79 50 81 - 85 5.1	υ	SAND; MEDIUMTO CUARSE, SUBZOUNDO WILL SUZZED, CHERT, GUARTZ: WIT; TRACE							
DATA REVORDEBLET:	TADATE:	DATA CHECKED & REVIEWED	BY:	DATE:					

			Page 4ct 4
LITHOLOGIC	LOG BORING/WELL N	10.: MW35F	3
Facility: PGDP	>		Site: (2746ULDADA
Client: Env	vironmental Restoration	Division ')	
Contractor: GAIC	Drill Contractor:	165	Driller: H. DPCHDRCH
Start Time: 1337	3/5/02 End Time: 0	950 3/6/02	Borehole Dia(s): 81/1
Drill Method/Rig Type:	81/4 HSA, CHES'	SPOCE, CHE 85	Total Depth: 85
Logged By: V.HU	ILLINS		Protection Level: D
Depth SAMPLE (ft) Interral Number	FIELD MEASUREMENTS acovery ALPHAI Bets/Gamma VOC's	LITHOLOGIC DESCRIPTION	CRAPH LOG COMMENTS
82		CUBBLE GIZAVIEL (2 EACH & REATER THAN 3") STRUNG BROWN 7.542.578 IN END CAP <u>CLAY</u> , LIGHT BROWN GRAN (10486/2) HUS UITH YELLONGISH BROW (104 E576) MEDIUM S TRACE FINE SAWD, TR MICH TD = 85.0 FT /3"	ST 7220 7742 725 725
DATA REVORDEBUTT	ADATE: I	DATA CHECKED & REVIEWED	EY: DATE:

	Well No MW 359 Installation: N/A	
	Project No. Client/Project: BJC/C-746-U Well Aband	onment/Replacement Site: PGDP/C-746-U Landfill
,	Contractor: SAIC	Drill Contractor: Miller Government Services
(Start Date: 2/21/2002 (1510 M)	End Date: 2/23/2002 (1010 M)
		Well Coord: N 6448.02
	3.0'	W 2840.71
	Elev.: (Well pump) 369.07	Protective Casing
	Elev.: (Brass cap) 366.65	Material Type: Steel
		Diameter: 10-in. Depth BGS: 1.5' Weep Hole? (Y / N): Yes
	Death ****	Guard Posts (Y / N): Yes Type: Steel
		Surface Pad
		Riser Pipe
		Type: SCH 40 PVC Diameter: 4-in.
		Total Depth (TOR to TOS): 31.9'
		Grout
		Composition & Proportions: High Solids Bentonite Grout Type: Pure Gold
		Weight (Ibs/gal.): 10.0
		Tremied (Y / N): Yes
		Centralizers (Y / N): Yes
		Depth(s): 40.0'
		Type: 3/8-inch TR 30 Bentonite Pellets
		Source: Pel-Plug Setup/Hydration Time: 20 hr Vol. Fluid Added: N/A
C		Tremied (Y/N): Yes
C		Graduation Designation: #1
		Grain Size: Project File Type: DSI #1 Filter Sand
	22.3'	Amt. Used: 900 lb Tremied (Y / N); Yes
	3.2'	Source: Project File
		Screen
		Type: SCH 40 PVC Diameter: 4-in.
		Length: 10-ft Slot Size and Type: 0.010 Herizoptel Slot
		Interval BGS: 29.0' – 39.0'
		Isolation Casing (Y / N): No Type: N/A
		Diameter: N/A
	39.0*	Sump (Y/N): Yes
		Interval BGS: 39.0' – 41.0' Bottom Cap (Y / N): Yes - 0.4-ft
		Backfill Plug
		Setup/Hydration Time: N/A
		Tremied (Y/N): No
1111	ds T. Carife	
Jun all	Setation for POC	DOE OAK RIDGE OPERATIONS
in log		PADUCAH GASEOUS DIFFUSION PLANT
OH CON		BECHTEL BECHTEL JACOBS COUPANY, LLC MANAGED FOR THE US DEPARTMENT OF ENERGY UNDER US GOVERNMENT CONTRACT DEAC 05 980 072700
Ÿ.	1 PG 2220 / 57 -52 55	Science Applications
in de	Solonitoring Well 359 Construction Log	International Corporation P.O. Box 2502
1111	Ofession Care	Oak Ridge, Tennessee 37831 FIGLIRE No. 000001cask200 ppt
Λ.	reconsecution of In 7200	DATE <u>10-29-01</u>
cpus	G-	18
	8/21/02	

.....

							(2-17		Page_lot_Z
I	LITH	HOLC)GIC	LOC	G BC	DRING/WE	IL N	10.: MW359		
	Facility	r: PGE	OP						Si	te: (274/e U
	Client		Er	vironr	nenta	al Restora	tion	Division BJC 1)		
	Contra	ctor: (SAIC	-	Dri	11 Contractor	· /	16,5	Driller: 5	5. Gautary
1	Start 1	Time:	510	.2/2	1/00	End Tir	ne:	408 2/22/02	Borehole	Dia(s): 8.5
1	Drill M	lethod/R	ig Type:	8 4 Hol	low 5	Im Augors	LML	5' 5 COON (ME 75>	Total De	pth: 48
	Logged	Ву: /	Napl	4. fr	1				Protectio	n Level: D
	Depth		SAMPLE		1	FIELD	5	LITHOLOGIC DESCRIPTION	GRAPH	CONVENTS
	(ft)	Interval	Number	(ft)	ALPHA	Beta/Gamma	VOC's (ppma)		LOG	COMMENTS
	28									
	31						4			
	37 -									
	34									
	36 -									
	37 -									
	36 -							Prill out to 39	7	
A1182A/26JUL **	40 -		Runt		3.6			39.0-42.0 SANDY SELT inth Elay, chink yplowis brange 10412 blb Mocium to Finp Grained 54.0 Moist firm	Me	
	DATA	rivoriai	B1/59:	• • •		DETE:		DATA CHECKED & REVIEWE	d By:	DATE:

LITH	OLOGIC	LOGB	ORING/WEL	L NO.	: Mh1359	;	·····		
Facility:	PGDP						Site	:	
Client:	Env	vironmen	al Restorat	ion Di	vision BIC	1)			
Contrac	tor: (SAIC	Dr	ill Contractor:	MG	5	Dr	iller: 4	- Gautorer	
Start Ti	ime:		End Time	51		Bo	rehole I	Dia(s): 8.5	
Drill Me	thod/Rig Type:	11/4	ell.pc			To	tal Dept	h: 48	
Logged	By: 84 Hollow Sr	IPM AUGON	5, CME 5'5/	0041 ,	(ME 75 7	Pr	otection	Level: D	}
Depth (ft)	SAMPLE Interval Number	(ft)	FIELD MEASUREMENTS	OC's L	THOLOGIC DESCR	IPTION	CRAPH LOG	COMMEN	TS
42	Runt Cont.								
46 48	Runt	3.5.		2 3 m b k	14.0- 47.5 AND interference orderate you rown 104R s Painded blac Pounded peb	wHP lowish 5/4 k NI bles	HINO	76P 26. 48 BI	A Z 15
49		-							
!									





						C	2-17		Pag	re_1 cf_3	-
	LITH	OLOGIC	LOC	BOR	ING/WE	LL N	10.: MW 361				1
	Facility:	PGDP					· · · · · · · · · · · · · · · · · · ·	S	ite: C-746U	LANDFI	LL
Ċ	Client:	E	nvironn	nental	Restora	tion	Division BTC)				-
	Contracto	or: (SAI	C	Drill	Contractor	: M	65	Driller:	MILLER C	W. See	120
	Start Tin	ne: 1405	- 2/	9/02	End Tin	ne: /(05 2/9/02	Borehol	Dia(s): 8.5		
	Drill Met	hod/Rig Type	184 Hol	low sta	m Augers	, CMI.	ES SPOON, CIME 75 7	Total De	pth: 33		
	Logged B	s: Wound	h.pa_					Protecu	on Level: D		-
	Depth (ft) in	SAMPLE	Recovery	ME.	FIELD ASUREMENTS ta/Gamma	VOC's	LITHOLOGIC DESCRIPTION	GRAPH	Сомме	NTS	
•		Lerral Number	3.5 4.0 3.7 5.0	(CEPTR2)			0.0-0.1 Grass Roots scil 0.1-3.5 SILTY CLAY Trifee Sand divik yellowish brow- 10424/2 to grayish orange 104 7/4, firm, subangvilau Moist, <u>CLAY</u> some silt truev sand, Very pale orange 10428/2, MT moderate yellowish brown 104125/4 firm, moist Subangvilau		Poor folo bocavie bluge 10	ирг Ч о <i>F</i> л : 5роон	
A1182/A252	10 11 1Z 13	Runt 3	5.0				CLAY Some Silly Trace Sand light bivish gray SB 7/1 to Yullow Gray SY 8/1, mt moderate brown SYR 4/4, firm Subangu lar, Mois		- - - - - - - -		
(DATA RÉ	earder and the second sec	• • • •	···· / ·	DÁTE:		DATA CHECKED & REVIEWED	BY:		DATE:] ·

C-	1	7
-	-	-

...

Page Zol 3

	LITHOLOGIC LOC	BORING/WELL	NO.: MW361	<u> </u>		1
F	Facility: PGDP		·	Site: (-746 ULAN	DFIL
	Client: Environm	nental Restoration	Division BJC)			T
	Contractor: (SAIC	Drill Contractor:	165	Driller: 5. Go	autney	
	Start Time:	End Time:		Borchole Dia	(s): 8.5	
	Drill Method/Rig Type:84 Hug.	ors, EME 5 SPOO	ON CME 75>	Total Depth:	33	
	Logged By: WaypH. K			Protection Le	evel: D	~
	Depth SAMPLE (ft) Interval Number (ft)	FIELD MEASUREMENTS ALPHAJBeta/Gamma VOC' (cpm) (cpm) (ppm)	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS	
A1182A/26JUIn7	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		CLAY, Some sile trice sund iight greenist gra 54 8/1, Mt moden yehowsh brown 104, 5/4, Subangulan moist firm 190-20.3 SANDY CLAY, trice gravel, moderale yellowish brown 10412 5/4 Mt grayIsh Yellow 10412 7/4 Subangulan, Kirm muist Zw.3 - 23.5 SILTY CLAY, trice Sund, moderate yellowish Drown i0412 5/4, Subangu Meist firm 24.0-27.5 OLAYEY SILT, Trace Sand, Yory Pale organ 1042 8/2, MT dark Yellowish Origin			
(DATA REVORDEBLES:	JERTE:	DATA CHECKED & REVIEWED) by:	DATE:	-

LITH	OLC	GIC	LO	; BC	RING/W	ELL	NO .: MIN 34 1		<u>, , , , , , , , , , , , , , , , , , , </u>
Facility:	· 1	COP	2					Sit	.e:
Client:	/	E	nvironi	nenta	al Restor	ration	Division BTC .)	I	<u> </u>
Contract	or: (SAIC		Dril	11 Contract	.or: n	16 5	Driller: d	5. Gauturpy
Start Tir	ne:	*****			End 1	ime:		Borehole	Dia(s): 8.5
Drill Met	hod/R	ig Type:	84 Ha	llow 4	Hem Augo	V5, C	MES'SPLON IMETST	. Total Dep	pth: 33
Logged B	зу: //	VIMIA	1.Ph				, ,	Protectio	n Level: ρ
		SAMPLE			FELD			lening	
(ft) ir	aterval	Number	Recovery (ft)	ALPHAI (cpm)	Beta/Gamm	VIS La IVOC'a (ppm)	LITHOLOGIC DESCRIPTION	N LOG	COMMENTS
z9 30 31 32 33		Lout. Lout. Pun 7	1.2 4.0			-	SAND some silts the Chy yollowish gray 54 3/1, for riy southed Subery vie from, mois CLAY Some Silt, Tri Sund, yollowish gray SY 3/1, MT moderat Yollowish brown 1042 firm MOIST SILT Y CLAY, Trice to SLAD, Very Palt Organge 1042 ElZ MT moderate yellow brown 1042 5/4 from MDIST Subengulo.		MNÜ Steins

DATA RE	CORDED	LEF:	· · · · .	••.*	DATE:		DATA CHECKED & REVIEWE	D BY:	DATE:

, <i>-</i>			i i	u-1/		Page 1 of 3	<u>)</u>
	LITHOLOG	IC LOG	BORING/WELL I	NO .: MW 361			7
	Facility: PC,	DP		· · · · · · · · · · · · · · · · · · ·	Site: (1-746 LI LAND	ARC
	Client:	Environme	ental Restoration	Division ' ,)]
	Contractor: 6A	sie	MAS	Driller: D.	BISHOP		
	Start Time: 125	5 2/24/0	30 2/26/01	Borehole Di	a(s): 81/4		
	Drill Method/Rig	Type: 8/4 HS	DN, CHE 75	Total Depth: 75			
	Logged By: BRA	DBAKER/	KEN DAVIS		Protection 1	Level: D	-
	Depth SAM	CPLE Recovery ALL	FIELD MEASUREMENTS PHAI Beta / Gamma VOC's	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS	
A1182A/26JUL	(rt) Interval Nur 35 36 37 37 37 37 37 37 37 37 40 41 42 40 41 42 40 41 42 40 43 40 44 45 45 47 48	2.5 5.0 2.5 5.0		SILTY CLAN W/LITTLE GRENUEL, SILT 40% CLAY 45% ARAVEL 15%, REDDISH VELLAN 7.5 YR 78, SUBENING W. POCELY SUBERED, MED, MOIST SAND W/SUME SILT SAND, SAND 70% SILT 30%, STRONG BROWN 7.5 YR 78, VARIMETED REDDISH YELLON 7.5 YR 78, SAND SCHESILT, SAND MEDIUM 70% SILT 30% WITRACE 0=HUSCOVITE AND MND2, REDDISH YELLON 7.5 YR 78; SUBENNED; MEDIUM, MUST SOZTED; MEDIUM, MCIST SANDY GRAVEL SAND; COAZSE 40% GRANCE ANG ULAC; POOLLY SURTED, LOOSE, WIT		-	
(DATA REVORTEBUE	<u> </u>	<u>।</u> 	DATA CHECKED & REVIEWED I	<u> </u>	DATE:	L T

...

G-26

.

1

			412			1110-261		
Faci	lity: 7	GOP					S	ite: C-746 U LANE
Clie	nt:	Enviror	iment	al Restora	tion	Division ()		
Cont	ractor:	GAIC	Dr	ill Contractor	<u>, </u>	165	Driller:	P.BISHOP
Star	t Time:	1255 2/2	6/02	End Tir	ne:		Borehol	e Dia(s): 8 1/4
Drill	Method/	Rig Type: 8/	1 HS	, CHE 5	(SPI	oor, CMR 75	Total De	epth: 75
Logg	ed By: A	RAD BAICH	m	KEN DA	VIS		Protecti	on Level: D
Dent	b	SAMPLE	1	FIELD	5		GRAPH	
(ft)	Interval	Number (ft)	ALPH.	Beta/Gamma	VOC's (ypus)	LITHOLOGIC DESCRIPTION	LOG	COMMENTS
49	1					· ·	$\left \right\rangle$	
50		5.0				SAND W/LITTLE SILT, COARSE 85, SILT 15% REDISISH YELLOW 7.542 7/8,	8	
52	1 100-55				-	503200002D; ΗΦΟ 502720, 20052, WA		
54 55								
56		3.0				2000 NJ 30,44 Q RAVEL, 30,00 HE COARSE 6070, BRAU 2570, 5165 1570, REDOIST VELLAN	iv- iiL	
57 58	55-60				-	7.54R78, SUB- ROUNDID, ARAVEL BUBANGULAL, POUR SOZTED, LOOSE, WE		
59								
41 42		22/5.0				GILAVEL (50 TO) TODO SUZTED (UP TO 3"), BUBZOUNDED TO SUN ANGULAZ, SAND(35°, MEDIUM DECOMESE, SUBZOUNDED, SILT (1572), YELLOZ	24 3- 39	

	1	
<u> </u>		

۰,



•

1	•										Page 2 cf
	LITI	HOLO	OGIC]	LOC	; BO	RING/WE	ELL I	NO .: MW 36/			
	Facilit	y: PG	DP							Sit	e: C-746 U LANDFIC
	Client	:	Envi	ronn	nenta	l Restora	ation	Division ')			
	Contra	ctor: §	SAIC		145	Driller: D. BISHOP					
	Start '	Time: (255 2/	26/0		Boreh	ole	Dia(s): 814			
	Drill M	lethod/R	lig Type: Ş	CON; CHE75	Total	Dep	nth: 75				
	Logged	By: P	BRADE	BAK	EC,	KEN I	DAI	113	Prote	ctio	n Level: D
	Depth (ft)	Internal	SAMPLE Number	wery A	M	FIELD EASUREMENT Beta/Gamma	S VOC's	LITHOLOGIC DESCRIPTION	GRA	PH	COMMENTS
	4 3 -	.5			((1))	(Churt)			Ń		
	64 -	60		\setminus						\setminus	
	65 -			$-\gamma$						7	
	44		3.	2 10				(12AVELCY SAND (12AVELCY SAND JAND (5572) HEDIUM			
	67 -							21) ROUNDED TO SUL			
	60	65-70						Louse, WRT (0.5, WRT (4.6-48.2 FMESAK			NOTE: SAUD IS
	69							WELL SOUTED, EDILLARD FIRH, WAT W/CLAY LIMINATOLS (2010) LAMINATOLS (2010)			HILD (HUSCONTER) AND 1-2720PAQUE HINTIZALS
•	20			\downarrow				LIGHT CHUCH (104R 7/ NEDIVA PLASTICITY, HOD 70-70.2)	Y	
	7(-			3000 AS ABOVE 70.2 -74.9 CLAVEY SILT 50% SILT, V.			
	72 -		4.	210				TO HASSIE LIGHT	2		
а.	73 -	70-75	5					GRANK (1042 7/1) 10TH OVER BUTTON 3ST, HEDINA PLASTICITY	ERNAL		
	74 -							FIRH, NO IST			
11								•			
182A/26	/5 -							TD = 75.0 FT F	345		
101-1		<u> </u>						Λ			
(DATA	REVORDE	B1/FF:	··	·- ·	DATE:		DATA CHECKED & REVIEWED	BY:		DATE:

′ G-28



				C-17		Page_L c1_
LITHO	LOGIC	LOG	BORING/WELL	NO .: MW 362		
Facility:	DGDP		<u></u>		Site:	C-7464LANDE
Client:	Env	ironmen	tal Restoration	Division BJC .)		
Contractor:	(SATC	D	rill Contractor: /	ULS	Driller: (hel un
Start Time:	11.00	2/10/	To Fad Time:	KIIA Shelas	Borehole F	1407110
Delli Kath	ΠΨΨ	2-119/0	De End Inte.	1910 2417/02	Tatal Dank	
Drill Method	/Rig Type: 8	4 Hollow	stem Pugers CM	NE 5 4 POOD [ME +5 2	Iotal Dept	<u>a: 4/</u>
Logged By:	1/ lis pla	· ifu		1	Protection	Level: 1
Depth	SAMPLE	1	FIELD		GRAPH	
(ft) Inter	val Number	(ft) ALPE	Al Beta / Gamma VOC'a	LITHOLOGIC DESCRIPTION	LOG	COMMENTS
28 -						
1				*		
29 -						
•]						
30 -						
-, -				5		
3/ 1						
77						
26				0 1/ 1/ 1/ 22.0		
32				Drill our to sse		
				33.0-35.4 LAND LITHE CLAY	••••	
34		23,		friger filt, dould yelk		
-		5.6		bunced black NI	÷.	
35 -	Runt	2.1		fire grained, soft		
-	1			TO FILM I		
76				34.5-30. ittle yravel	·	
-				Vollowish orque 10%	12	
37-		$\setminus $		6/6 5UDA 14U EN MOIS		
		\mathbf{N}		<i>p</i> ,		
38				CAND 38.0-41.0		
24	Rut	6		dy. IL yollowish arsy	e ·	
	Ku.	20		104R616, Mr black		
40 -	2	7° ° (moderatly sorte C	·	
]].				SUBANGU 14- 59 tura	ted -	
41					· · · ·	
1					1 •	
DATA REED	SEBUER:	·	4-DATE:	DATA CHECKED & REVIEWED	BY:	I DATE:





-	1	-
	4	1
<u> </u>	Ŧ	

3

LITHOLOGIC LOG BORING/WELL NO .: MIN 364												
ŀ	Facilit	J: P.	SPP	10	<u> </u>		_		Sit	e: Provide La Marca		
T	Client	:	E	nviron	mental	Restorat	ion	Division BJC .)				
	Contra	actor: (SATO	2	Drill	Contractor:	n	165	Driller: 5	. Gautney		
L	Start '	Time:)	210	2/1	102	End Time	- 17	350 2/11/02	Borchole	Dia(s): 8.5		
	Drill M	fethod/R	ig Type:	84 Ho!	low ste	m Augers	CM	ESSPOON, CME752	Total Dep	pth: 36		
	Logged	Ву: /	Venp	1. Fu				*	Protectio	n Level: D		
	Depth (ft)	Interval	SAMPLE Number	Recovery (ft)	ME ALPHA Be	FIELD ASUREMENTS ta/Gamma V (cpm) (OC's	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS		
	1		Run th	3.5				B.B-B. SILT Some Clay, trace SILT Some Clay, trace Sunc, dark yellowish Drown ioyk 4/17 Subangula, maist firm 0.8-3.5 STLAY CLAY, Trace Sunt Very Pale organice 10428 MT dark yellowish organic ioy12 Glb, subangula firm to stifk, muist	n duge			
	4 5 6 2		₹UN 100 10 10 10	4.2				4.0-8.2 <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEY NO.34</u> <u>CLAYEY STLT</u> <u>CLAYEY STLT</u> <u>CLAYEN STL</u>				
	9		Run H	4.5				9.0-12.2 <u>SILT</u> some clay, +rate sund, moderate brown ioyi2 5/4 mt very pute brange ibyl2 8/2, subliquiter muist 12.2-12.5 Sume as above		MNO		
	13 - DATA	REVORDE	31/5 9 :			DÆTE:		PXCPDF Strong MWO Stains) BY:	12:5-13: <u>SILT</u> some clay Traces <u>moderate</u> yalowish Drow- 10425/4, MT very plap 0r4nge 104/28/2 <u>moist stiff to firm</u> DATE:		

~	1	7
<u> </u>	T	1

LITHOLOGIC LOG BORING/WELL NO .: MW 364											
Facility: Pt	SOP			Site:							
Client:	Environ	nental Restoration	Division BJC .)								
Contractor: (SAFC.	Drill Contractor:	M6 5	Driller: 5.6	qu Iney						
Start Time:		End Time:		Borchole Dia	(s): 8.5						
Drill Method/F	Rig Type: 84 Ho	How Stron Augors, CM.	NESSABOON CME757	Total Depth:	36						
Logged By: //	Inst-Fre			Protection L	evel: D						
Depth (ft) Interval	SAMPLE Recovery Number (11)	FIELD MEASUREMENTS ALPEAI Beta/Gamma VOC's (cpm) (cpm) (ppm)	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS						
14 15 16 17 17 18 19	4 5.0		14.0-19.0 CLAYEY SILT Very Pale orange POYRS Mt moderale xellowis brown, moist firm to stiff	8/2							
20	Punt 5-0		19.0 - 22.5 <u>CLAY</u> , some silt, Track Sund, right brownish gray 542 6/2, mt moderate ypllowish brown 104R.5/4 Subgagukir, Stikk MOIST		Possible 1.5 A matrial pushed down hole						
25	punt 5.0 6 5.0		24.0-29.0 SILTY CLAY derk yellowish ording 10412 616, mt very PG4e bringe loyRS, muist stift								

acility: PGPT			Site:
lient: Environ	mental Restoration Div	ision BJC)	
ontractor: (SAIC	Drill Contractor:	Driller	: 5. 64. Frey
tart Time:	End Time:	Boreho	bie Dia(s): 8.5
rill Method/Rig Type: 84 Ho	How Hem Augors CME:	500 0 Total	Depth: 3,6
ogged By: Way dit fa	,	Protec	tion Level: 1
(ft) Interval Number (ft)	FIELD MEASUREMENTS ALPHAI Beta / Gamma VOC's (crmi) (crmi) (prm)	HOLOGIC DESCRIPTION	COMMENTS
zg with			
$\begin{array}{c} 29 \\ 30 \\ 30 \\ 31 \\ 32 \\ 32 \\ 33 \end{array}$	29 <u>C</u> da 67 97 37 10 <i>M</i>	d-33.4 AY some silt ill yollowish iby124 inge, mt moderate llowish brown iby12 4, mt pare brown YR 6/2 ist stift	
34 35 - Punt 2.4 8 2.4 36		-0-36.0 ne 95 90000 16477 MNO taining @ 35 5 - 36.0	

·.	a,						0	C-17		Page cf /
	LITI	HOLO	OGIC	LO	G BC	DRING/WE	LL I	NO .: MW 364		
	Facilit	y: PG	DP						s	ite: C-746U LANDER
	Client	:	E	nviron	ment	al Restora	tion	Division 'BTC)		. '
	Contra	ctor: (SAT	C	Dri	11 Contractor	:: M	65	Driller:	5. Gautney
	Start 7	Time: K	520	2/25	7570 2/28/02	Borehol	e Dia(s): 85			
+	Drill M	P=.	ig Type:	84 Hoi	E 5 5000, CME 757	Protect	epth: 88			
	rogged	By:	Wind /	t. Fu					TIUCECL	
	Depth (ft)	Interval	SAMPLE Number	Recovery (ft)	ALPEA (cpta)	FIELD MEASUREMENT Beta/Gamma (cpm)	S VOC'a (ppm)	LITHOLOGIC DESCRIPTION	GRAPH	COMMENTS
	43							Duillout TO 44.0 BLS		
	45		Run ^{ft}	5.0				644.B-49.B SILTY CLAY Tight granish gran 56 8/4 mt moderate yellowish brown 10412 514 firm to stiff	7 MWO N'WO	
	48 415 50 51 52 53		Punt 2	4.0				49. \$ - 52. \$ 54m \$ 65 G bore 52. 4 - 53. \$ <u>5 AND</u> 1: He Clay Hrace G. Gvel, Cark Vellow, 34 orgage 109 616 medium to Coarse Granec, poort Sorted, Soft, Saturat Pebbles to 1"		TOP RGA@ 52 BLS
A1182A/26JUI"	54 - 55 - Data	REVORDE	Ruint 3	2.5				54.0-55.0 Sund, Sime Grave, 140 Olay, moderate yellowism b 1042574, medium Gravm 56 to-2490 SUFE Poble TO I'' Alloweter DATA CHECKED & REVIEWED	e Politica BY:	DATE:
1										

C-17

1													
	LITI	HOLC	GIC	LOC] BO	RING/WE	LL I	NO .: MW364					
	Facilit	5: P1	HDP						Si	.e:			
	Client	:	E	nvironn	nenta	al Restora	tion	Division ' ,)					
	Contra	ctor: (SAI	C	Dril	1 Contractor	· . N	165	Driller:	5. figutney			
	Start '	Time:	<u></u>			End Tir	ne:		Borehole	Dia(s): 8.9	5		
	Drill H	lethod/R	ig Type:	84 40	llow	stem Augon	15,6	ME5 5000 (ME 75)	Total De	pth: 88			
ł	Logged	By:	Min P	H.m					Protectio	n Level: 1	>		
	Depth (ft)	Interval	SAMPLE Number	Recovery	ALPEAN (CPEAN	FIELD TEASUREMENT. Beta/Gamma	VOC's	LITHOLOGIC DESCRIPTION	GRAPH LOG	Сомме	NTS		
	56	•	Runtz	2.5/1.0				55. p-56.5 SAND 1: He Clay dorte yellowish orging	e line				
	57			\backslash				TO YR 6/6, m perion Gra Mocheraty Gortec, Suban Sutural C	1944		İ		
	58		2010 #					58.0-60.5 6 RAVELY SAND Truce Clay, durk					
	60		4	2.5 5.0			,	pounded solt to firm, seturated	610 5. 0 5. 0 5. 0 5. 0	Hit Aug Rokussi	6		
C	67 67									Adjust Adjust	рооч 43/27/02 5рооч		
	63							· · · · ·		,			
	64		Punt 5	\$.¢ 5.\$				NO ROCOUPTY					
	65 -												
>	67 - [4]												
1182A/26JUI **	69	· · ·	Runt	-0.3				SAND 68.0-68.3 Train Clay, moderaty yellowish brown 104k medium frained, 100. To very Borly 500	25/4				
(DATA	REVORDE	B1/F¥:	• •.•	··•	- JOATE:		5064 56 tore Pr	D BY:		DATE:		

- 4		
	-	•

	LITHOI	LOGIC	LOG	BORING/WE	LL N	10.: MW3.64						
	Facility:	PGDP	7					Site:				
	Client:	En	vironm	ental Restora	tion	Division 'BJC.)						
1	Contractor:	(5A2	CC	Drill Contractor	: M	65	Driller	: 4. Gautnay	/			
	Start Time:			End Tir	ne:		Boreh	bie Dia(s): 8.3	-			
	Drill Method	/Rig Type:	8-4 Holl	low Stem Augor	5 LA	NE 5' 30000 (ME 75>	Total	Depth: 88				
	Logged By:	WarpH.	h				Protec	tion Level: D				
ľ	Depth	SAMPLE	ecovery AI	FIELD MEASUREMENTS LIPHAI Bets/Gamma VOC's			GRAI	сомме	NTS			
	74	Grat		epozi (epozi)		NU ROCOVEY.						
	79	Pun # 7	0.0		,	NO ROCOVERY						
	76 72											
	28 79	punt	0.0			NO Recovery	$\left \right $					
	8¢	6										
A1182A/26JUL "/	82 B3	Runt										
(JATA REVOR	DEBIEY:	• • • •	· · · · · · · · · · · · · · · · · · ·	 1	DATA CHECKED & REVIEWED	BY:		DATE:			

. .**†** .

C-17

1.	r.									Page - 17 a	
	LITH	HOLO	OGIC	LO	G BO	RING/WE	ELL N	NO .: MW 364			
	Facilit	r: PG	OP						Sit	.e:	
	Client		E	nvironr	nenta	al Restora	tion	Division '37C)			
	Contra	ctor: (SAI	C	Dril	1 Contractor		MAS	Driller:	5. Goutapy	
	Start 1	lime:				End Ti	ne:		Borchole	Dia(s): 8.5	
	Drill M	lethod/R	ig Type:	84 4.	llow si	tem Augus,	IME	5'SPOON (ME 75>	Total Dep	pth: 88	
	Logged	By:	way	1.00	2~			/ /	Protectio	n Level: D	
$\left \right $			SAMPLE			FIELD		<u></u>	CRIPH		
	(ft)	interval	Number	Recovery (ft)	ALPHAI) (cpmil	EASUREMENT Beta/Gamma (cpm)	VOC's	LITHOLOGIC DESCRIPTION	LOG	COMMENTS	
	84		Run H	1.9							
	86		(0***	5.0				Light gray N7 Moist, firm; sticky, Ivon	17	TOP MC NUTT @ 86.2	
	88							[PMPH + C []4 110 >			
	1										
A1182A/26JUL**											
5	Start Time: End Time: Borehole Dia(s): g. 5 Drill Method/Rig Type: 8 / M/// Law Shan M. Carls, IME 5's Marn, IME 25's Total Depth: 8' Total Depth: 8' Logged By: MAP/I// Law Shan M. Carls, IME 5's Marn, IME 25's										



1

Ĩ

						C	-17		Page	c1_1
LITI	HOLC	GIC	LOC	J BC	RING/WE	LL N	10.: MW 365			
Facilit	5: ¥	GDF	>						Site: (2746 U Ca	600
Client	:	Er	vironr	nenta	al Restora	tion	Division 'BJC,)			
Contra	ictor: (SAIC		Dril	1 Contractor	· M	65	Driller	: 5. Gautory	
Start	Time: 15	530	2/12	102	End Tin	ne: 14	55 2/12/02	Boreho	ole Dia(s): 8.5	
Drill }	fethod/Ri	ig Type:	84 Hul	low fr	tom Augors	, 50	MESPOON, IME75>	Total	Depth: 54	
Logged	By: V	Vingel,	[]_ [/	m				Protec	tion Level: D	
Depth		SAMPLE		3	FIELD	5	LITHOLOGIC DESCRIPTIO	N GRAI	COMMENTS	
(ft)	interval	Number	(ft)	ALPEA (crema)	Heta/Gamma (cpm)	(ypza)		1 100	· · · · · · · · · · · · · · · · · · ·	
							· .			
29 -										
\$0 -										
31 -										
							,			
32 -										
33 -										
							Duill out to 34	1.0		
34 -							34.0 - 38.2 ITIL ::: 440 1/44			
76	-	a.t					1: HAR Sand		·	
97 -		Kun	4.2				paip y pilowish prod 10412 6/2, mt			
36 -	-	١	5.0				prown roy12 51	4 =:		
	-						Guldangular, firs	m	 .	
37-							M0137			
38 -	-							=		
	-		\backslash							
39.							24.0 - 40.5		: m-	
40.		Runt					54me 17 - 19			
, (].	2	4.5							
41			15.0							
DIT	DIVACA	R1/6-9.	• • •	,-	1.2206		DATA CHECKED 1. DENN			
DATA	ALCOADE!		145 (1)		"DATE:		DATA CRECKED & REVIEWI	ш 81: —	DAT	E:

	C-17	Page_Ect_Z
LITHOLOGIC LO	G BORING/WELL NO .: MW 365	
Facility: PGOP		Site:
Client: Environ	mental Restoration Division \mathcal{BTC} .)	
Contractor: (5/4 IC	Drill Contractor:	Driller: 4 Gootney
Start Time:	End Time:	Borehole Dia(s): 8.5
Drill Method/Rig Type: 84 Hol	low stom Auger, 5' CME 50009, CME 75-7	Total Depth: 54
Logged By: Wikell. A.		Protection Level: D
Depth SAMPLE	FIELD MEASUREMENTS LITHOLOGIC DESCRIPTION	GRAPH COMMENTS
(ft) interval Number (ft)	(cpm) (cpm) (cpm)	
472 - Runta 272 - Runta 274 - Zhi 2011 - Zhi	CLAY some silt <u>CLAY</u> some silt <u>very</u> pule orange ioyR mt durk yellowish urgan loyR 6/6 firm mois	\${2 14 2 2
44 45 46 46 47 47 47 47 47 47 47 47 47 47 47 47 47	SILTY CLAY 1:44+Greenish G.a 56 8/1, mt moderate yellowish Drown 10412 514, firm to Stift	Ф 7 ПЛО 1 ПЛО
$ \begin{array}{c} $	249 0-57. Ø SGMP 95 9600 52. Ø-53. 5 SAND, 1: HAP pebbli Hygte Clay Ogek ypllowish bygn 1042 6/6, medium Grained, Subangu, Poorly Souted to Very Poorly Souted	HOT TOP RGA @ 52.0 BLS 141 1 1 1 1 1 1 1 1 1 1 1 1 1
DATA REVORDEBLES:		D BY: DATE:





			C-17		Page Loi 3
LITHOLOGIC	LOGB	ORING/WELL	NO .: MW-367	-	,
Facility:				Si	te:
Client: I	Environment	al Restoration	n Division ')		
Contractor: (Dr	ill Contractor: /	1:11e-	Driller:	Harley Uychang
Start Time: /330/3/260	72	End Time:	2/8/02 1600	Borehole	Dia(s): 814
Drill Method/Rig Typ	e: 145A / (ME-85		Total De	pth:
Logged By: Bran	Jaker	<u>,</u>		Protectio	on Level: D
Depth SAMPLE (ft) Interval Number	Recovery ALPHA	FIELD MEASUREMENTS Beta/Gamma VOC' (ppm)	LITHOLOGIC DESCRIPTIO	N GRAPH LOG	COMMENTS
	4.5' 5.0 90%		5. Hy CLAY (OL), sitt -4 chun ~60%; frace V. fin schod; Brown 7.5 YR Z leafy organics present s broundit V. poorly sorth Gott; meint	0%) 14. 14. 14.	44"ID HSA w/ CME 5A sample-
2			5. Ity CLAY (CL); silt nye clay 609: trice v. Fine s Pole Brown 107 R93; various yellorich brown and It of gebrounded; v. porty seried	iand. ted	
4 - 5	-		Serr, mersh		-
6	3/5.		sitta 70%, chy -30% trace v, fire sard ; Pat Brenn 10 TR & Ha yellandet brown and It g subrounded; V. pourly se soft, moviet	rijstel neeg v het	
<i>3</i> <i>9</i> <i>10</i>				X	
//	4'5' 39%		SILT W/ Sime charge Sill 75%, chy -25% tree V. Fine sauch; Pak Brown 184R & Var yellowich brown conf H. gray; subrequell, V. poerty sorted; st moist		10.0-15.0 Pt run lith description 10-14.0 Pt-
DATA REEGREBLES:	<u> </u>	11	DATA CHECKED & REVIEWE	D BY:	DATE:

						(C-17		Page Z cl Z
L	THOL	OGIC	LOG	BOR	ING/WE	LL I	NO .: MW-367	2	
Fa	cility:							Si	te:
CI	ient:	Er	ivironm	ental	Restora	tion	Division ' ,)		
Co	ntractor:	(Drill (Contractor	:: M	11 ller	Driller:	herley legebarch
Su	art Time:/3	30 / 8 Feb l	12.		End Tir	ne:		Borehole	: Dia(s):
Dr	ill Method/	Rig Type:	HSA	(ME	- 35			Total De	pth:
Lo	gged By:	Brad	Bala				<u></u>	Protecti	on Level: V
De (ft) Interval	SAMPLE	Recovery A	MEA LPHA Bet	FIELD SUREMENT	S VOC'a (ppm)	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS
1	14-							X	See previews pog for 10.0-14.0 Pt litt dagig
-1	5		5/5				SILT in some clay (M) siH + 75%, clay +25 trace v. fine sound; tak Brown Wilk 5 varigation y ellevish box and H, gray sidnered v. poerly sorted; stiff; meight	4). 7. 1.	
) 2 2 2	9 20 1 72 T		5', 5 100 / 4 100 / 4				Clayey SILT (ML); clay = 40%, siltate Lt Gray 104 7 varia the geller, job ander, v. prody castal: stift bein	97.	
2 2 2 2	3 7 7 5 6		4.6'				SELT W/ Some clay [A s:14 ~ 75%, clay 25 Yeller 10 YR & varian	12)	
Z	7	B ¹ 59:	72%				V. ported; still Mo'st DATA CHECKED & REVIEWED	BY:	DATE:

e.		(C-17		Page 3 cf 3
LITHOLOGIC	LOG BOR	ING/WELL N	NO .: MW-367	2	
Facility:				Si	te:
Client: En	nvironmental	Restoration	Division ')		
Contractor: (Drill C	Contractor:	Ailler	Driller:	tarky Hycharch
Start Time: 1330 / 81	doz	End Time:	· · · · · · · · · · · · · · · · · · ·	Borehole	Dia(s):
Drill Method/Rig Type:	HSA/ (ME	- 85		Total De	pth:
Logged By: Brad	Bate /			Protectio	on Level: 1)
Depth SAMPLE (ft) Interval Number	MEA Recovery ALPEA Bet	FIELD SUREMENTS a/Gamma VOC's	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS
28-	4.6 5.0 92%		SILT w/ some chang (M. silt 175%, chy h25%, Yellow 10 th & yurig the Higray; sabrander V. prechy sorted; stiff; moint		25.0-30.0 A run 25.0-29,6 A recovered
30-31-32-33-	4,5', '5,0 90%		SILT w/ some chay (M. sill ~75%, cha ~25 Tellow 104 RS & varies H. grang; sub recorder V. poorly sorted' stiff, moist	Dire Her / 1	
34				X	
DATA RECORDEBLES	رتم بر ر	ÁTE:	DATA CHECKED & REVIEWEI) by:	DATE:

C-17

•••

Page L ci 4

. .

	<u>.</u>						-
	LITHOLOG	GIC LOG E	ORING/WELL	NO .: MW 367			
	Facility: PG	DP			Site:C-	746DLONDA	FICE
	Client	Environmen	tal Restoration	Division ' ,)			
	Contractor: 6	AIC DI	ill Contractor:	195	Driller: D. Z	BISHOP]
	Start Time: 14	32 3/7/08	End Time: /	710 3/7/02	Borehole Dia(s): 844]
	Drill Method/Rig	Type: 8/4 H	SA, CHE 5'	SPOONS, CHE-75	Total Depth:	85	
	Logged By: V	MULLINS			Protection Lev	el: 70	~
	Denth SA	WPLE	FIELD	1	CRAPH		
	(ft) Interval Nu	mber (ft) (rpm)	MEASUREMENIS	LITHOLOGIC DESCRIPTION	LOG	COMMENTS	
	35 -			CLAY, SILTY (4072	3		
				VERY STIFF, HUTT			
	36 -			BROWNISH VELLON	-		
	37 - 25	59%		BROWNISH GIZAN			
	JUL	5.0		(IUYIZ 6/2), MOIST, LESS SILT UPPER			
	38-			ZFT			
	27 -						
1	40 1						L
				CLAN, SANDERUT	Ð		
	41 -			RUNDED, FINE			
		5.0		GIZLIND, HOIST, VERY STIFF, WOTCH	62		
	42-40-	5.0		LIGHT YELLCWISH			
	145			AND VELLOWISH			
	42-			BRUNN (10426/6),			
	44 -						
	45			45-49.5			
				CLAY, SULDY (40)	2		
	46 - 45-	6.0		SUIZED, SUBZENNO	22,		
ATT	47 50	5.0		HONTERD STRONSA			
82A/2				BIZOWN (7.5 YE 5/2) AND GRAY (7.5 YZ			
1019	48-			(6),) TRACE BLACK			
(1
(DATA REVORDEBUT	5 7 :	DATE:	DATA CHECKED & REVIEWED	BY:	DATE:	
		and the second		the second se	and the second se		.1

U-17

Page of 4

TITITOLOG	to toda		10.17/~	7							
LITHOLOG	LITHOLOGIC LOG BORING/WELL NO.: MW 367										
Facility: PGT)P			Sit	1e:C-746D Lawo,						
Client:	Environmenta	al Restoration	Division ' ,)								
Contractor: 6A	LIC Dril	I Contractor:	195	Driller:	P. BISHOP						
Start Time: 143	317/02	End Time: /	710 3/7/02	Borehole	Dia(s): 8-14						
Drill Method/Rig 1	Type: 814 HE	SA, CMR 5'	SPOONS, CHE-75	Total Dep	pth: 85						
Logged By: V.	MULLINS			Protectio	on Level: 7D						
Depth SAM (ft) Interval Nuz	APLE	FIELD (EASUREMENTS Beta (Gamma VOC's (span)	LITHOLOGIC DESCRIPTION	CRAPH LOG	COMMENTS						
44 -			44.5-50 0 CLA SANDA (452) FINE SUZTED, SUBBUNDA STRUIN BIZON NO.54	wal 25/i)	B2022:57 47244 (10 412.6/2), 140.57						
50 51 52 55 53 57 57	2.0		GAND FINE TO MEDIUM GEAINED, WELL SWETEP, SW3 ROWWARD, TRACE MICA, MOIST; LIAM VELLOWIGH BZOWN (INYR 6/4) IN END CAP GRAVEL, CUARSE GRAVEL, CUARSE GRAVEL, CUET, LIAM VILLOWIGH BZOWN	384							
55 56 57 55 57 55 58 60	28" 5-0'		HEAVEL, FINE SAN TO VILLY COARSE PER GIZAVAL, VILLY POOR SULTED, SUBMINUL TO SUBJED, CHERT, GVARTE, TRACE MICA, SATURATED STREAM, BREWIN (7. SYR 5/6)								
53- 60 61-W 62-05	15		SAME AS ABOVE								
DATA REVORDEBLES		DATE:	DATA CHECKED & REVIEWED) BY:	DATE:						

		C-17	Page 2	c1 4						
LITHOLOGI	C LOG BORING/WE	IL NO .: MW 367	Ζ.							
Facility: PGD	P		Site: C-746DLa	نم (Dي لي الي						
Client: Environmental Restoration Division ')										
Contractor: 64	C Drill Contractor	MAS	Driller: D. BISHOP							
Start Time: 1433	2 3/7/02 End Tiz	ne: 17-10 3/7/02	Borehole Dia(s): 8-14							
Drill Method/Rig Typ	e: 8/4 HSA, CMR	Total Depth: 85								
Logged By: V. A	Protection Level: 7D									
Depth SAMPL	E FIELD MEASUREMENT		GRAPH							
(ft) Interval Numb	Recovery ALPHA Beta / Gamma	VOC's IIIIOLOGIC DESCRIPTION	LOG COMMENTS							
69										
69 67 67 70 69	<u>23"</u> <u>5.01</u>	<u>SAND</u> CONZEC (60 (1720/2.), (720/2.), (720/2.), (720/2.), (720/2.), (720/2.), (720/2.), (720/2.), (720/2.), (7.578.								
71 71 72 72 75 73 74	<u>28''</u> 5.0	SADIE AS PREVIOUS SAUPCE								
75		NO RECOVERY								
U-17

LITHOLOGIC LOG BORING/WELL NO.: MW 367 Facility: PGDP SueC-744 ULawor Client Environmental Restoration Division ') Contractor: GA-(C_ Drill Contractor: MGS Driller: D. 51.5.4007 Start Time: 1432 3/7/02 End Time: 1740 3/7/02 Borehole Dia(s): 8/4/ Drill Method/Rig Type: 8/4 HSA CHE 5'57045 CHE-75 Tetal Depth: 85 Logged By: V. MUCLINS Protection Level: TD Depth SAMPLE Recently Interfactors 77 78 80 80 81 81 82 81 82 83 84 84 85 87 87 87 87 87 87 87 87 87 87	· ·					rag					
State The state State The state Client Environmental Restoration Division ') State Deiller: D. SI, SHOP Contractor: GALC_ Drill Contractor: MGS Deiller: D. SI, SHOP Stat Time: J432_3/7/02_ End Time: The J7/02_ Borchole Dia(s): 8½ Drill Method/Rig Type: B/H HSA_CHE 5'S revels CHE-755 Total Depth: 855 Legged By: V. MUCLINS Protection Lere: TO Depth Statement Protection Lere: TO Statement Protection Lere: TO 77 Image: Statement Protection Lere: Free Protection Lere: To Statement Protection Lere: 78 Image: Statement Protection Lere: To Image: Statement Protection Lere: To 81 Image: Statement Protection Lere: To Image: Statement Protection Lere: To 82 Image: Statement Protection Lere: Image: Statement Protection Lere: To Statement Protection Lere: 83 Image: Statement Protection Lere: Image: Statement Protection Lere: To 84 Image: Statement Protection Lere: To Statement Protection Lere: 85 Image: Statement Protection Lere: To Statement Protection Lere: 84	LITHOLÒGI	C LOG BO	RING/WELL N	NO.: MW 367	÷						
Client: Environmental Restoration Division ') Contractor: GA.(C. Drill Contractor: MG.S. Driller: D. S.(5,170) Start Time: (1432) 3/7/02 End Time: 1710'''''''''''''''''''''''''''''''''''	Facility: PGD	P			1	Site: (-746 L	Land				
Contractor: $GA_1(C)$ Drill contractor: MG_1 Driller: $D, B_1, M_0 > P$ Start Time: $(432, 3)/7/02$ End Time: $7/7/02$ Borehole Dia(s): $8/4$ Drill Method/Rig Type: $B/4$ HSA_1 MG_2 $S/7/02$ Borehole Dia(s): $8/4$ Drill Method/Rig Type: $B/4$ HSA_1 MG_2 $S/7/02$ Forection Level: TD Depth SAMPLE $M_1 \in S^1 S Powuld, CAE = 7S^1 Total Depth: ST Total Depth: ST Protection Level: TD SMPLE MG_2 = 100000000000000000000000000000000000$	Client:	Client: Environmental Restoration Division ')									
Start Time: 1/432 3/7/62 End Time: 17/02 Borchole Dials): 8/4 Drill Method/Rig Type: 8/4 H-8A, CMC 5'S Peruss CMC -755 Total Depth: 855 Logged By: V. MUCLINS Protection Level: TD Depth SAMPLE Interval Number Failure (New York ALPENDATE Command York Intelligence Command York Inteligence Command York Intelligence Command York Intelige	Contractor: 64	C Drill	Contractor:	195	Driller:	D. BISHO	>				
Drill Method/Rig Type: B/4 HSA, CMC 5'S Provids, CMC-75 Total Depth: 855 Logged By: V. MUCLINS Protection Level: TD Depth SAMPLE MARSTREEMTS (M) INTERPORT Interval Number, Registreements, Vocal Network COMMENTS 10 Sample: NIPRI Here (reasonal Vocal Network) Interval Number, Registreements, Vocal Network Interval Number, Registreements, Vocal Network COMMENTS 77 Interval Number, Registreements, Vocal Network, Vocal Net	Start Time: 1432	2 3/7/02	End Time: /	710 3/7/02	Boreho	le Dia(s): 84	1				
Lagged By: V. MUCLINS Depth SAMPLE Interval Number MERSINE CRAIMENTS Interval Numer MERSINE C	Drill Method/Rig Ty	pe: 814 HS	A CHE 5'	SPOONS, CHE-75	Total D	epth: 85					
Depth SAMPLE IN ATTENTY Interval Number ALP PAN Beta (Samma 100° a) IITHOLOGIC DESCRIPTION GRAPH 77 Interval Number ALP PAN Beta (Samma 100° a) IITHOLOGIC DESCRIPTION GRAPH 78 Interval Number Interval Number Interval Number Interval Number 78 Interval Number Interval Number Interval Number Interval Number 79 Interval Number Interval Number Interval Number Interval Number 79 Interval Number Interval Number Interval Number Interval Number 79 Interval Number Interval Number Interval Number Interval Number 79 Interval Number Interval Number Interval Number Interval Number 70 Interval Number Interval Number Interval Number Interval Number 80 Interval Number Interval Number Interval Number Interval Number 81 Interval Number Interval Number Interval Number Interval Number 82 Interval Number Interval Number Interval Number Interval Number 83 Interval Number Interval Number Interval Number Interval Number 84 Interval Number	Logged By: V.A	TUCLINS			Protect	ion Level: 7D	,				
Depth SAMPE UPASTREENTS Interval Number ALPERINDETS 77 Interval Number 78 Interval Number 79 Interval Number 70 Interval Number 71 Interval Number 72 Interval Number 73 Interval Number 74 Interval Number 75 Interval Number 76 Interval Number 77 Interval Number 78 Interval Number 79 Interval Number 70 Interval Number 71 Interval Number 72 Interval Number 73 Interval Number 74 Interval Number 75 Interval Number 76 Interval Number 77 Interval Number 78 Interval Number 79 Interval Number 70 Interval Number 71 Interval Number 72 Interval Number 73 Interval Number 74 Interval Number 75 Interval Number 76 Interval Number 77 Interval Number 77 Interv											
77	Depth SAMP (ft) Interval Numt	Recovery ALPHAIL	EASUREMENTS Bets Gamma VOC's (com) (ppm)	LITHOLOGIC DESCRIPTION	GRAP	Н СОММЕ?	TS .				
20 81 81 82 82 83 84 85 10 10 10 10 10 10 10 10 10 10	77			NORECONTRY							
	80 81 82 83 83 84 84	15.0		IN END CAP <u>CLUY</u> GANDY (30) FINE TO MEDINA GIZONDED; SUBALAN STFF, MOTTLED BROWNSA YELLOW (104126/8) AND CIGHT GROY (KUY27/1)							
DATA REVORDEBUTE: DATA CHECKED & REVIEWED BY				TD 85.0 FT 84	Č.						
TALA UTILINE CARACTER CARACTER CONTRACTOR CONTRACT	DATA REVORTERIES.	·	. a)úrr.	DATA CHECKED & DEURSHIN							







_

LITHOLOGIC LOG BORING/WELL NO.: MW370											
Facility	Facility: Paducah Gaseous Diffusion Plant										
Client: Bechtel Jacobs Company											
Contractor: SAIC Drill Contractor: Miller Government Services Driller: W. Doug Bishop											
Start Ti	me: 13:4	1 ол 02-08	-02			End Tim	ne: 15:10 on 02-08-02	Bore	hole Dia(s): 8.5 inches	
Drill M	ethod/Rig	g Type: Ho	ollow Stem	Augers	with CME 75			Tota	I Depth: 3:	5 feet	
Logged	By: Ken	neth R. Da	ivis of SAI	2				Prote w/Ni	ction Leve trile/latex	el: D gloves	
DEPTH (ft)	Interval	SAMPLE Number	Recovery (ft)	FIELD Alpha (cpm)	MEASUREM Beta/Gamma (cpm)	VOCs (ppm)	LITHOLOGIC DESCRIPTION		GRAPH LOG	COMMENTS	
							Silt, clayey, brown (7.5 YR5/2) grading downward yellow (10YR7/6), moist	to		Soil horizon	
5 11 11 11			2.4				Silt, clayey, light gray (10YR7/1)mottled with yello (10YR7/6), moist	w			
		3.0			Silt, clayey, with trace sand, fine, light gray (10YR7 mottled with yellow (10YR7/6) grading downward t light gray (10YR7/1), moist	/1) o					
20 11 11 11			Sand (50%), fine, well sorted, silty, clayey, yelk (10YR7/6) grading downward to medium sand, sorted, very pale brown (10YR8/2) with little ye (10YR8/6) mottling, moist				Mottling may mark bedding planes.				
				Sand, fine, silty, light gray (10YR7/1), moist Sand, medium, well sorted, subrounded to subang quartz with trace opaque minerals, brownish yello (10YR6/8), wet	zular, ow		Water from over-reamed hole at 30 feet				
		• [2.5 2.5				Sand, fine, well sorted, subrounded to subangular quartz with trace opaque minerals, very silty grad downward to silty, yellow-(10YR7/6)-mottled-wi light-gray-(10YR7/1) to brownish yellow (10YR6/6), trace gravel at 32.2 ft and 33.6 ft, moi	ing th-			
35	ECORDE	DBY:	2.5	DATE:		DATA	quartz with trace opaque minerals, very silty grad downward to silty, yellow-(10YR7/6)-mottled-wi light-gray-(10YR7/1) to brownish yellow (10YR6/6), trace gravel at 32.2 ft and 33.6 ft, moi	ing th- st	DA	 TE:	

.

(

(

,

7	1	۰.	
1	1	_	L
1	1	-	L

			•	Page c1_	1
LITHO	LOGIC LO	G BORING/WEL	L NO .: MW 370		
Facility:	PGDP			Site: C-746 U LANZ	DEIL
Client:	Enviro	nmental Restorati	on Division 'BJC.)		_
Contractor:	(SAIC	Drill Contractor:	M65	Driller: 5. 6Gutney	
Start Time:	1635 3/7	102 End Time	:0920 3/8/02	Borchole Dia(s): 8.5	
Drill Method	d/Rig Type: 84	Hollow stem Augers	COME 5'SFOON COME 75>	Total Depth: 78'	_
Logged By:	Min PH: HL	1		Protection Level: 2001 D	
Depth (ft) Inter	SAMPLE Recover	FIELD MEASUREMENTS 7 ALPEA Beta (camma V((cam))	LITHOLOGIC DESCRIPTION	N GRAPH COMMENTS	
34 35 36 37 38 39 40 41	Rum H 4.0 2 4.0 4.0 4.0 4.0 4.0		34 0-38.5 <u>SILT</u> Some Clay. Traip Sunch moderate yellowish Brown 1042 5/4 Mt Ury sule Ovany 1042 8/2, subangu. Muist, 5t. FF <u>39.0-4/2.0</u> SANNY SILT, track Clay. Pole reddish Drown 1012 5/4 Mt Grayish Orang	e Adjust rig hereine	
DATA REEDA	OEBLES:	DATE:	DATA CHECKED & REVIEWED	D HY: DATE:	
and the second division of the second divisio		فتكرب المحيي فتتحدي وسلامي بالتوريك والتعريب ففتحه	and the second se		

						(C-17		Page Zof 4	
LITHOLOGIC LOG BORING/WELL NO .: MW 370										
Facil	ity: /	PGDI	2				, , , , , , , , , , , , , , , , , , , ,	Sit	.e:	
Clier	Client: Environmental Restoration Division 35C)									
Cont	ractor: (SAI	C	Drill	Contractor	: /	M65	Driller:	5. bautney	
Start Time: End Time: Borehole Dia(s): 8.5										
Drill Method/Rig Type: 8 4 Hollow Stem Augor's CME 5' spoon CMET Jotal Depth: 76										
Logge	ed By:	1/init	the					Protectio	n Level: D	
Dept	h	SAMPLE		W	FIELD	5		GRAPH		
(ft)	Interval	Number	Recovery (ft)	ALPHAIE (cpm)	Beta/Gamma (cpm)	VOC's (ppm)	LITHOLOGIC DESCRIPTION	LOG	COMMENTS	
47	T 1 1 1	Runt cout.					42.6-42.6 STLTT SHALL modulite yellow ish boow 1042 514, Subangular I Mois firm to shift	, , , , , , , , , , , , , , , , , , ,	tours of Ivon Compart	
43			1.0				13.\$ -44.\$ 54ND, 1-610 Ciay			
44	liit	2 vint					davil ypillowish orange 104R 6/6 modium To cogise grained			
46							ju bungu law, 50 th Su bungu law, 50 th Su tura lad			
47	والمع									
48		H	1.0.				SAND, Some gravel Some Elay Pale reddish brown			
50	بعبيلية	2044					Rounded pobles			
51							76 1"			
52	-								TO Pack Sumple for	
53							. 54:0-54.1	-0:0 0:0-	pplovery	
55		fur t	.1				GRAVEL Some Sund Trace Clay, Pale roddish brown 10257 Very Pouriy Sorted Saturated To Moist	14		
DAT	A RECORDE	B1 59 :	· •.	••,•	DATE:		DATA CHECKED & REVIEWED	EY:	DATE:	

·	C-17	Page <u>3</u> c1 <u>4</u>									
LITHOLOGIC LOG BORING/WELL NO .: MW 370											
Facility: PGPP		Site:									
Client: Environmental Restoration Division (BTC)											
Contractor: (44EC Drill Contractor: MG 5 Driller: 5. Gauture 4											
Start Time: End Time: Borehole Dia(s): 8.5											
Drill Method/Rig Type: 84 Atollow stan Auger LIME 5 spaces. CME75> Total Depth: 78											
Logged By: Winn H. Fan	Logged By: Windle Kann Protection Level: D										
Depth SAMPLE (ft) Interval Number (ft)	FIELD MEASUREMENTS ALPHAI Beta / Gamma VOC's (cpm) (ppm)	N CRAPH COMMENTS									
56 - Runt 5-2 - Runt 5-8 - Cont. 5-8											
60 - Runt	54.0-54.3 SANDY GRAVEL Trice Clay dark yphouish oviend 10472 6/6 Quivel 2000 dec To subrounded Sand subrounded Suturated form	· O O D. ·									
64 65 66 67 68	SANDY GRAVEL OGARK VP/Rensh orgast 104/2 6/5 Nounced to Subrounded firm to soft Saturated	64.8 0 0 0									
DATA RECORDEBUES:	JOYATE: DATA CHECKED & REVIEWED	DATE:									

۰.					*		, c	-17		Pag	e_Zct_
Ī	ITI	HOLO	OGIC	LOC	BO	RING/WE	LL N	10.: MW370			
5	Facility	γ: · μ	GDP						Sit	te:	
	Client: Environmental Restoration Division BTC ,)										
L	Contractor: (SAIC Drill Contractor: MGS Driller: S. Goutary										
Ŀ	Start Time: End Time: Borehole Dia(s): 8.5										
	Drill Method/Rig Type: 84 Hallow Stem Augor LME 5' spoon, CME 75> Total Depth: 78										
	logged	By:	1/11.0 14	. hr	~				Protectio	n Level: D	
F	Depth		SAMPLE			FIELD	;		GRAPH		
L	(ft)	Interval	Number	Recovery /	LPHAI	Beta/Gamma	VOC's (ypen)	LITHOLOGIC DESCRIPTION	LOG	COMME	STS
	70 -		Run H	1.0/5.0				GRAVEL Somport	0000		
	7/ -		8					da-16 yellowish			[.
	1		cont					orange 104R, 616			[
	7Z -							Saturisted Soft			
								•			
	73-										
	24	÷		V	-		1				
I				2.0,				75.0-76.0	00		
	75		Runt	4.0.				Trace Ring	54		
1	74		9	•				da, kyplowish owing t	20		
	70							5041 76.0 - 76.2	TII		
	77							Tron Lemented Sauce 76.2-770			
								CLAY Some Silt the			· · ·
	78-			17				Muist Shiff Stick	7		
	-										
				1							
				-							
	-										
		1									
1822		1.									
196 JUL	-										
स्		7					1				
4	DATA	REVORDE	Blær:	·	•.•	-DATE:		DATA CHECKED & REVIEWEI	BY:		DATE:

t viji







-	÷	-
0-	Ŧ	1

2).

C

			C-17		Page _1 cf _4
LITHOLO	DGIC LO	G BORING/WELL	NO .: MW373		
Facility: PG	OP			Site: (-74	16ULANDAL
Client:	Environ	mental Restoration	Division BTC .)		
Contractor: (SATC	Drill Contractor: M	45	Driller: 5. 17401	ney
Start Time:	158 2/10	/02. End Time: /	305 2/10/02	Borchole Dia(s):	8.5
Drill Method/H	Rig Type: 84 4	allows stem Augous CM	ESSPEON LMETS>	Total Depth: 3	0
Logged By:	Warill. Ka	n		Protection Level:	D
p	SAMPLE	FIELD	<u></u>	CRUPH	
(ft) Interval	Number (ft)	ALPHAIBeta/Gamma VOC'a (cpm) (cpm) (ppm)	LITHOLOGIC DESCRIPTION	LOG	COMMENTS
	201 7.5 201 4.4		Q. U-1.2 CLAY, Some Silf, Trace Sand, davk yp/lowish Drowin 1042 4/2 Subangula, Moist, firm 1.2-3.5 SILTY CLAY Trace San Vory Pole orange		*
3		-	yp//wwish proven en Orige 104/2 6/6 SUDANGULA-, MOIST Firm 4.0-6.0 SILT Some Clay		-
5 8 7	2 5.0		Yollowigh brown NYR. Subling Ulding Moist firm to Stiff 6.4-9 O SILTY CLAN, Trace Sa. Pale brown 10412 6	5/4 	
8			mt moderate Druw n 14412 5/4, subanquia moist, fir m		
11 12 11 11 11 11	Zunt 4.2 3 5.6		STLT, some iky, Trace sund Iight brownish gray 5 412 6/1, mt moden Yellowish Browniey 514, subanguing me	1	
13			FIRM to Stiff 12.3 - 13.2 CLAY, Some Silt Trace Sand, ignt Gray NF, Mt dark Y Allowish Organ 104/K AUDONING		р мпо 17.15 Ф 1
DATA RECORDE	Blær:	ADATE:	DATA CHECKED & REVIEWED	BY:	DATE:

U-1/		

LITHOLOGIC LOG BORING/WELL NO .: MW 373									
Facility: PGPP Site:									
Client: Environn	nental Restoration Division $B_{\mathcal{T}}$	(۲							
Contractor: (SAIC Drill Contractor: MGS Driller:									
Start Time: End Time: Borehole Dia(s): 8.5									
Drill Method/Rig Type: 84 Hollow Stem Rowrs (MES' 56001, (ME75) Total Depth: 30									
Logged By: // G. J. //- Kanner Protection Level: D									
Depth SAMPLE (ft) Internal Number (n)	FIELD MEASUREMENTS ALPHAIBeta/Gamma VOC's	CREPTION GRAPH COMMENTS							
$ \begin{array}{c} $	(epm) + (epm	Clay, $i = i = i = i = i = i = i = i = i = i =$							
DATA RECORDER	··· DATA CHECKED & H	REVIEWED BY: DATE:							

; ;	C-17 Par										
ſ	LITI	HOLO)GIC	LOC	J BC	RING/WE	IL N	10.: MW 373			
	Facilit	y: Pa	SOP						s	ite:	
	Client		Er	nvironr	nenta	al Restora	tion	Division BTC .)			
	Contractor: (SAIC Drill Contractor: MGS Driller: 5. Gaufny										
	Start 1	e Dia(s): 8.5									
	Drill M	lethod/R	ig Type:	Total D	epth: 30						
	Logged	By:	litagele	1. Km	/				Protecti	on Level:	
ł	Denth		SAMPLE			FIELD	5		GRAPE	r	
	(ft)	Interval	Number	Recovery (ft)	ALPHA (cpm)	Beta/Gamma	VOC'a (ppca)	LITHOLOGIC DESCRIPTION	LOG	COMMENTS	
	28 -		Kuntt Gont.	$\overline{\langle}$				· .			
	29 -		RUNE	1.4				STLTYCLAY, truch			
			7	1.4				54.10,		-	
	30							moderate yellowis brown ioyiz 5/4	4		
	-			×.	ŕ			moist, stiff			
	-										
	-										
Ċ	-										
1	-										
	-										
	-										
1	-										
	-										
	-										
	-										
	-										
ATT											
24/26		.									
101.	-							- 			
(RIÆG		·						
		KECONDEL	····변호:			DATE:		DATA CHECKED & REVIEWED	BY:	DATE:	

,1							(C-17			Page Lof 4		
1	LITH	OLC	GIC	LO	G BC	RING/WE	LL I	NO .: MW 37	3				
	Facility:	PG	DP							Site: C-746	OLAWOFIC	< <u>(</u>	
	Client:		E	nvironi	nenta	al Restorat	tion	Division ' ,)					
Contractor: GA(C) Drill Contractor: H(75) Driller: H. UPCHWZCH Start Time: Ø840 3/11/00 End Time: 1040 3/11/02 Borehole Dia(s): 8/4													
	Drill Met	hod/Ri	ig Type:	8/4	HSA	, CHE	5'	FRON, CHE.85	Total Depth: 75				
	Logged E	sy: U) AYL	E Pa	SIZICI	EZ_			Protec	tion Level:	D	~	
	Depth (ft) in	aterval	SAMPLE Number	Recovery (ft)	ALPHAI)	FIELD TEASUREMENTS Beta / Gamma (cpm)	VOC's (ppm)	LITHOLOGIC DESCRIPTION	GRAF	CO	DMMENTS		
	30 31 32 33 34		PUN #1	22/50				SILT (6070) SAND(CLAY (3070), HOD YELLONISH BROWN LIVYE 7/41) MT LIGH GEAY (N7) MT LIGH SNIGHLULAZ, FIRM	HT HT				
	36 37 38 37 38 39 40		2UNU #2	5/15				35.0 - 37.0 SULT (50% (LAY (30%), 5000 (ZOW), HODREATE YRUCHISH BROWN (WYR 5/4)NT LIGHT GIRAY (NT), HOIST, SUBANGULAR, FIRM 577FE 37.0 - 38.0 CLAN (55% 21.17 (40%), SAND(50%) PALE YELLEWISH BROW (WYR 5/2)MT. LIGHT BROWN (542 576) HO SUBANGULAR, FIRM 3810-40.0 SILT (60) CLAY (20%), SAND (20%)	2)/ 10 12/2 13 2) 2/2 14 1/2 1/2 1/2 1/2 1/2 1/2 1/2 1/2 1/2 1/2	35-40 20 Hoist; Firmt	но Польсіў 50822 чосях 0 577 FF		
/"INF92/72811Y	4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4		2UN #3	3:00				4/010-420 50402 A5 6:3042 42.0-43.0 GEANAL(SAND (4070), CLANGO DAEK YALLOWISH ORACA (1042.6/6), SATUZATE ROINADED TO SUB- ROUNDED TO SUB- ROUNDED TO SUB- ROUNDED TO FIRM	o ration				
(DATA RE	VARDED	1 59 :	•		JDATE:		DATA CHECKED & REVIEWED	BY:		DATE:		

• •

(

					0	2-17		Page Del 4			
LITH	OLOGIC	C LO	G BO	RING/WE	LL I	NO .: MW 37	3				
Facility:	PGDP						Si	te: (-746 ULANDE			
Client:	Client: Environmental Restoration Division ' ,)										
Contract	tor: GAIC	-	Drill	Contractor	:	(65	Driller:	H. UPCHUZCH			
Start Tin	me: Ø840	3/11	192	End Tir	ne: /4	040 3/11/42	Borehole	Dia(s): 81/4			
Drill Met	thod/Rig Type	Total De	pth: 75								
Logged E	By: WAYA		Protectio	n Level: D							
Denth	SAMPLE			FIELD	-		CRIPH				
(ft)	nterval Number	Recovery (ft)	ALPEALE (cpm)	eta/Gamma (com)	VOC's (pper)	LITHOLOGIC DESCRIPTION	LOG	COMMENTS			
44 -		\bigwedge				· ,	\square				
45		2.0				45,0-44.0 GRANEL (5070), 500, CHOTO), CLAY (1070), DARIL YR,LUDDISH UZH (1040 (114), SATURINT	, inc				
47	RUN					PRESIES TO 2"U, POOR SUTTO, SUFT 46.0-47.0 SUND (9020), CLAY	ur				
49						(1072) VELY PALE. ORANGE (1042 8/2), SNBAUGULAR IWIN SUZTEP, SATURATED, SUITEP, SATURATED,					
50		0:3:0				(ARAVEL (7070), DAUD (25%), CLAY (5%), DARC VELICUISH CRANKE					
52	RUND	$\left \right\rangle$				10426/6, 500000000 ROUNIED PERSIES SULT	» \				
53 -	14										
55		10.5				SAND (100720), PRAZ	6	-			
56	· 2000	0,50			(3070), CLAY (1072), DARK YELWINSH BEA (104R412), PRASECCO TO 1/2"1200020, SATURATO, SONT, SI AUGU CAP TO SUB-	113				
DATA RE	NORDEBIES:		<u> </u>	JÚTE:	 	DATA CHECKED & REVIEWED	BY:	DATE:			

. •							(17		Page 3 of 4	6		
Ĩ	LITI	HOLO	OGIC	LO	G BC	RING/WH	ELL I	NO .: MW 37.	3				
	Facilit	r PC	7DP						Si	te: C-746 ULANDFI	c		
	Client: Environmental Restoration Division ')												
	Contractor: GAIC Drill Contractor: H(1)S Driller: H. UPCHW2004												
-	Start Time: 0840 3/11/07 End Time: 1040 3/11/42 Borehole Dia(s): Drill Method/Rig Type: 81/4 H.S.A. OME 5'STOON CHE 85 Total Depth: 75												
+	Drill L	lethod/R	lig Type:	8/4	SPOON, CHE 85	Total De	pth: 75						
Ł	Logged	ву: Ц	JAYA	DE PA	SIZKI	ek		·	Protectio	n Level: D	~		
F	Depth (ft)	Interval	Number	Recovery (ft)	ALPHAI)	FIELD EASUREMENT Beta/Gamma	S VOC'a	LITHOLOGIC DESCRIPTION	GRAPH LOG	COMMENTS			
	58 -							· ·			•		
	62 63		PUN F					NO ZELOVERY					
	64 65 67 68		2000	50				<u>SAND</u> (55%) PEB3LES (40%), CLAY (5 ⁴ 72), DAZIL YELLOWISH B20K2 (10 YR 4/2), PEB3LES TO 3/4" ROUNCO, SOFT SATUZATED					
A1182A/26JUL"	69 - 70 - 71 -	· · ·	といい	5.0				70, 0-70, 8 GIZAVIL (65%); SAVI (30%) (LAY (5%)) DA VIE LLOGISSH OZANIE (1942 G/6) U. 700214 SUN (1942 G/6) U. 700214 SUN (1942 G/6) U. 700214 SUN	25.27	70,8-70.9 ERON CEHENTED FARTO HEDILM SONED			
ς	DATA	REVORDE	Slæt:	• • • •	··	DRTE:]	DATA CHECKED & REVIEWED	BY:	DATE:			

(

					(C-17		Page 4 cl 4
LITHO	LOGIC	LOC	BOR	ING/WE		10.: MW 37	3	
Facility:	PGDP						S	ite: C-746 ULANOF
Client:	E	nvironn	nental	Restora	tion	Division ' ,)		
Contracto:	- GAIC	~	Drill	Contractor	: <i> </i> -	LAS	Driller:	H. UPCHNZCH
Start Tim	: Ø84Ø	3/11/	42	End Tir	ne: 14	940 3/11/42	Borchole	Dia(s): 814
Drill Meth	od/Rig Type	: 81/4	HSA,	CME	- 5'a	FOON, CHE 85	Total De	pth: 75
Logged By	: WAYN	JE PA	,ZKE	2			Protecti	on Level: D
Depth	SAMPLE	Recovery	ME/	FIELD SUREMENT	S VOC's	LITHOLOGIC DESCRIPTION	CRAPH LOG	COMMENTS
72 73 74 75 74 75 74 75 74 75 75 74 75 75 75 75 75 75 75 75 75 75 75 75 75						70.9-75.0 (LAY (SO R)), SHT (1572), SHUD (522) LIGHT BEDWIST, SLITH GIZLEY (5YIZG/1), STIFF, MOIST, SLITH BURECHIS KI"SAUD FILLED BANDED, DA YELLIMISH ORANIZ TTD 75.0 FTB	-)/ er -	

7



•	°.							C-17			Pagecf	-
	LÍTI	HOLO	DGIC	LOO	J BC	DRING/WE	ELL	NO.: 1	11-37	4		٦
	Facilit	.y:									Site:	
	Client: Environmental Restoration Division ')											
	Contra	ictor: (5A.	Driller:	Paul Gibson							
	Start	Time: /5	195 9	Boreho	Borehole Dia(s): 3							
	Drill 1	lethod/F	tig Type:	441	15.4/	\$ 5-10.00	Tinucids	Split Spoon	Total I	Depth: 93,077	_	
	Logged	By:	Bray	nake-						Protect	Ion Level:	
	Depth		SAMPLE	Recovery		FIELD MEASUREMENT	S	LITHOLO	GIC DESCRIPTION	GRAP	H COMMENTS	7
	(11)	Interval	Number	(ft)	(CPEL)	(cpm)	(ppma)	1 79	et dilla			-
								ant-	previews/1			
	38 -			<i>u'</i>				103500	1			ŀ
		28-33		5				23.0-3	2.0 A	,	1 Al de	
	-			40%				clay 9 32	19 · Brownia	h	mantaide of	
	_							Yellow 1	OVRES VOU.	the	Sprin	
								It gray	5 Sprandle	1		
	-							V. peor	aticity if	14)		
	27 -							medium	V. Moir	F		
												\downarrow
	33-			<i>c</i> 1				Clayry SIL	Fi day & Yele,	ller	5 to rated 51, 15	
		33-34		51				RYR G	Jorigated	1	on outside of	
				100%				It. gray	y sorted ;		Spean . And tring (2)	
	-							med gla	esticity ;	i.A	35.Ff - gataratet	
								inen. det	wity, 0:000	2	Soils piesery an	
	53 -	1						38.0- 4	11.0 ft	25		
		74.43		3,5				5:14 w/se	Binnis filler	TC/2		
				TAG				1048.03	station of 1	4		
	-			<i>p</i> =0				913 (h	1 subversedis	1.		
×1								V. poort	sorta ; ma			
82A/2		4						V, MOI	int int			
6JUL ~/	41 -											
Ć						~						
	DATA	RECORDE	DT.B.A:			DATE:		DATA CHEC	KED & REVIEWEI) BY:	DATE:	

•			ruge								
	LITHOLOGIC LOG BORING/WELL NO .: MIN-374										
	Facility:		Site:								
	Client: Enviro	nmental Restoration Divisi	on ' ')								
	Contractor: (5,4JC	Drill Contractor: Milly -	Driller: for bibson	~							
	Start Time: 1545 9.Ma	t 02 End Time: Off	10 Au 02 Borehole Dia(s): 8	"							
	Drill Method/Rig Type: 44	"HSAIN 5. Pt split spl	Total Depth: 43, 6	2							
	Logged By: Bear 1	Paker	Protection Level: D								
ł	Depth SAMPLE	FIELD	GRAPH	GRAPH							
	(ft) Interval Number (ft)	7 ALPHAiBeta/Gamma VOC's LITHO (cpm) (cpms) (ppm) 4/.0	- 41,5 42 LOG COMMENT	rs							
JUL22/X2911X	43	TI	$\frac{1}{3} \frac{1}{3} \frac{1}$								
Ċ											
	PAIR DECORATEPEDE:	DATA CH		DATE:							